UNITED STATES SECURITIES AND EXCHANGE COMMISSION WASHINGTON, D.C. 20549 FORM 8-K

CURRENT REPORT

Pursuant to Section 13 or 15(d)

of the Securities Exchange Act of 1934

Date of Report (Date of earliest event reported): November 30, 2021

INNOVID CORP.

(Exact name of registrant as specified in its charter)

Delaware001-4004887-3769599(State or other jurisdiction
of incorporation)(Commission
File Number)(IRS Employer
Identification No.)

30 Irving Place, 12th Floor
New York, NY 10003
(Address of principal executive offices, including zip code)

Registrant's telephone number, including area code: (212) 966-7555

ION ACQUISITION CORP 2 LTD.

(Former name or former address, if changed since last report)

Check the appropriate box below if the Form 8-K filing is intended to simultaneously satisfy the filing obligation of the registrant under any of the following provisions:		
	Written communications pursuant to Rule 425 under the Securities Act (17 CFR 230.425)	
	Soliciting material pursuant to Rule 14a-12 under the Exchange Act (17 CFR 240.14a-12)	
	Pre-commencement communications pursuant to Rule 14d-2(b) under the Exchange Act (17 CFR 240.14d-2(b))	
	Pre-commencement communications pursuant to Rule 13e-4(c) under the Exchange Act (17 CFR 240.13e-4(c))	
Securities registered pursuant to Section 12(b) of the Act		

Title of each class	Trading Symbol(s)	Name of each exchange on which registered
Common stock, par value \$0.0001 per share	CTV	New York Stock Exchange
Warrants to purchase one share of common stock, each at an exercise	CTVWS	New York Stock Exchange
price of \$11.50 per share		

Indicate by check mark whether the registrant is an emerging growth company as defined in Rule 405 of the Securities Act of 1933 (§230.405 of this chapter) or Rule 12b-2 of the Securities Exchange Act of 1934 (§240.12b-2 of this chapter).

If an emerging growth company, indicate by check mark if the registrant has elected not to use the extended transition period for complying with any new or revised financial accounting standards provided pursuant to Section 13(a) of the Exchange Act.

INTRODUCTORY NOTE

Unless the context otherwise requires, "we," "us," "our," and the "Company" refer to Innovid Corp., a Delaware corporation, and its consolidated subsidiaries. All references herein to the "Board" refer to the board of directors of the Company. Terms used but not defined herein, or for which definitions are not otherwise incorporated by reference herein, shall have the meaning given to such terms in the Proxy and such definitions are incorporated herein by reference.

Merger Transaction

As previously announced, on June 24, 2021, ION Acquisition Corp. 2 Ltd. (<u>'ION</u>", which changed its jurisdiction of incorporation from the Cayman Islands to the State of Delaware on November 29, 2021 (the "<u>Domestication</u>")), entered into an Agreement and Plan of Merger (the '<u>Merger Agreement</u>") with Inspire Merger Sub 1, Inc., a Delaware corporation and a direct, wholly owned subsidiary of ION ("<u>Merger Sub 1</u>"), Inspire Merger Sub 2, LLC, a Delaware limited liability company and a direct, wholly owned subsidiary of ION ("<u>Merger Sub 2</u>" and, together with Merger Sub 1, the '<u>Merger Subs</u>"), and Innovid Inc., a Delaware corporation (<u>'OldCo</u>").

On November 30, 2021, as contemplated by the Merger Agreement and described in the section titled 'The Business Combination Proposal' beginning on page 100 of the final prospectus and definitive proxy statement, dated November 8, 2021 (the "Proxy") and filed with the Securities and Exchange Commission (the 'SEC"), ION consummated the merger transaction contemplated by the Merger Agreement (the "Closing"), whereby (i) Merger Sub 1 merged with and into OldCo (the "First Merger") with OldCo continuing as the surviving corporation of the First Merger, (ii) immediately following the First Merger, OldCo merged with and into Merger Sub 2 (the "Second Merger" and together with the First Merger, the "Mergers") with Merger Sub 2 continuing as the surviving entity of the Second Merger, (iii) immediately following the Second Merger, ION changed its name to "Innovid Corp." (the "Name Change"), and (iv) in connection with the Mergers and the Name Change, Innovid Corp. (formally ION) issued (a) 86,901,792 shares of common stock (the "Registered Shares"), par value \$0.0001 per share ('Company Common Stock') to former equityholders of OldCo and (b) 20,000,000 shares of Company Common Stock to PIPE Investors (as defined below).

Pursuant to the Merger Agreement, immediately prior to the Domestication, each issued and outstanding Class B ordinary share, par value \$0.0001 per share, of ION automatically converted, on a one-for-one basis, into one (1) Class A ordinary share, par value \$0.0001 per share, of ION ("ION Class A Share") in accordance with the terms of ION's organizational documents. Immediately following such conversion, upon the Domestication and the Mergers, (i) each issued and outstanding unit representing one (1) ION Class A Share and one-eighth (1/8) of the warrant to purchase one (1) ION Class A Share at a price of \$11.50 per share (the "ION Warrants") was automatically separated into the underlying ION Class A Share and one-eighth of an ION Warrant, (ii) each ION Class A Share issued and outstanding immediately prior to the Domestication was automatically converted into one share of Company Common Stock, (iii) each whole ION Warrant was automatically converted into a redeemable warrant exercisable for one share of Company Common Stock on the same terms as the ION Warrants (the "Public Warrants"), and (iv) each whole private placement warrant, exercisable for one ION Class A Share at \$11.50 per share, issued and outstanding prior to the Domestication was automatically converted into a warrant exercisable for one share of Company Common Stock on the terms and subject to the conditions set forth in the applicable warrant agreement (the "Private Placement Warrants" and together with the Public Warrants, the "Company Warrants"). No fractional Company Warrants were issued in connection with such separation or conversion such that if a holder of such units was entitled to receive a fractional Company Warrant, the number of such warrants to be issued to such holder upon such separation or conversion was rounded down to the nearest whole number of Company Warrant.

Subscription Agreements

As previously announced, on June 24, 2021, concurrently with the execution of the Merger Agreement, ION entered into subscription agreements, pursuant to which certain accredited investors (the "<u>PIPE Investors</u>") agreed to purchase an aggregate of 15,000,000 Shares of Company Common Stock at \$10.00 per share for an aggregate commitment amount of \$150,000,000 (the "<u>Initial PIPE Investment</u>"). On October 18, 2021, ION entered into new subscription agreements (the "<u>Additional Subscription</u> Agreements", and collectively with the Initial Subscription

Agreements, the "<u>Subscription Agreements</u>") with certain PIPE Investors, including funds affiliated with ION, pursuant to which some of the PIPE Investors collectively subscribed for an additional 5,000,000 shares of Company Common Stock at \$10.00 per share for an aggregate commitment amount of \$50,000,000 (the "<u>Additional PIPE Investment</u>" and together with the Initial PIPE Investment the "<u>PIPE Investment</u>").

Item 1.01. Entry into a Material Definitive Agreement.

Investor Rights Agreement

In connection with the Closing, the Company entered into investor rights agreement (the 'Investor Rights Agreement') among certain former ION equityholders and certain former OldCo equityholders (each, an "Eligible Investor"). The Investor Rights Agreement, subject to the terms thereof, requires the Company to, among other things, file a resale shelf registration statement on behalf of such Eligible Investors and their respective permitted transferees within thirty (30) calendar days following the Closing. The Investor Rights Agreement also provides for certain demand rights and piggyback registration rights in favor of each of the Eligible Investors and their respective permitted transferees, subject to customary underwriter cutbacks. The Company has agreed to pay certain fees and expenses relating to registrations under the Registration Rights Agreement.

The foregoing description of the Investor Rights Agreement does not purport to be complete and is qualified in its entirety by the terms and conditions of the Investor Rights Agreement, which is attached hereto as Exhibit 10.2 and is incorporated herein by reference.

Sponsor Support Agreement

As described in the Proxy, in connection with the execution of the Merger Agreement, ION, OldCo, the Sponsor, and certain members of the board of directors and management team of ION (the "<u>Insiders</u>"), entered into a Sponsor Support Agreement (the '<u>Sponsor Support Agreement</u>") pursuant to which the Sponsor and Insiders agreed, among other things, to not to (i) transfer any of its shares of Company Common Stock or warrants for certain periods of time as set forth in the Sponsor Support Agreement, subject to certain customary exceptions or (ii) enter into any voting arrangement that was inconsistent with the commitment under the Sponsor Support Agreement to vote in favor of the approval and adoption of the Business Combination.

In addition, in connection with the Merger Agreement, certain of OldCo's institutional investors entered into separate support agreements (the <u>OldCo Stockholder Support Agreements</u>") pursuant to which such investors agreed, among other things, (i) to vote all shares of capital stock of OldCo held by such investor at the time of such vote (a) in favor of the approval and adoption of the Business Combination, the Merger Agreement and each of the Transaction Proposals (as defined in the Merger Agreement), (b) against any other business combination proposal or related proposals and (c) against any proposal, action or agreement that would reasonably be expected to impede, frustrate, or prevent the Business Combination or the satisfaction of any of the conditions thereto and (ii) not to transfer any of its shares of Company Common Stock or warrants for certain periods of time as set forth in the OldCo Stockholder Support Agreements, subject to certain customary exceptions. Each such investor further represented and agreed that such investor has not entered into, and will not enter into, any agreement that would restrict, limit or interfere with the voting agreement made in its OldCo Stockholder Support Agreement.

Lock-up

As noted in the Proxy, the Sponsor Support Agreements also provide that certain equity securities held by Innovid stockholders immediately following the consummation of the Closing will be locked-up for the earlier of (i) one hundred eighty (180) days after the Closing Date; or (ii) the Lock-Up Termination Date as defined in the Sponsor Support Agreement except that the Additional Stockholder Support Agreement provides for a lock-up period solely of one hundred eighty (180) days following the Closing Date. The lock-up obligations in the Sponsor Support Agreements described above are subject to certain customary exceptions (including transfer to any affiliates).

Indemnification Agreements

On the Closing Date, the Company entered into indemnification agreements with each of its directors and executive officers.

Each indemnification agreement provides for indemnification and advancements by the Company of certain expenses and costs relating to claims, suits or proceedings arising from each director or executive officer's service to the Company, or, at the Company's request, service as directors or officers of other entities, in each case, to the maximum extent permitted by applicable law.

The foregoing description of the indemnification agreements does not purport to be complete and is qualified in its entirety by the terms and conditions of the indemnification agreements, a form of which is attached hereto as Exhibit 10.8 and is incorporated herein by reference.

The Innovid Corp. Incentive Plan

On November 30, 2021, the Innovid Corp. Incentive Plan (the 'Innovid Corp. Incentive Plan') became effective. At the special meeting of the ION shareholders held on November 29, 2021 (the "Special Meeting"), the ION shareholders approved the Innovid Corp. Incentive Plan.

The purposes of the Innovid Corp. Incentive Plan are to attract, retain and motivate officers and key employees (including prospective employees), directors, consultants and others who may perform services for the Company to compensate them for their contributions to the long-term growth and profits of the Company and to encourage them to acquire a proprietary interest in the success of the Company. These incentives are provided through the grant of stock options (including incentive stock options intended to be qualified under Section 422 of the Code), stock appreciation rights, restricted stock, restricted stock units, dividend equivalent rights and other stock-based awards. Any of these awards may, but need not, be made as performance-based incentive awards.

A total number of Company Common Stock equal to 10% of the fully-diluted shares outstanding following the Closing will initially be authorized and reserved for issuance under the Innovid Corp. Incentive Plan, which is 15,617,049 shares of Company Common Stock. The number of shares authorized and reserved for issuance will be subject to an annual increase for ten years on the first day of each calendar year beginning January 1, 2022, equal to the lesser of (A) 5% of the aggregate number of shares of Company Common Stock outstanding on the final day of the immediately preceding calendar year and (B) such smaller number of shares as is determined by the board of directors of Innovid Corp. The maximum number of shares of Company Common Stock that may be issued pursuant to the exercise of incentive stock options ("ISOs") granted under the Innovid Corp. Incentive Plan will be equal to 30% of the total number of issued and outstanding shares of Company Common Stock on a fully diluted basis as of the Closing.

If shares covered by an award are not purchased or are forfeited or expire, or otherwise terminate without delivery of any shares subject thereto, then such shares will, to the extent of any such forfeiture, termination, cash-settlement or expiration, be available for future grant under the Innovid Corp. Incentive Plan. The payment of dividend equivalent rights in cash in conjunction with any outstanding awards will not be counted against the shares available for issuance under the Innovid Corp. Incentive Plan, and shares tendered by a participant, repurchased by the Company using proceeds from the exercise of stock options or withheld by the Company in payment of the exercise price of a stock option or to satisfy any tax withholding obligation for an award will not again be available for future awards.

A more complete summary of the terms of the Innovid Corp. Incentive Plan is set forth beginning on page 151 of the Proxy, in the section titled *The Innovid Corp. Incentive Plan Proposal*." That summary and the foregoing description of the Innovid Corp. Incentive Plan are qualified in their entirety by reference to the text of the Innovid Corp. Incentive Plan, which is filed as Exhibit 10.3 hereto and incorporated herein by reference.

The Innovid Corp. Employee Stock Purchase Plan

On November 30, 2021, the Innovid Corp. Employee Stock Purchase Plan (the 'ESPP") became effective. At the Special Meeting, the ION shareholders approved the ESPP.

The ESPP will be comprised of two distinct components in order to provide increased flexibility to grant options to purchase shares of Company Common Stock under the ESPP. Specifically, the ESPP will authorize (1) the grant of options to employees that are intended to qualify for favorable U.S. federal tax treatment under Section 423 of the Code (the "Section 423 Component"), and (2) the grant of options that are not intended to be tax-qualified under Section 423 of the Code to facilitate participation for employees who are not eligible to benefit from favorable U.S. federal tax treatment and, to the extent applicable, to provide flexibility to comply with non-U.S. laws and other considerations (the "Non-Section 423 Component"). The Non-Section 423 Component will generally be operated and administered on terms and conditions similar to the Section 423 Component, except as otherwise required by applicable law, rule or regulation.

A total of 2,868,438 shares of Company Common Stock will be initially reserved for issuance under our ESPP. The compensation committee of our board of directors will be the plan administrator of the ESPP and will have authority to interpret the terms of the ESPP and determine eligibility of participants. In addition, on the first day of each calendar year beginning on January 1, 2022 and ending on (and including) January 1, 2031, the number of shares available for issuance under the ESPP will be increased by a number of shares equal to the lesser of (i) 1% of the shares outstanding on the final day of the immediately preceding calendar year, and (ii) such smaller number of shares as determined by the board of directors. If any right granted under the ESPP terminates for any reason without having been exercised, the shares subject thereto that are not purchased under such right will again be available for issuance under the ESPP. Notwithstanding the foregoing, no more than 17,383,002 shares of Company Common Stock may be issued under the Section 423 Component of the ESPP.

A more complete summary of the terms of the ESPP is set forth beginning on page 157 of the Proxy in the section titled 'The Innovid Corp. ESPP Proposal.' That summary and the foregoing description of the ESPP are qualified in their entirety by reference to the text of the ESPP, which is filed as Exhibit 10.4 hereto and incorporated herein by reference.

Item 2.01. Completion of Acquisition or Disposition of Assets.

The disclosure set forth under "Introductory Note — Merger Transaction" above is incorporated into this Item 2.01 by reference.

Forward-Looking Statements

This Current Report contains forward-looking statements. These forward-looking statements include, without limitation, statements relating to expectations for future financial performance, business strategies or expectations for our business. These statements are based on the beliefs and assumptions of the management of the Company. Although Company believes that their respective plans, intentions and expectations reflected in or suggested by these forward-looking statements are reasonable, Company cannot assure you that either will achieve or realize these plans, intentions or expectations. These statements constitute projections, forecasts and forward-looking statements, and are not guarantees of performance. Such statements can be identified by the fact that they do not relate strictly to historical or current facts. When used in this Current Report, words such as "anticipate", "believe", "can", "continue", "could", "estimate", "expect", "forecast", "intend", "may", "might", "plan", "possible", "potential", "project", "seek", "should", "strive", "target", "will", "would" and similar expressions may identify forward-looking statements, but the absence of these words does not mean that a statement is not forward-looking.

Kost Forer Gabbay & Kasierer, a member of Ernst & Young Global, Company's independent auditor, has not examined, compiled or otherwise applied procedures with respect to the accompanying forward-looking financial information presented herein and, accordingly, expresses no opinion or any other form of assurance on it.

Forward-looking statements in this Current Report and in any document incorporated by reference in the Proxy may include, for example, statements about Company prior to the Business Combination and the Company following the Business Combination, including:

- the ability of Company to realize the benefits expected from the Business Combination;
- the ability to maintain the listing of the Company Common Stock on NYSE;
- · the ability to raise financing in the future and to comply with restrictive covenants related to long-term indebtedness;
- · the future financial performance of the Company following the Business Combination;
- the Company's ability to retain or recruit, or to effect changes required in, its officers, key employees or directors following the Business Combination;
- changes in the market for Company's products and services and the Company's ability to effectively compete in the advertising technology industry following the Business Combination:
- · the Company's ability to comply with laws and regulations applicable to its business; and
- · expansion plans and opportunities.

These forward-looking statements are based on information available as of the date of this Current Report and Company's management teams' current expectations, forecasts and assumptions, and involve a number of judgments, known and unknown risks and uncertainties and other factors, many of which are outside the control of the Company and their respective directors, officers and affiliates. Other risks and uncertainties indicated in this Current Report, including those set forth under the section entitled "Risk Factors" beginning on page 44 of the Proxy.

Business

The business of ION prior to the Business Combination is described in the Proxy in the section titled 'Information About ION' and that information is incorporated herein by reference. The business of Company is described in the Proxy in the section titled "Information about Innovid" and that information is incorporated herein by reference.

Risk Factors

The risk factors related to Company's business and operations are set forth in the Proxy in the section titled Risk Factors" and that information is incorporated herein by reference

Financial Information

Reference is made to the disclosure set forth in Item 9.01 of this Current Report on Form 8-K concerning the financial information of ION and OldCo. Reference is further made to the disclosure contained in the Proxy in the sections titled "ION's Management's Discussion and Analysis of Financial Condition and Results of Operations", and "Innovid's Management's Discussion and Analysis of Financial Condition and Results of Operations", which are incorporated herein by reference. Reference is further made to the disclosure contained in ION's Quarterly Report on Form 10-Q/A (the "ION 10-Q") for the quarterly period ended September 30, 2021 and filed with the SEC on November 26, 2021 in the section entitled Management's Discussion and Analysis of Financial Condition and Results of Operations.

Management's Discussion and Analysis of Financial Condition and Results of Operations

The Management's Discussion and Analysis of Financial Condition and Results of Operations for OldCo for the nine months ended September 30, 2021 is filed herewith as Exhibit 99.3 and incorporated herein by reference.

The Management's Discussion and Analysis of Financial Condition and Results of Operations of ION for the nine months ended September 30, 2021 is described in the ION 10-Q in the section titled "Management's Discussion and Analysis of Financial Condition and Results of Operations," which is incorporated herein by reference.

Quantitative and Qualitative Disclosures about Market Risk

Reference is made to the disclosure contained in the Proxy in the section titled "Management's Discussion and Analysis of Financial Condition and Results of Operations of Innovid—Quantitative and Qualitative Disclosures About Market Risk", which is incorporated herein by reference.

Reference is also made to the disclosure contained in the ION 10-Q in the section titled "Quantitative and Qualitative Disclosures Regarding Market Risk," which is incorporated herein by reference.

Security Ownership of Certain Beneficial Owners and Management

The following table sets forth information known to the Company regarding the beneficial ownership of Company Common Stock as of the Closing Date of the Business Combination by:

- a. each person who is a named executive officer or director of Company;
- b. all executive officers and directors of Company as a group; and
- c. each person who is a beneficial owner of more than 5% of Company Common Stock.

Beneficial ownership is determined according to the rules of the SEC, which generally provide that a person has beneficial ownership of a security if he, she or it possesses sole or shared voting or investment power over that security, including options and warrants that are currently exercisable or exercisable within 60 days. Unless otherwise indicated, Company believes that all persons named in the table below have sole voting and investment power with respect to the voting securities beneficially owned by them.

The beneficial ownership of Company Common Stock is based on 118,941,618 shares of Company Common Stock and 10,222,500 warrants to purchase shares of Company Common Stock issued and outstanding as of the Closing Date.

Name and Address of Beneficial Owner(1)	Number of Shares of Company Common Stock	0/0
5% Holders		
Genesis Partners III L.P.(1)	19,350,638	16.3 %
Sequoia Capital Israel IV Holdings L.P.(2)	17,697,224	14.9 %
Special Situations Investing Group II, LLC(3)	9,876,359	8.3 %
Lauderdale GMBH & Co. KG(4)	7,277,981	6.1 %
ION Holdings 2, LP(5)	17,560,000	14.0 %
Directors and Executive Officers		
Gilad Shany	_	_
Rachel Lam	_	_
Jonathan Saacks	_	_
Steven Cakebread	_	_
Michael DiPiano	_	_
Zvika Netter(6)	6,374,089	5.3 %
Tal Chalozin(7)	4,701,861	4.0 %

Tanya Andreev-Kaspin	464,276	*
All directors and officers as a group (eight individuals)	11,540,227	9.7 %

- * Less than one percent
- (1) Genesis Partners III L.P. is controlled by Eyal Kishon and Eddy Shalev. Kishon and Shalev otherwise disclaim beneficial ownership over the shares beneficially owned by Genesis Partners III L.P. The address for Genesis Partners III L.P. is 13 Basel st., Herzeliya, Israel, 4666013.
- (2) SC ISRAEL IV GENPAR, LTD is the general partner of SC ISRAEL IV MANAGEMENT, L.P., which is the general partner of SEQUOIA CAPITAL ISRAEL IV L.P., which owns 100% of SEQUOIA CAPITAL ISRAEL IV HOLDINGS, L.P. As such, SC ISRAEL IV GENPAR, LTD shares voting and dispositive power with respect to the shares held by SEQUOIA CAPITAL ISRAEL IV HOLDINGS, L.P. The directors and stockholders of SC ISRAEL IV GENPAR, LTD. who exercise voting and investment discretion with respect to the shares held by SEQUOIA CAPITAL ISRAEL IV HOLDINGS, L.P. are Shmuel Levy and Haim Sadger. As a result, and by virtue of the relationship described in this footnote, each such person may be deemed to share voting and dispositive power with respect to the shares held by the Sequoia Capital Israel IV Holdings, L.P. The address for these entities is 50 Eli Landau Blvd, Herzelia, Israel 4685150.
- (3) The shares are held of record by Special Situations Investing Group II, LLC, which is an affiliate of Goldman Sachs & Co. LLC, a New York limited liability company and a broker-dealer. Goldman Sachs & Co. LLC is a member of the New York Stock Exchange and other national exchanges. Goldman Sachs & Co. LLC is a direct and indirect wholly-owned subsidiary of The Goldman Sachs Group, Inc. ("GS Group"). GS Group is a public entity and its common stock is publicly traded on the New York Stock Exchange and it is managed by its board of directors. The shares of common stock held by Special Situations Investing Group II, LLC were acquired in the ordinary course of its investment business and not for the purpose of resale or distribution. GS Group may be deemed to beneficially own the securities held by Special Situations Investing Group II, LLC. GS Group disclaims beneficial ownership of such securities except to the extent of its pecuniary interest therein. The mailing address for Special Situations Investing Group II, LLC is 200 West Street, New York, New York 10282.
- (4) Lauderdale GmbH & Co. KG, a German limited partnership ("Lauderdale"), is the record holder of such shares. All investment and disposition decisions for Lauderdale are made by an investment committee comprised of Mr. Vicente Vento Bosch (chair) and Mr. Raphael Kuebler. As a result, Mr. Kuebler and Mr. Bosch may be deemed to share dispositive power with respect to the shares held by Lauderdale and thus to have beneficial ownership of such shares. Each of Mr. Kubler and Mr. Bosch otherwise disclaims beneficial ownership of such shares other than to the extent of any pecuniary interest therein.

Lauderdale is managed by its managing limited partner, Deutsche Telekom Capital Partners Management GmbH, a German limited liability company (DTCP-M"). As the managing limited partner, DTCP-M has voting power with respect to the shares held by Lauderdale and therefore may be deemed to have beneficial ownership of such shares.. None of the members of DTCP-M's investment committee (*Beirat*) is deemed a beneficial owner of the Shares under Section 13(d) of the Exchange Act due to the approval standard for committee action. Thus, each such committee member disclaims any beneficial ownership of the shares held by Lauderdale other than to the extent of any pecuniary interest therein.

DTCP-M is controlled by Deutsche Telekom AG, a publicly traded company organized under the laws of Germany ("DTAG"), and Deutsche Telekom Capital Partners Executive Pool GmbH & Co. KG, a German limited liability company ("DTCP Executive Partner"). Each of DTAG and DTCP Executive Partner may be deemed to share voting and dispositive power over the shares held directly by Lauderdale and therefore to have beneficial ownership of such shares. DTAG's stock is traded on seven stock exchanges in Germany, Including the Frankfurt Stock Exchange and OTCQX, and it is managed by its board of directors. DTCP Executive Partner is controlled by Mr. Bosch. DTAG and Mr. Bosch otherwise disclaim beneficial ownership of the shares owned by Lauderdale except to the extent of any pecuniary interest therein. The mailing address for Lauderdale, DTCP-M, DTCP

Executive Partner, Mr. Kubler and Mr. Bosch is Am Sandtorpark 2, 20457 Hamburg, Germany,. The mailing address for DTAG is Friedrich-Ebert-Allee 140, 53113 Bonn, Germany.

- (5) ION Holdings 2, LP is the record holder of such shares. As the general partner of ION 2 LP, ION Acquisition Corp GP Ltd. ("ION GP") has voting and investment discretion with respect to the ordinary shares held by ION 2 LP. An investment committee comprised of five individuals Jonathan Kolber, Gilad Shany, Avrom Gilbert, Stephen Levey and Jonathan Half makes voting and investment decisions in the ordinary shares indirectly owned by ION GP. As a result, ION GP may be deemed to have beneficial ownership of the shares held directly by ION 2 LP. However, none of the ION GP investment committee's members is deemed a beneficial owner of the shares held by ION 2 LP under Section 13(d) of the Exchange Act, due to the approval standard for committee action. Thus, each such investment committee member disclaims any beneficial ownership of the shares held by ION 2 LP, other than to the extent of any pecuniary interest therein. Includes 7,060,000 Company Warrants held by ION Holdings 2, IP
- (6) Consists of 3,441,907 shares held directly by Mr. Netter (including 1,027,914 shares underlying options) and 977,394 shares held by each of the Zvika Netter 2021 Family Trust #1, the Zvika Netter 2021 Family Trust #3, respectively.
- (7) Consists of 2,294,806 shares held directly by Mr. Chalozin (including 45,132 shares underlyin options) and 1,203,528 shares held by each of the Tal Chalozin 2021 Family Trust #1 and the Tal Chalozin 2021 Family Trust #2, respectively.

Directors and Executive Officers

The Company's directors and executive officers after the consummation of the Transactions are described in the Proxy in the section titled 'Management the Company Following the Business Combination' and that information is incorporated herein by reference.

Director Independence

Information with respect to the independence of the Company's directors is set forth in the Proxy in the section titled "Management of the Company Following the Business Combination —Director Independence" and that information is incorporated herein by reference.

Committees of the Board of Directors

Information with respect to the composition of the committees of the Board of Directors (the 'Board'') immediately after the Closing is set forth in the Proxy in the section titled "Management of the Company Following the Business Combination — Committees of Board' and that information is incorporated herein by reference.

Executive Compensation

A description of the compensation of the named executive officers of ION before the consummation of the Business Combination and the named executive officers of Company after the consummation of the Business Combination is set forth in the Proxy in the section titled "Directors, Officers, Executive Compensation and Corporate Governance of ION Prior to the Business Combination" and that information is incorporated herein by reference.

Reference is made to the disclosure set forth above in Item 1.01 of this Current Report on Form 8-K under the headings "The Innovid Corp. Incentive Plan" and "The Innovid Corp. Employee Stock Purchase Plan," which is incorporated herein by reference.

Director Compensation

On November 30, 2021, the Board approved the Innovid Corp. Non-Employee Director Compensation Program (the "<u>Director Compensation Program</u>"). The Director Compensation Program became effective on November 30, 2021, and is designed as a compensation program for our non-employee directors under which each

non-employee director will receive the following amounts for their services on the Board and that will supersede any prior compensation arrangements of our non-employee directors for service on our board of directors:

- a. upon the director's initial election or appointment to our board of directors that occurs after the effective date of the program, a number of restricted stock units equal to \$350,000 divided by the closing price of our common stock on the date of grant,
- b. on the date of our annual meeting, if the director has served on our board of directors for at least six months as of the date of an annual meeting of stockholders, a stock based award with a grant date fair market value equal to \$175,000, which will be granted in the form of restricted stock units equal to \$175,000 divided by the closing price of our common stock on the date of grant (such award, the "Subsequent Award"),
 - c. an annual director fee of \$30,000, and
 - d. if the director serves on a committee of our board of directors or in the other capacities stated below, an additional annual fee as follows:
 - 1. chair of the audit committee, \$20,000,
 - 2. audit committee member other than the chair, \$10,000,
 - 3. chair of the compensation committee, \$20,000,
 - 4. compensation committee member other than the chair, \$10,000,
 - 5. chair of the nominating and corporate governance committee, \$20,000, and
 - 6. nominating and corporate governance committee member other than the chair, \$10,000.

Restricted stock units granted upon a director's initial election or appointment will vest in three equal annual installments following the date of grant. The Subsequent Award granted annually to directors will vest in a single installment on the earlier of the day before the next annual meeting or the first anniversary of the date of grant. All vesting of equity awards is subject to a director's continued service on our Board through the vesting date. In addition, all unvested restricted stock units or stock based awards, as applicable, will vest in full upon the occurrence of a change in control.

Director fees under the program will be payable in arrears in quarterly installments not later than the fifteenth day following the final day of each calendar quarter, provided that the amount of each payment will be prorated for any portion of a quarter that a director is not serving as a non-employee director or a committee member on our board and no fee will be payable in respect of any period prior to the effective date of the program. The foregoing description of the Director Compensation Program is qualified in its entirety by reference to the text of the Director Compensation Program which is filed as Exhibit 10.7 hereto and incorporated herein by reference.

Certain Relationships and Related Party Transactions

Certain relationships and related party transactions of the Company are described in the Proxy in the section titled "Certain Relationships and Related Person Transactions" and that information is incorporated herein by reference.

In connection with Closing, loans to Innovid's founders, Zvika Netter and Tal Chalozin, with a total principal amount of \$1,199,000 (the Founders Loans") and related interest were forgiven in November 2021. \$740,000 of the Founders Loans principal amount was used to exercise fully vested options held by the founders on the date of the grant of the Founders Loans and the remainder in the amount of \$459,000 was used for other purposes. For additional information, see Note 10 to Exhibit 99.1 to this Current Report on Form 8-K.

Legal Proceedings

Reference is made to the disclosure regarding legal proceedings in the section of the Proxy titled "Information About ION—Legal Proceedings" and "Information About Innovid —Legal Proceedings" and that information is incorporated herein by reference.

Market Price of and Dividends on the Registrant's Common Equity and Related Stockholder Matters

Information about the ticker symbol, number of stockholders and dividends for ION's securities is set forth in the Proxy in the section titled 'Market Price and Dividends Of Securities' and such information is incorporated herein by reference.

As of the Closing Date, there were approximately 275 holders of record of Company Common Stock and approximately two holders of record of Public Warrants.

The Company Common Stock and the Public Warrants began trading on NYSE under the symbols "CTV" and "CTVWS", respectively, on November 30, 2021. ION's public units automatically separated into their component securities upon consummation of the Business Combination and, as a result, no longer trade as a separate security and were delisted from the New York Stock Exchange.

The Company has not paid any cash dividends on shares of its common stock to date. The payment of cash dividends in the future will be dependent upon it revenues and earnings, if any, capital requirements and general financial condition. The payment of any dividends will be within the discretion of the Board.

Recent Sales of Unregistered Securities

Reference is made to the disclosure set forth below under Item 3.02 of this Current Report on Form 8-K concerning the issuance and sale by ION of certain unregistered securities, which is incorporated herein by reference.

Description of Registrant's Securities to Be Registered

The description of the Company's securities is contained in the Proxy in the section titled "Description of Innovid's Securities" and that information is incorporated herein by reference.

Immediately following Closing, there were 125,878,826 shares of Company Common Stock issued and outstanding, held of record by approximately 275 holders and 10,222,500 warrants to purchase Company Common Stock issued and outstanding, held of record by two holders.

Indemnification of Directors and Officers

The information set forth under Item 1.01 of this Current Report on Form 8-K is incorporated herein by reference.

Financial Statements and Exhibits

The information set forth under Item 9.01 of this Current Report on Form 8-K is incorporated herein by reference.

Changes in and Disagreements with Accountants on Accounting and Financial Disclosure

Not applicable.

Item 3.02. Unregistered Sales of Equity Securities.

Transaction Consideration

The information set forth in the "Introductory Note-Subscription Agreements" above is incorporated into this Item 3.02 by reference.

The shares of Company Common Stock issued by the Company to the PIPE Investors in the private placement on the Closing Date were issued pursuant to and in accordance with the exemption from registration under the Securities Act under Section 4(a)(2) and/or Regulation D promulgated under the Securities Act.

Item 3.03. Material Modification to Rights of Security Holders.

Upon the consummation of the Business Combination, the Company adopted the Company Charter. The material terms of the Company Charter and the general effect upon the rights of holders of the Company's capital stock are discussed in the Proxy in the sections titled "The Organizational Documents Proposal" beginning on page 141, and "The Domestication Proposal — Comparison of Shareholder Rights under Applicable Corporate Law Before and After Domestication" beginning on page 134, which are incorporated by reference herein.

As disclosed below in Item 8.01, in accordance with Rule 12g-3(a) under the Securities Exchange Act, the Company is the successor issuer to ION and has succeeded to the attributes of ION as the registrant. In addition, the shares of Company Common Stock, as the successor to ION, are deemed to be registered under Section 12(b) of the Exchange Act.

Amended and Restated Certificate of Incorporation (Company Charter)

Upon consummation of the Business Consummation, ION's amended and restated certificate of incorporation, was replaced with the Company's Charter, which, among other things:

- (a) changed the authorized capital stock of ION from 500,00,000 ION Class A Ordinary Shares, 50,000,000 ION Class B Ordinary Shares and 5,000,000 ION preference shares to 500,000,000 shares of Company Common Stock and 500,000 Innovid preference shares;
- (b) authorizes the provisions permitting that the number of directors on the Board be fixed from time to time solely by resolution of the Board, and dividing the Board into three classes;
 - (c) amends the terms for the authorizations providing for the increase and decrease of the Company's capital stock; and
 - (d) provides for certain additional changes, including, among other things, removing certain provisions related to ION's status as a blank check company.

The shareholders of ION approved these amendments at the Special Meeting. This summary is qualified in its entirety by reference to the text of the second amended and restated certificate of incorporation, which is included as Exhibit 3.1 hereto and incorporated herein by reference.

Amended and Restated Bylaws

Upon the closing of the Business Combination, ION's bylaws were amended and restated to be consistent with the Company Charter and to make certain other changes that Company's board of directors deemed appropriate for a public operating company. This summary is qualified in its entirety by reference to the text of the amended and restated bylaws, which is included as Exhibit 3.2 hereto and incorporated herein by reference.

Item 4.01. Change in Registrant's Certifying Accountant.

Not applicable.

Item 5.01. Changes in Control of Registrant.

Reference is made to the disclosure in the Proxy in the section titled "The Business Combination Proposal," which is incorporated herein by reference. Further reference is made to the information contained in Item 2.01 to this Current Report on Form 8-K, which is incorporated herein by reference.

Immediately after giving effect to the Business Combination, there were 118,941,618 shares of Company Common Stock outstanding, 7,060,000 Public Warrants outstanding and 3,162,500 Private Placement Warrants

outstanding. As of such time, Company's executive officers and directors and their affiliated entities held approximately 10% of the total voting power of Company's outstanding capital stock. Following the Closing, the PIPE Investors held approximately 13% of the Company's voting power, the former ION shareholders held approximately 6% of the Company's voting power, excluding in each case outstanding warrants.

Item 5.02. Departure of Directors or Certain Officers; Election of Directors; Appointment of Certain Officers; Compensatory Arrangements of Certain Officers.

Upon the Closing, and in accordance with the terms of the Merger Agreement, the following six (6) individuals, who will take office immediately (other than Gilad Shany, an existing director) and will constitute all the members of the Company's Board of Directors: (i) Jonathan Saacks and Steven Cakebread as Class I directors, (ii) Gilad Shany and Rachel Lam as Class II directors, and (iii) Zvika Netter and Michael DiPiano as Class III directors.

Upon the consummation of the Business Combination, the Company established the following three committees of the Board: audit committee, compensation committee and nominating and corporate governance committee. Steven Cakebread, Rachel Lam and Michael DiPiano were appointed to serve on the Company's audit committee, with Steven Cakebread serving as the chair and qualifying as an audit committee financial expert, as such term is defined in Item 407(d)(5) of Regulation S-K. Gilad Shany and Michael DiPiano were appointed to serve on the Company's compensation committee. Gilad Shany, Rachel Lam and Jonathan Saacks were appointed to serve on the Company's nominating and corporate governance committee.

Additionally, upon consummation of the Business Combination, Zvika Netter was appointed as the Company's President and Chief Executive Officer; Tanya Andreev-Kaspin was appointed as Chief Financial Officer; and Tal Chalozin was appointed as Chief Technology Officer.

Reference is made to the disclosure described in the Proxy in the section titled "Management of the Company Following the Business Combination" beginning on page 268 for biographical information about each of the directors and officers following the Transactions, which is incorporated herein by reference.

The information set forth under Item 1.01. "Entry into a Material Definitive Agreement—Indemnification Agreement", "— Innovid Corp. Incentive Award Plan" and "— Innovid Corp. Employee Stock Purchase Plan" of this Current Report on Form 8-K is incorporated herein by reference.

The information set forth under Item 2.01. "Completion of Acquisition or Disposition of Assets – Director Compensation" of this Current Report on Form 8-K is incorporated herein by reference.

Item 5.03. Amendments to Articles of Incorporation or Bylaws; Change in Fiscal Year.

The disclosure set forth in Item 3.03 of this Current Report on Form 8-K is incorporated in this Item 5.03 by reference.

Item 5.06. Change in Shell Company Status.

As a result of the Business Combination, which fulfilled the definition of an "initial business combination" as required by ION's organizational documents, ION ceased to be a shell company upon the closing of the Business Combination. The material terms of the Business Combination are described in the section titled "The Business Combination Proposal" beginning on page 100, of the Proxy, and is incorporated herein by reference.

Item 8.01. Other Events.

By operation of Rule 12g-3(a) under the Exchange Act, the Company is the successor issuer to ION and has succeeded to the attributes of ION as the registrant, including ION's SEC file number (001-40048) and CIK Code (0001835378). The Company's Common Stock and Public Warrants are deemed to be registered under Section 12(b) of the Exchange Act, and the Company will hereafter file reports and other information with the SEC using ION's SEC file number (001-40048).

Company's Common Stock and Public Warrants are listed for trading on The New York Stock Exchange under the symbols "CTV" and "CTVWS," respectively, and the CUSIP numbers relating to the Company's Common Stock and Public Warrants are 457679108 and 457679116, respectively.

Holders of uncertificated shares of ION's common stock immediately prior to the Business Combination have continued as holders of shares of uncertificated shares of Company Common Stock.

Holders of ION's shares who have filed reports under the Exchange Act with respect to those shares should indicate in their next filing, or any amendment to a prior filing, filed on or after November 30, 2021 ("the <u>Closing Date"</u>) that the Company is the successor to ION.

Item 9.01. Financial Statements and Exhibits.

(a) Financial statements of businesses acquired.

The unaudited condensed consolidated financial statements of OldCo, as of June 30, 2021, and for the six months ended June 30, 2021 and 2020, and the consolidated financial statements of OldCo. as of and for the years ended December 31, 2020 and 2019, the related notes and the report of independent registered public accounting firm thereto are set forth in the Proxy beginning on page F-36 and are incorporated herein by reference.

The unaudited condensed consolidated financial statements of OldCo. as of September 30, 2021 and for the nine months ended September 30, 2021 and 2020 and the related notes thereto are filed herewith as Exhibit 99.1 and incorporated herein by reference.

(b) Pro forma financial information.

Certain pro forma financial information of the Company as of and for the nine months ended September 30, 2021 and for the year ended December 31, 2020 is attached hereto as Exhibit 99.2 and is incorporated herein by reference.

(c) Financial statements of ION.

The financial statements of ION as of December 31, 2020 and for the period from November 23, 2020 (inception) through December 31, 2020, and the related notes and the report of independent registered public accounting firm thereto are set forth in the Proxy beginning on page F-2 and are incorporated herein by reference. The unaudited condensed consolidated financial statements of ION as of September 30, 2021 and for the nine months ended September 30, 2021 and the related notes thereto with respect to such dates and period are set forth in the ION 10-Q and are incorporated herein by reference.

(d) Exhibits.

Exhibit Number	Description
2.1†	Agreement and Plan of Merger, dated June 24, 2021, by and among ION, Merger Sub 1, Merger Sub 2, and OldCo (incorporated by reference to Exhibit 2.1 of ION's Current Report on Form 8-K filed with the SEC on June 29, 2021).
3.1	Amended and Restated Certificate of Incorporation of ION Acquisition Corp 2.
3.2	Bylaws of Innovid Corp.
4.1	Specimen Common Stock Certificate of Innovid Corp.
4.2	Specimen Warrant Certificate of Innovid Corp.
4.3	Warrant Agreement, dated February 10, 2021, by and between ION and Continental Stock Transfer & Trust Company, as warrant agent (incorporated by reference to Exhibit 4.1 of ION's Form 8-K (File No. 333-252440), filed with the SEC on February 18, 2021).

10.1	Letter Agreement, dated February 10, 2021, by and among ION, ION Holdings 2, LP, and each of the officers and directors of ION (incorporated by reference to Exhibit 10.1 of ION's Form 8-K (File No. 333-252440), filed with the SEC on February 18, 2021).
10.2	Investor Rights Agreement, dated November 30, 2021, by and among the Company, the SPAC Holders, and the OldCo Equity Holders.
10.3	Form of Innovid Corp. Incentive Plan (incorporated by reference to Annex E to Amendment No. 3 to the Registration Statement on Form S-4 (File No. 333-258472), filed with the SEC on November 5, 2021).
10.4	Form of Innovid Corp. Employee Stock Purchase Plan (incorporated by reference to Annex D to Amendment No. 3 to the Registration Statement on Form S-4 (File No. 333-258472), filed with the SEC on November 5, 2021).
10.5	Sponsor Support Agreement, dated as of June 24, 2021, by and amongOldCo, ION, Sponsor and the other parties thereto (incorporated by reference to Annex G to Amendment No. 3 to the Registration Statement on Form S-4 (File No. 333-258472), filed with the SEC on November 5, 2021)
10.6	Form of Subscription Agreement between ION and the PIPE Investors, dated June 24, 2021 (incorporated by reference to Exhibit 10.1 of ION's Form 8-K (File No. 333-252440), filed with the SEC on June 29, 2021).
10.7	Innovid Corp. Non-employee Director Compensation program.
10.8	Form of Indemnification Agreement with Executive Officers and Directors of Innovid Corp.
21.1	List of Subsidiaries of the Company.
99.1	Unaudited condensed financial statements of Innovid, Inc. for the nine months ended September 30, 2021 and 2020
99.2	Unaudited pro forma condensed combined financial information of ION Acquisition Corp. 2 Ltd. and Innovid, Inc. for the nine months ended September 30, 2021.
99.3	Management's Discussion and Analysis of Financial Condition and Results of Operations for Innovid, Inc. for the nine months ended September 30, 2021.

The annexes, schedules, and certain exhibits to this Exhibit have been omitted pursuant to Item 601(b)(2) of Regulation S-K. The Registrant hereby agrees to furnish supplementally a copy of any omitted annex, schedule or exhibit to the SEC upon request.

Certain confidential information contained in this Exhibit has been omitted because it is (i) not material and (ii) of the type that the registrant treats as private or confidential.

SIGNATURES

Pursuant to the requirements of the Securities Exchange Act of 1934, the registrant has duly caused this report to be signed on its behalf by the undersigned hereunto duly authorized.

INNOVID CORP.

By: /s/ Tanya Andreev-Kaspin

Name: Tanya Andreev-Kaspin Title: Chief Financial Officer

Date: December 6, 2021

AMENDED AND RESTATED CERTIFICATE OF INCORPORATION OF ION ACQUISITION CORP 2

ION Acquisition Corp 2 (the "Corporation"), a corporation organized and existing under the General Corporation Law of the State of Delaware (the "DGCL"), does hereby certify as follows:

- 1. The name of the Corporation is ION Acquisition Corp 2. The Corporation was incorporated under the name ION Acquisition Corp 2 by the filing of its original Certificate of Incorporation with the Secretary of State of the State of Delaware on November 29, 2021.
- 2. This Amended and Restated Certificate of Incorporation (this "Restated Certificate"), which amends, restates and further integrates the certificate of incorporation of the Corporation as heretofore in effect, has been approved by the Board of Directors of the Corporation (the "Board of Directors") in accordance with Sections 242 and 245 of the DGCL, and has been adopted by the written consent of the stockholders of the Corporation in accordance with Section 228 of the DGCL.
- 3. The text of the certificate of incorporation of the Corporation, as heretofore amended, is hereby amended and restated by this Restated Certificate to read in its entirety as set forth in EXHIBIT A attached hereto.

IN WITNESS WHEREOF, the Corporation has caused this Restated Certificate to be signed by a duly authorized officer of the Corporation, on November 30, 2021.

ION ACQUISITION CORP 2, a Delaware corporation

By:

Name: Anthony Reich Title: Chief Financial Officer

[Signature Page to Amended and Restated Certificate of Incorporation]

EXHIBIT A

ARTICLE I

The name of the corporation is Innovid Corp. (the "Corporation").

ARTICLE II

The address of the Corporation's registered office in the State of Delaware is The Corporation Trust Company, 1209 Orange Street, Corporation Trust Center, Wilmington, Delaware 19801, County of New Castle, and the name of its registered agent at such address is The Corporation Trust Company.

ARTICLE III

The purpose of the Corporation is to engage in any lawful act or activity for which corporations may be organized under the General Corporation Law of the State of Delaware (the "<u>DGCL</u>") as it now exists or may hereafter be amended and supplemented.

ARTICLE IV

The Corporation is authorized to issue two classes of stock to be designated, respectively, "Common Stock" and "Preferred Stock." The total number of shares of capital stock which the Corporation shall have authority to issue is 500,500,000. The total number of shares of Common Stock that the Corporation is authorized to issue is 500,000,000, having a par value of \$0.0001 per share, and the total number of shares of Preferred Stock that the Corporation is authorized to issue is 500,000, having a par value of \$0.0001 per share.

ARTICLE V

The designations and the powers, preferences, privileges and rights, and the qualifications, limitations or restrictions thereof in respect of each class of capital stock of the Corporation are as follows:

A. COMMON STOCK.

- 1. <u>General</u>. The voting, dividend, liquidation, and other rights and powers of the Common Stock are subject to and qualified by the rights, powers and preferences of any series of Preferred Stock as may be designated by the Board of Directors of the Corporation (the "Board of Directors") and outstanding from time to time.
- 2. <u>Voting.</u> Except as otherwise provided herein or expressly required by law, at all meetings of stockholders and on all matters submitted to a vote of stockholders of the Corporation generally, each holder of Common Stock, as such, shall be entitled to vote on each matter submitted to a vote of stockholders and shall be entitled to one (1) vote for each share of Common Stock held of record by such holder as of the record date for determining stockholders entitled to vote on such matter. Except as otherwise required by law, holders of Common Stock, as such, shall not be entitled to vote on any amendment to this Restated Certificate (including any Certificate of Designation (as defined below)) that relates solely to the rights, powers, preferences (or the qualifications, limitations or restrictions thereof) or other terms of one or

more outstanding series of Preferred Stock if the holders of such affected series are entitled, either separately or together with the holders of one or more other such series, to vote thereon pursuant to this Restated Certificate (including any Certificate of Designation) or pursuant to the DGCL.

Subject to the rights of any holders of any outstanding series of Preferred Stock, the number of authorized shares of Common Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the stock of the Corporation entitled to vote, irrespective of the provisions of Section 242(b)(2) of the DGCL.

- 3. <u>Dividends</u>. Subject to applicable law and the rights and preferences of any holders of any outstanding series of Preferred Stock, the holders of Common Stock, as such, shall be entitled to the payment of dividends on the Common Stock when, as and if declared by the Board of Directors in accordance with applicable law.
- 4. <u>Liquidation</u>. Subject to the rights and preferences of any holders of any shares of any outstanding series of Preferred Stock, in the event of any liquidation, dissolution or winding up of the Corporation, whether voluntary or involuntary, the funds and assets of the Corporation that may be legally distributed to the Corporation's stockholders shall be distributed among the holders of the then outstanding Common Stock <u>pro rata</u> in accordance with the number of shares of Common Stock held by each such holder.

B. PREFERRED STOCK

Shares of Preferred Stock may be issued from time to time in one or more series, each of such series to have such terms as stated or expressed herein and in the resolution or resolutions providing for the creation and issuance of such series adopted by the Board of Directors as hereinafter provided. Any share of Preferred Stock which may be redeemed, purchased or acquired by the Corporation may be reissued except as otherwise provided by law.

Authority is hereby expressly granted to the Board of Directors from time to time to issue the Preferred Stock in one or more series, and in connection with the creation of any such series, by adopting a resolution or resolutions providing for the issuance of the shares thereof and by filing a certificate of designation relating thereto in accordance with the DGCL (a "Certificate of Designation"), to determine and fix the number of shares of such series and such voting powers, full or limited, or no voting powers, and such designations, preferences and relative participating, optional or other special rights, and qualifications, limitations or restrictions thereof, including without limitation thereof, dividend rights, conversion rights, redemption privileges and liquidation preferences, and to increase or decrease (but not below the number of shares of such series then outstanding) the number of shares of any series as shall be stated and expressed in such resolutions, all to the fullest extent now or hereafter permitted by the DGCL. Without limiting the generality of the foregoing, the resolution or resolutions providing for the creation and issuance of any series of Preferred Stock may provide that such series shall be superior or rank equally or be junior to any other series of Preferred Stock to the extent permitted by law and this Restated Certificate (including any Certificate of Designation). Except as otherwise required by law, holders of any series of Preferred Stock shall be entitled

only to such voting rights, if any, as shall expressly be granted thereto by this Restated Certificate (including any Certificate of Designation).

The number of authorized shares of Preferred Stock may be increased or decreased (but not below the number of shares thereof then outstanding) by the affirmative vote of the holders of a majority of the stock of the Corporation entitled to vote, irrespective of the provisions of Section 242(b)(2) of the DGCL.

ARTICLE VI

For the management of the business and for the conduct of the affairs of the Corporation it is further provided that:

- A. <u>General Powers</u>. Except as otherwise expressly provided by the DGCL or this Restated Certificate, the business and affairs of the Corporation shall be managed by or under the direction of the Board of Directors.
- B. <u>Number of Directors</u>. The number of directors which shall constitute the whole Board of Directors shall be fixed exclusively by one or more resolutions adopted from time to time by the Board of Directors.
- C. <u>Classes of Directors</u>. The directors of the Corporation shall be classified with respect to the time for which they severally hold office into three classes, designated as Class I, Class II and Class III. Each class shall consist, as nearly as possible, of one-third of the total number of the whole Board of Directors. The initial Class I directors shall serve for a term expiring at the first annual meeting of the stockholders following the filing and effectiveness of this Restated Certificate with the Secretary of State of the State of Delaware (the "<u>Effective Time</u>"); the initial Class III directors shall serve for a term expiring at the second annual meeting of the stockholders following the Effective Time; and the initial Class III directors shall serve for a term expiring at the third annual meeting following the Effective Time. At each annual meeting of stockholders of the Corporation beginning with the first annual meeting of stockholders following the Effective Time, subject to any special rights of the holders of one or more outstanding series of Preferred Stock to elect directors, the successors of the class of directors whose term expires at that meeting shall be elected to hold office for a term expiring at the annual meeting of stockholders held in the third year following the year of their election. Each director shall hold office until his or her successor is duly elected and qualified or until his or her earlier death, resignation, disqualification or removal. No decrease in the number of directors shall shorten the term of any incumbent director. The Board of Directors is authorized to assign members of the Board of Directors already in office to Class I, Class III and Class III.
- D. <u>Term and Removal</u>. The Board of Directors or any individual director may be removed from office at any time, but only for cause and only by the affirmative vote of the holders of at least two-thirds of the voting power of all of the then outstanding shares of voting stock of the Corporation entitled to vote at an election of directors.
- E. <u>Vacancies and Newly Created Directorships</u>. Except as otherwise provided by law, any vacancies on the Board of Directors resulting from death, resignation, disqualification, retirement, removal or other causes and any newly created directorships

resulting from any increase in the number of directors shall be filled exclusively by the affirmative vote of a majority of the directors then in office, even if less than a quorum, or by a sole remaining director (other than any directors elected by the separate vote of one or more outstanding series of Preferred Stock), and shall not be filled by the stockholders. Any director appointed in accordance with the preceding sentence shall hold office until the expiration of the term of the class to which such director shall have been appointed or until his or her earlier death, resignation, retirement, disqualification, or removal.

- F. <u>Amendments</u>. In furtherance and not in limitation of the powers conferred by statute, the Board of Directors is expressly authorized to adopt, amend or repeal the Bylaws of the Corporation. In addition to any vote of the holders of any class or series of stock of the Corporation required by applicable law or by this Restated Certificate (including any Certificate of Designation in respect of one or more series of Preferred Stock) or the Bylaws of the Corporation, the adoption, amendment or repeal of the Bylaws of the Corporation by the stockholders of the Corporation shall require the affirmative vote of the holders of at least two- thirds of the voting power of all of the then outstanding shares of voting stock of the Corporation entitled to vote generally in an election of directors.
 - G. Vote by Ballot. The directors of the Corporation need not be elected by written ballot unless the Bylaws so provide.

ARTICLE VII

The Corporation renounces any interest or expectancy of the Corporation in, or in being offered an opportunity to participate in, any Excluded Opportunity. An "Excluded Opportunity" is any matter, transaction or interest that is presented to, or acquired, created or developed by, or which otherwise comes into the possession of, any director of the Corporation who is not an employee or officer of the Corporation or any of its subsidiaries (a "Covered Person"), unless such matter, transaction or interest is presented to, or acquired, created or developed by, or otherwise comes into the possession of, a Covered Person expressly and solely in such Covered Person's capacity as a director of the Corporation.

ARTICLE VIII

A. <u>Consent of Stockholders In Lieu of Meeting.</u> Any action required or permitted to be taken by the stockholders of the Corporation must be effected at an annual or special meeting of the stockholders of the Corporation, and shall not be taken by consent in lieu of a meeting. Notwithstanding the foregoing, any action required or permitted to be taken by the holders of any series of Preferred Stock, voting separately as a series or separately as a class with one or more other such series, may be taken without a meeting, without prior notice and without a vote, to the extent expressly so provided by the applicable Certificate of Designation relating to such series of Preferred Stock, if a consent or consents in writing or by electronic transmission, setting forth the action so taken, shall be signed by the holders of outstanding shares of the relevant series of Preferred Stock having not less than the minimum number of votes that would be necessary to authorize or take such action at a meeting at which all shares entitled to vote thereon were present and voted and shall be delivered to the Corporation in accordance with the applicable provisions of the DGCL.

- B. <u>Special Meeting of Stockholders</u>. Subject to the special rights of the holders of one or more series of Preferred Stock, special meetings of the stockholders of the Corporation may be called, for any purpose or purposes, at any time only by or at the direction of the Board of Directors, the Chairperson of the Board of Directors, the Chief Executive Officer or President, in each case, in accordance with the Bylaws, and shall not be called by any other person or persons.
- C. <u>Stockholder Nominations and Introduction of Business, Etc.</u> Advance notice of stockholder nominations for the election of directors and of other business proposed to be brought by stockholders before any meeting of the stockholders of the Corporation shall be given in the manner provided in the Bylaws of the Corporation. Any business transacted at any special meeting of stockholders shall be limited to matters relating to the purpose or purposes identified in the notice of meeting.

ARTICLE IX

No director of the Corporation shall have any personal liability to the Corporation or its stockholders for monetary damages for any breach of fiduciary duty as a director, except to the extent such exemption from liability or limitation thereof is not permitted under the DGCL as the same exists or hereafter may be amended. Any amendment, repeal or modification of this Article IX, or the adoption of any provision of this Restated Certificate inconsistent with this Article IX, shall not adversely affect any right or protection of a director of the Corporation with respect to any act or omission occurring prior to such amendment, repeal, modification or adoption. If the DGCL is amended after approval by the stockholders of this Article IX to authorize corporate action further eliminating or limiting the personal liability of directors, then the liability of a director of the Corporation shall be eliminated or limited to the fullest extent permitted by the DGCL as so amended.

ARTICLE X

The Corporation shall have the power to provide rights to indemnification and advancement of expenses to its current and former officers, directors, employees and agents and to any person who is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust or other enterprise.

ARTICLE XI

Unless the Corporation consents in writing to the selection of an alternative forum, (a) the Court of Chancery (the "Chancery Court") of the State of Delaware (or, in the event that the Chancery Court does not have jurisdiction, the federal district court for the District of Delaware or other state courts of the State of Delaware) shall, to the fullest extent permitted by law, be the sole and exclusive forum for (i) any derivative action, suit or proceeding brought on behalf of the Corporation, (ii) any action, suit or proceeding asserting a claim of breach of a fiduciary duty owed by any current or former director, officer, other employee or stockholder of the Corporation to the Corporation or to the Corporation's stockholders, (iii) any action, suit or proceeding arising pursuant to any provision of the DGCL or the Bylaws of the Corporation or this Restated Certificate (as either may be amended from time to time) or as to which the DGCL

confers jurisdiction on the Chancery Court or (iv) any action, suit or proceeding asserting a claim against the Corporation governed by the internal affairs doctrine; and (b) subject to the preceding provisions of this Article XI, the federal district courts of the United States of America shall be the exclusive forum for the resolution of any complaint asserting a cause of action arising under the Securities Act of 1933, as amended. If any action the subject matter of which is within the scope of clause (a) of the immediately preceding sentence is filed in a court other than the courts in the State of Delaware (a "Foreign Action") in the name of any stockholder, such stockholder shall be deemed to have consented to (x) the personal jurisdiction of the state and federal courts in the State of Delaware in connection with any action brought in any such court to enforce the provisions of clause (a) of the immediately preceding sentence and (y) having service of process made upon such stockholder in any such action by service upon such stockholder's counsel in the Foreign Action as agent for such stockholder.

To the fullest extent permitted by law, any person or entity purchasing or otherwise acquiring or holding any interest in any security of the Corporation shall be deemed to have notice of and consented to this Article XI. Notwithstanding the foregoing, the provisions of this Article XI shall not apply to suits brought to enforce any liability or duty created by the Securities Exchange Act of 1934, as amended, or any other claim for which the federal courts of the United States have exclusive jurisdiction.

If any provision or provisions of this Article XI shall be held to be invalid, illegal or unenforceable as applied to any circumstance for any reason whatsoever, (a) the validity, legality and enforceability of such provisions in any other circumstance and of the remaining provisions of this Article XI (including, without limitation, each portion of any paragraph of this Article XI containing any such provision held to be invalid, illegal or unenforceable that is not itself held to be invalid, illegal or unenforceable) shall not in any way be affected or impaired thereby and (b) the application of such provision to other persons or entities and circumstances shall not in any way be affected or impaired thereby.

ARTICLE XII

- A. <u>Amendment</u>. Notwithstanding anything contained in this Restated Certificate to the contrary, in addition to any vote required by applicable law, the following provisions in this Restated Certificate may be amended, altered, repealed or rescinded, in whole or in part, or any provision inconsistent therewith or herewith may be adopted, only by the affirmative vote of the holders of at least 66 2/3% of the total voting power of all the then outstanding shares of stock of the Corporation entitled to vote thereon, voting together as a single class: Part B of Article V, Article VI, Article VII, Article IX, Article XI, and this Article XII.
- B. <u>Severability</u>. If any provision or provisions of this Restated Certificate shall be held to be invalid, illegal or unenforceable as applied to any circumstance for any reason whatsoever: (i) the validity, legality and enforceability of such provisions in any other circumstance and of the remaining provisions of this Restated Certificate (including, without limitation, each portion of any paragraph of this Restated Certificate containing any such provision held to be invalid, illegal or unenforceable that is not itself held to be invalid, illegal or unenforceable) shall not, to the fullest extent permitted by applicable law, in any way be affected or impaired thereby and (ii) to the fullest extent permitted by applicable law, the provisions

of this Restated Certificate (including, without limitation, each such portion of any paragraph of this Restated Certificate containing any such provision held to be invalid, illegal or unenforceable) shall be construed so as to permit the Corporation to protect its directors, officers, employees and agents from personal liability in respect of their good faith service to or for the benefit of the Corporation to the fullest extent permitted by law.

Amended and Restated Bylaws of Innovid Corp.

(a Delaware corporation)

TABLE OF CONTENTS

		Page
Artic	ele I - Corporate Offices	1
1.1	Registered Office	1
1.2	Other Offices	1
Artic	cle II - Meetings of Stockholders	1
2.1	Place of Meetings	1
2.2	Annual Meeting	1
2.3	Special Meeting	1
2.4	Notice of Business to be Brought Before a Meeting	1
2.5	Notice of Nominations for Election to the Board	5
2.6	Additional Requirements for Valid Nomination of Candidates to Serve as Director and, if Elected, to be Seated as Directors	7
2.7	Notice of Stockholders' Meetings	9
2.8	Quorum	9
2.9	Adjourned Meeting; Notice	9
2.10	Conduct of Business	10
2.11	Voting	10
2.12	Record Date for Stockholder Meetings and Other Purposes	10
2.13	Proxies	11
2.14	List of Stockholders Entitled to Vote	11
2.15	Inspectors of Election	12
2.16	Delivery to the Corporation	12
	cle III - Directors	13
3.1	Powers	13
3.2	Number of Directors	13
3.3	Election, Qualification and Term of Office of Directors	13
	Resignation; Vacancies; Removal	13
3.5	Place of Meetings; Meetings by Telephone	13
3.6	Regular Meetings	14
3.7	Special Meetings; Notice	14
	Quorum	14
3.9	Board Action Without a Meeting	15
	Fees and Compensation of Directors	15
	cle IV - Committees	15
4.1	Committees of Directors	15
	Committee Minutes	15
	Meetings and Actions of Committees	15
	Subcommittees	16
	ele V - Officers	16
	Officers	16
5.2	Appointment of Officers	16

5.3	Subordinate Officers	16
5.4	Removal and Resignation of Officers	16
5.5	Vacancies in Offices	17
5.6	Representation of Shares of Other Corporations	17
5.7	Authority and Duties of Officers	17
5.8	Compensation	17
Arti	icle VI - Records	17
Arti	icle VII - General Matters	18
7.1	Execution of Corporate Contracts and Instruments	18
7.2	Stock Certificates	18
7.3	Special Designation of Certificates	18
7.4	Lost Certificates	19
7.5	Shares Without Certificates	19
7.6	Construction; Definitions	19
7.7	Dividends	19
7.8	Fiscal Year	19
7.9	Seal	19
7.10	Transfer of Stock	19
7.11	Stock Transfer Agreements	20
7.12	2 Registered Stockholders	20
7.13	3 Waiver of Notice	20
Arti	icle VIII - Notice	20
8.1	Delivery of Notice; Notice by Electronic Transmission	20
Arti	icle IX - Indemnification	21
9.1	Indemnification of Directors and Officers	21
9.2	Indemnification of Others	22
9.3	Prepayment of Expenses	22
9.4	Determination; Claim	22
9.5	Non-Exclusivity of Rights	22
9.6	Insurance	22
9.7	Other Indemnification	23
9.8	Continuation of Indemnification	23
9.9	Amendment or Repeal; Interpretation	23
Arti	icle X - Amendments	24
Arti	icle XI - Definitions	24

Amended and Restated Bylaws of Innovid Corp.

Article I - Corporate Offices

1.1 Registered Office.

The address of the registered office of Innovid Corp. (the "<u>Corporation</u>") in the State of Delaware, and the name of its registered agent at such address, shall be as set forth in the Corporation's certificate of incorporation, as the same may be amended and/or restated from time to time (the "<u>Certificate of Incorporation</u>").

1.2 Other Offices.

The Corporation may have additional offices at any place or places, within or outside the State of Delaware, as the Corporation's board of directors (the "Board") may from time to time establish or as the business of the Corporation may require.

Article II - Meetings of Stockholders

2.1 <u>Place of Meetings</u>.

Meetings of stockholders shall be held at any place within or outside the State of Delaware, designated by the Board. The Board may, in its sole discretion, determine that a meeting of stockholders shall not be held at any place, but may instead be held solely by means of remote communication as authorized by Section 211(a)(2) of the General Corporation Law of the State of Delaware (the "<u>DGCL</u>"). In the absence of any such designation or determination, stockholders' meetings shall be held at the Corporation's principal executive office.

2.2 <u>Annual Meeting</u>.

The Board shall designate the date and time of the annual meeting. At the annual meeting, directors shall be elected and other proper business properly brought before the meeting in accordance with Section 2.4 of these Bylaws may be transacted. The Board may postpone, reschedule or cancel any previously scheduled annual meeting of stockholders.

2.3 Special Meeting.

Special meetings of the stockholders may be called only by such persons and only in such manner as set forth in the Certificate of Incorporation.

No business may be transacted at any special meeting of stockholders other than the business specified in the notice of such meeting. The Board may postpone, reschedule or cancel any previously scheduled special meeting of stockholders.

2.4 Notice of Business to be Brought Before a Meeting.

(a) At an annual meeting of the stockholders, only such business shall be conducted as shall have been properly brought before the meeting. To be properly brought before an annual meeting, business must be (i) specified in a notice of meeting given by or at the direction of the Board, (ii) if not specified in a notice of meeting, otherwise brought before the meeting by the Board or the Chairperson of the Board or (iii) otherwise properly brought before the meeting by

a stockholder present in person who (A) (1) was a record owner of shares of the Corporation both at the time of giving the notice provided for in this Section 2.4 and at the time of the meeting, (2) is entitled to vote at the meeting, and (3) has complied with this Section 2.4 in all applicable respects or (B) properly made such proposal in accordance with Rule 14a-8 under the Securities Exchange Act of 1934, as amended, and the rules and regulations thereunder (as so amended and inclusive of such rules and regulations, the "Exchange Act"). The foregoing clause (iii) shall be the exclusive means for a stockholder to propose business to be brought before an annual meeting of the stockholders. The only matters that may be brought before a special meeting are the matters specified in the notice of meeting given by or at the direction of the person calling the meeting pursuant to Section 2.7, and stockholders shall not be permitted to propose business to be brought before a special meeting of the stockholders. For purposes of this Section 2.4, "present in person" shall mean that the stockholder proposing that the business be brought before the annual meeting of the Corporation, or a qualified representative of such proposing stockholder, appear at such annual meeting. A "qualified representative" of such proposing stockholder shall be a duly authorized officer, manager or partner of such stockholder or any other person authorized by a writing executed by such stockholder or an electronic transmission delivered by such stockholder to act for such stockholder as proxy at the meeting of stockholders and such person must produce such writing or electronic transmission, or a reliable reproduction of the writing or electronic transmission, at the meeting of stockholders. Stockholders seeking to nominate persons for election to the Board must comply with Section 2.5 and Section 2.6 and this Section 2.4 shall not be applicable to nominations except as expressly provided in Section 2.5 and Section 2.6.

- (b) Without qualification, for business to be properly brought before an annual meeting by a stockholder, the stockholder must (i) provide Timely Notice (as defined below) thereof in writing and in proper form to the Secretary of the Corporation (the "Secretary") and (ii) provide any updates or supplements to such notice at the times and in the forms required by this Section 2.4. To be timely, a stockholder's notice must be delivered to, or mailed and received at, the principal executive offices of the Corporation not less than ninety (90) days nor more than one hundred twenty (120) days prior to the one-year anniversary of the preceding year's annual meeting which, in the case of the first annual meeting of stockholders following the closing of the Corporation's initial underwritten public offering of common stock, the date of the preceding year's annual meeting shall be deemed to be June 8, 2021; provided, however, that if the date of the annual meeting is more than thirty (30) days before or more than sixty (60) days after such anniversary date, notice by the stockholder to be timely must be so delivered, or mailed and received, not later than the ninetieth (90th) day prior to such annual meeting or, if later, the tenth (10th) day following the day on which public disclosure of the date of such annual meeting was first made by the Corporation (such notice within such time periods, "Timely Notice"). In no event shall any adjournment or postponement of an annual meeting or the announcement thereof commence a new time period (or extend any time period) for the giving of Timely Notice as described above.
 - (c) To be in proper form for purposes of this Section 2.4, a stockholder's notice to the Secretary shall set forth:
 - (i) As to each Proposing Person (as defined below), (A) the name and address of such Proposing Person (including, if applicable, the name and address that appear on the Corporation's books and records); and (B) the class or series and number of shares of the Corporation that are, directly or indirectly, owned of record or beneficially owned (within the meaning of Rule 13d-3 under the Exchange Act) by such Proposing Person, including any shares of any class or series of capital stock of the Corporation as to which such Proposing Person and such beneficial owner or any of its affiliates or associates has a right to acquire beneficial ownership at any time in the future, except that such Proposing Person shall in all events be deemed to beneficially own any shares of any

class or series of the Corporation as to which such Proposing Person has a right to acquire beneficial ownership at any time in the future (the disclosures to be made pursuant to the foregoing clauses (A) and (B) are referred to as "Stockholder Information");

As to each Proposing Person, (A) the full notional amount of any securities that, directly or indirectly, underlie any "derivative security" (as such term is defined in Rule 16a-1(c) under the Exchange Act) that constitutes a "call equivalent position" (as such term is defined in Rule 16a- 1(b) under the Exchange Act) ("Synthetic Equity Position") and that is, directly or indirectly, held or maintained by such Proposing Person with respect to any shares of any class or series of shares of the Corporation; provided that, for the purposes of the definition of "Synthetic Equity Position," the term "derivative security" shall also include any security or instrument that would not otherwise constitute a "derivative security" as a result of any feature that would make any conversion, exercise or similar right or privilege of such security or instrument becoming determinable only at some future date or upon the happening of a future occurrence, in which case the determination of the amount of securities into which such security or instrument would be convertible or exercisable shall be made assuming that such security or instrument is immediately convertible or exercisable at the time of such determination; and, provided, further, that any Proposing Person satisfying the requirements of Rule 13d-1(b)(1) under the Exchange Act (other than a Proposing Person that so satisfies Rule 13d-1(b)(1) under the Exchange Act solely by reason of Rule 13d-1(b)(1)(ii)(E)) shall not be deemed to hold or maintain the notional amount of any securities that underlie a Synthetic Equity Position held by such Proposing Person as a hedge with respect to a bona fide derivatives trade or position of such Proposing Person arising in the ordinary course of such Proposing Person's business as a derivatives dealer, (B) any rights to dividends on the shares of any class or series of shares of the Corporation owned beneficially by such Proposing Person that are separated or separable from the underlying shares of the Corporation, (C) any material pending or threatened legal proceeding in which such Proposing Person is a party or material participant involving the Corporation or any of its officers or directors, or any affiliate of the Corporation, (D) any other material relationship between such Proposing Person, on the one hand, and the Corporation, any affiliate of the Corporation, on the other hand, (E) any direct or indirect material interest in any material contract or agreement of such Proposing Person with the Corporation or any affiliate of the Corporation (including, in any such case, any employment agreement, collective bargaining agreement or consulting agreement), (F) a representation that such Proposing Person intends or is part of a group which intends to deliver a proxy statement or form of proxy to holders of at least the percentage of the Corporation's outstanding capital stock required to approve or adopt the proposal or otherwise solicit proxies from stockholders in support of such proposal, (G) any proxy (other than a revocable proxy given in response to a public proxy solicitation made pursuant to, and in accordance with, the Exchange Act), agreement, arrangement, understanding or relationship pursuant to which such Proposing Person or beneficial owner has or shares a right to, directly or indirectly, vote any shares of any class or series of capital stock of the Corporation, (H) any rights to dividends or other distributions on the shares of any class or series of capital stock of the Corporation, directly or indirectly, owned beneficially by such stockholder or beneficial owner that are separated or separable from the underlying shares of the Corporation, and (I) any performance-related fees (other than an asset based fee) that such stockholder or beneficial owner, directly or indirectly, is entitled to based on any increase or decrease in the value of shares of any class or series of capital stock of the Corporation or any Synthetic Equity Position and (J) any other information relating to such Proposing Person that would be required to be disclosed in a proxy statement or other filing required to be made in connection with solicitations of proxies or consents by such Proposing Person in support of the business proposed to be brought before the meeting pursuant to Section 14(a) of the Exchange Act

(the disclosures to be made pursuant to the foregoing clauses (A) through (J) are referred to as "Disclosable Interests"); provided, however, that Disclosable Interests shall not include any such disclosures with respect to the ordinary course business activities of any broker, dealer, commercial bank, trust company or other nominee who is a Proposing Person solely as a result of being the stockholder directed to prepare and submit the notice required by these Bylaws on behalf of a beneficial owner; and

As to each item of business that the stockholder proposes to bring before the annual meeting, (A) a brief description of the business desired to be brought before the annual meeting, the reasons for conducting such business at the annual meeting and any material interest in such business of each Proposing Person, (B) the text of the proposal or business (including the text of any resolutions proposed for consideration and in the event that such business includes a proposal to amend the Bylaws, the language of the proposed amendment), and (C) a reasonably detailed description of all agreements, arrangements and understandings (x) between or among any of the Proposing Persons or (y) between or among any Proposing Person and any other record or beneficial holder(s) or persons(s) who have a right to acquire beneficial ownership at any time in the future of the shares of any class or series of the Corporation or any other person or entity (including their names) in connection with the proposal of such business by such stockholder, including any agreements, arrangements or understandings relating to any compensation or payments to be paid to any such proposed nominee(s), pertaining to the nomination(s) or other business proposed to be brought before the meeting of stockholders (which description shall identify the name of each other person who is party to such an agreement, arrangement or understanding; and (D) any other information relating to such item of business that would be required to be disclosed in a proxy statement or other filing required to be made in connection with solicitations of proxies in support of the business proposed to be brought before the meeting pursuant to Section 14(a) of the Exchange Act; provided, however, that the disclosures required by this paragraph (iii) shall not include any disclosures with respect to any broker, dealer, commercial bank, trust company or other nominee who is a Proposing Person solely as a result of being the stockholder directed to prepare and submit the notice required by these Bylaws on behalf of a beneficial owner.

For purposes of this Section 2.4, the term "<u>Proposing Person</u>" shall mean (i) the stockholder providing the notice of business proposed to be brought before an annual meeting, (ii) the beneficial owner or beneficial owners, if different, on whose behalf the notice of the business proposed to be brought before the annual meeting is made, and (iii) any participant (as defined in paragraphs (a)(ii)-(vi) of Instruction 3 to Item 4 of Schedule 14A) with such stockholder in such solicitation.

(d) A Proposing Person shall update and supplement its notice to the Corporation of its intent to propose business at an annual meeting, if necessary, so that the information provided or required to be provided in such notice pursuant to this Section 2.4 shall be true and correct as of the record date for stockholders entitled to vote at the meeting and as of the date that is ten (10) business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to, or mailed and received by, the Secretary at the principal executive offices of the Corporation not later than five (5) business days after the record date for stockholders entitled to vote at the meeting (in the case of the update and supplement required to be made as of such record date), and not later than eight (8) business days prior to the date for the meeting or, if practicable, any adjournment or postponement thereof (and, if not practicable, on the first practicable date prior to the date to which the meeting has been adjourned or postponed) (in the case of the update and supplement required to be made as of ten (10) business days prior to the meeting or any adjournment or postponement thereof). For the avoidance of doubt, the obligation to update and supplement as set forth in this paragraph or

any other Section of these Bylaws shall not limit the Corporation's rights with respect to any deficiencies in any notice provided by a stockholder, extend any applicable deadlines hereunder or enable or be deemed to permit a stockholder who has previously submitted notice hereunder to amend or update any proposal or to submit any new proposal, including by changing or adding matters, business or resolutions proposed to be brought before a meeting of the stockholders.

- (e) Notwithstanding anything in these Bylaws to the contrary, no business shall be conducted at an annual meeting that is not properly brought before the meeting in accordance with this Section 2.4. The presiding officer of the meeting shall, if the facts warrant, determine that the business was not properly brought before the meeting in accordance with this Section 2.4, and if he or she should so determine, he or she shall so declare to the meeting and any such business not properly brought before the meeting shall not be transacted.
- (f) This Section 2.4 is expressly intended to apply to any business proposed to be brought before an annual meeting of stockholders other than any proposal made in accordance with Rule 14a-8 under the Exchange Act and included in the Corporation's proxy statement. In addition to the requirements of this Section 2.4 with respect to any business proposed to be brought before an annual meeting, each Proposing Person shall comply with all applicable requirements of the Exchange Act with respect to any such business. Nothing in this Section 2.4 shall be deemed to affect the rights of stockholders to request inclusion of proposals in the Corporation's proxy statement pursuant to Rule 14a-8 under the Exchange Act.
- (g) For purposes of these Bylaws, "public disclosure" shall mean disclosure in a press release reported by a national news service or in a document publicly filed by the Corporation with the Securities and Exchange Commission pursuant to Sections 13, 14 or 15(d) of the Exchange Act or by such other means as is reasonably designed to inform the public or stockholders of the Corporation in general of such information including, without limitation, posting on the Corporation's investor relations website.

2.5 <u>Notice of Nominations for Election to the Board.</u>

- (a) Nominations of any person for election to the Board at an annual meeting or at a special meeting (but only if the election of directors is a matter specified in the notice of meeting given by or at the direction of the person calling such special meeting) may be made at such meeting only (i) by or at the direction of the Board, including by any committee or persons authorized to do so by the Board or these Bylaws, or (ii) by a stockholder present in person (A) who was a record owner of shares of the Corporation both at the time of giving the notice provided for in this Section 2.5 and at the time of the meeting, (B) is entitled to vote at the meeting, and (C) has complied with this Section 2.5 and Section 2.6 as to such notice and nomination. For purposes of this Section 2.5, "present in person" shall mean that the stockholder proposing that the business be brought before the meeting of the Corporation, or a qualified representative of such stockholder, appear at such meeting. A "qualified representative" of such proposing stockholder shall be a duly authorized officer, manager or partner of such stockholder or any other person authorized by a writing executed by such stockholder or an electronic transmission delivered by such stockholder to act for such stockholder as proxy at the meeting of stockholders and such person must produce such writing or electronic transmission, or a reliable reproduction of the writing or electronic transmission, at the meeting of stockholders. The foregoing clause (ii) shall be the exclusive means for a stockholder to make any nomination of a person or persons for election to the Board at an annual meeting or special meeting.
- (b) (i) Without qualification, for a stockholder to make any nomination of a person or persons for election to the Board at an annual meeting, the stockholder must (1) provide Timely Notice (as defined in Section 2.4) thereof in writing and in proper form to the Secretary, (2) provide the information, agreements and questionnaires with respect to such

stockholder and its candidate for nomination as required to be set forth by this Section 2.5 and Section 2.6 and (3) provide any updates or supplements to such notice at the times and in the forms required by this Section 2.5 and Section 2.6.

- (ii) Without qualification, if the election of directors is a matter specified in the notice of meeting given by or at the direction of the person calling a special meeting, then for a stockholder to make any nomination of a person or persons for election to the Board at a special meeting, the stockholder must (i) provide timely notice thereof in writing and in proper form to the Secretary at the principal executive offices of the Corporation, (ii) provide the information with respect to such stockholder and its candidate for nomination as required by this Section 2.5 and Section 2.6 and (iii) provide any updates or supplements to such notice at the times and in the forms required by this Section 2.5. To be timely, a stockholder's notice for nominations to be made at a special meeting must be delivered to, or mailed and received at, the principal executive offices of the Corporation not earlier than the one hundred twentieth (120th) day prior to such special meeting and not later than the ninetieth (90th) day prior to such special meeting or, if later, the tenth (10th) day following the day on which public disclosure (as defined in Section 2.4) of the date of such special meeting was first made.
- (iii) In no event shall any adjournment or postponement of an annual meeting or special meeting or the announcement thereof commence a new time period or extend any time period for the giving of a stockholder's notice as described above.
- (iv) In no event may a Nominating Person provide Timely Notice with respect to a greater number of director candidates than are subject to election by stockholders at the applicable meeting. If the Corporation shall, subsequent to such notice, increase the number of directors subject to election at the meeting, such notice as to any additional nominees shall be due on the later of (i) the conclusion of the time period for Timely Notice, (ii) the date set forth in Section 2.5(b)(ii) or (iii) the tenth (10th) day following the date of public disclosure (as defined in Section 2.4) of such increase.
- (c) To be in proper form for purposes of this Section 2.5, a stockholder's notice to the Secretary shall set forth:
- (i) As to each Nominating Person (as defined below), the Stockholder Information (as defined in Section 2.4(c)(i), except that for purposes of this Section 2.5 the term "Nominating Person" shall be substituted for the term "Proposing Person" in all places it appears in Section 2.4(c)(i));
- (ii) As to each Nominating Person, any Disclosable Interests (as defined in Section 2.4(c)(ii), except that for purposes of this Section 2.5 the term "Nominating Person" shall be substituted for the term "Proposing Person" in all places it appears in Section 2.4(c) (ii) and the disclosure with respect to the business to be brought before the meeting in Section 2.4(c)(ii) shall be made with respect to the election of directors at the meeting); and
- (iii) As to each candidate whom a Nominating Person proposes to nominate for election as a director, (A) all information with respect to such candidate for nomination that would be required to be set forth in a stockholder's notice pursuant to this Section 2.5 and Section 2.6 if such candidate for nomination were a Nominating Person, (B) all information relating to such candidate for nomination that is required to be disclosed in a proxy statement or other filings required to be made in connection with solicitations of proxies for election of directors in a contested election pursuant to Section 14(a) under the Exchange Act (including such candidate's written consent to being named in the

Corporation's proxy statement as a nominee and to serving as a director if elected), (C) a description of any direct or indirect material interest in any material contract or agreement between or among any Nominating Person, on the one hand, and each candidate for nomination or his or her respective associates or any other participants in such solicitation, on the other hand, including, without limitation, all information that would be required to be disclosed pursuant to Item 404 under Regulation S-K if such Nominating Person were the "registrant" for purposes of such rule and the candidate for nomination were a director or executive officer of such registrant (the disclosures to be made pursuant to the foregoing clauses (A) through (C) are referred to as "Nominee Information"), and (D) a completed and signed questionnaire, representation and agreement as provided in Section 2.6(a).

For purposes of this Section 2.5, the term "Nominating Person" shall mean (i) the stockholder providing the notice of the nomination proposed to be made at the meeting, (ii) the beneficial owner or beneficial owners, if different, on whose behalf the notice of the nomination proposed to be made at the meeting is made, and (iii) any other participant in such solicitation.

- (d) A stockholder providing notice of any nomination proposed to be made at a meeting shall further update and supplement such notice, if necessary, so that the information provided or required to be provided in such notice pursuant to this Section 2.5 shall be true and correct as of the record date for stockholders entitled to vote at the meeting and as of the date that is ten (10) business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to, or mailed and received by, the Secretary at the principal executive offices of the Corporation not later than five (5) business days after the record date for stockholders entitled to vote at the meeting (in the case of the update and supplement required to be made as of such record date), and not later than eight (8) business days prior to the date for the meeting or, if practicable, any adjournment or postponement thereof (and, if not practicable, on the first practicable date prior to the date to which the meeting has been adjourned or postponed) (in the case of the update and supplement required to be made as of ten (10) business days prior to the meeting or any adjournment or postponement thereof). For the avoidance of doubt, the obligation to update and supplement as set forth in this paragraph or any other Section of these Bylaws shall not limit the Corporation's rights with respect to any deficiencies in any notice provided by a stockholder, extend any applicable deadlines hereunder or enable or be deemed to permit a stockholder who has previously submitted notice hereunder to amend or update any nomination or to submit any new nomination.
- (e) In addition to the requirements of this Section 2.5 with respect to any nomination proposed to be made at a meeting, each Nominating Person shall comply with all applicable requirements of the Exchange Act with respect to any such nominations.
 - 2.6 Additional Requirements for Valid Nomination of Candidates to Serve as Director and, if Elected, to be Seated as Directors
- (a) To be eligible to be a candidate for election as a director of the Corporation at an annual or special meeting, a candidate must be nominated in the manner prescribed in Section 2.5 and the candidate for nomination, whether nominated by the Board or by a stockholder of record, must have previously delivered (in accordance with the time period prescribed in Section 2.5 for delivery in a notice to such candidate given by or on behalf of the Board), to the Secretary at the principal executive offices of the Corporation, (i) a completed written questionnaire (in a form provided by the Corporation upon request) with respect to the background, qualifications, stock ownership and independence of such proposed nominee, and such additional information with respect to such proposed nominee as would be required to be provided by the Corporation pursuant to Schedule 14A if such proposed nominee were a participant in the solicitation of proxies by the Corporation in connection with such annual or

special meeting and (ii) a written representation and agreement (in form provided by the Corporation upon request) that such candidate for nomination (A) is not and, if elected as a director during his or her term of office, will not become a party to (1) any agreement, arrangement or understanding with, and has not given and will not give any commitment or assurance to, any person or entity as to how such proposed nominee, if elected as a director of the Corporation, will act or vote on any issue or question (a "Voting Commitment") or (2) any Voting Commitment that could limit or interfere with such proposed nominee's ability to comply, if elected as a director of the Corporation, with such proposed nominee's fiduciary duties under applicable law, (B) is not, and will not become a party to, any agreement, arrangement or understanding with any person or entity other than the Corporation with respect to any direct or indirect compensation or reimbursement for service as a director that has not been disclosed therein or to the Corporation, (C) if elected as a director of the Corporation, will comply with all applicable corporate governance, conflict of interest, confidentiality, stock ownership and trading and other policies and guidelines of the Corporation applicable to directors and in effect during such person's term in office as a director (and, if requested by any candidate for nomination, the Secretary of the Corporation shall provide to such candidate for nomination all such policies and guidelines then in effect), (D) if elected as director of the Corporation, intends to serve the entire term until the next meeting at which such candidate would face re-election and (E) consents to being named as a nominee in the Corporation's proxy statement pursuant to Rule 14a-4(d) under the Exchange Act and any associated proxy card of the Corporation and agrees to serve if elected as a director.

- (b) The Board may also require any proposed candidate for nomination as a Director to furnish such other information as may reasonably be requested by the Board in writing prior to the meeting of stockholders at which such candidate's nomination is to be acted upon in order for the Board to determine the eligibility of such candidate for nomination to be an independent director of the Corporation in accordance with the Corporation's Corporate Governance Guidelines.
- (c) A candidate for nomination as a director shall further update and supplement the materials delivered pursuant to this Section 2.6, if necessary, so that the information provided or required to be provided pursuant to this Section 2.6 shall be true and correct as of the record date for stockholders entitled to vote at the meeting and as of the date that is ten (10) business days prior to the meeting or any adjournment or postponement thereof, and such update and supplement shall be delivered to, or mailed and received by, the Secretary at the principal executive offices of the Corporation (or any other office specified by the Corporation in any public announcement) not later than five (5) business days after the record date for stockholders entitled to vote at the meeting (in the case of the update and supplement required to be made as of such record date), and not later than eight (8) business days prior to the date for the meeting or, if practicable, any adjournment or postponement thereof (and, if not practicable, on the first practicable date prior to the date to which the meeting has been adjourned or postponed) (in the case of the update and supplement required to be made as of ten (10) business days prior to the meeting or any adjournment or postponement thereof). For the avoidance of doubt, the obligation to update and supplement as set forth in this paragraph or any other Section of these Bylaws shall not limit the Corporation's rights with respect to any deficiencies in any notice provided by a stockholder, extend any applicable deadlines hereunder or enable or be deemed to permit a stockholder who has previously submitted notice hereunder to amend or update any proposal or to submit any new proposal, including by changing or adding nominees, matters, business or resolutions proposed to be brought before a meeting of the stockholders.
- (d) No candidate shall be eligible for nomination as a director of the Corporation unless such candidate for nomination and the Nominating Person seeking to place such candidate's name in nomination has complied with Section 2.5 and this Section 2.6, as applicable. The presiding officer at the meeting shall, if the facts warrant, determine that a

nomination was not properly made in accordance with Section 2.5 and this Section 2.6, and if he or she should so determine, he or she shall so declare such determination to the meeting, the defective nomination shall be disregarded and any ballots cast for the candidate in question (but in the case of any form of ballot listing other qualified nominees, only the ballots cast for the nominee in question) shall be void and of no force or effect.

(e) Notwithstanding anything in these Bylaws to the contrary, no candidate for nomination shall be eligible to be seated as a director of the Corporation unless nominated and elected in accordance with Section 2.5 and this Section 2.6.

2.7 <u>Notice of Stockholders' Meetings</u>.

Unless otherwise provided by law, the Certificate of Incorporation or these Bylaws, the notice of any meeting of stockholders shall be sent or otherwise given in accordance with Section 8.1 of these Bylaws not less than ten (10) nor more than sixty (60) days before the date of the meeting to each stockholder entitled to vote at such meeting. The notice shall specify the place, if any, date and time of the meeting, the means of remote communication, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such meeting, and, in the case of a special meeting, the purpose or purposes for which the meeting is called.

2.8 Quorum.

Unless otherwise provided by law, the Certificate of Incorporation or these Bylaws, the holders of a majority in voting power of the stock issued and outstanding and entitled to vote at the meeting, present in person, or by remote communication, if applicable, or represented by proxy, shall constitute a quorum for the transaction of business at all meetings of the stockholders. A quorum, once established at a meeting, shall not be broken by the withdrawal of enough votes to leave less than a quorum. If, however, a quorum is not present or represented at any meeting of the stockholders, then either (i) the person presiding over the meeting or (ii) a majority in voting power of the stockholders entitled to vote at the meeting, present in person, or by remote communication, if applicable, or represented by proxy, shall have power to recess the meeting or adjourn the meeting from time to time in the manner provided in Section 2.9 of these Bylaws until a quorum is present or represented. At any recessed or adjourned meeting at which a quorum is present or represented, any business may be transacted that might have been transacted at the meeting as originally noticed.

2.9 <u>Adjourned Meeting: Notice</u>.

Any meeting of stockholders, annual or special, may adjourn from time to time to reconvene at the same or some other place. When a meeting is adjourned to another time or place, unless these Bylaws otherwise require, notice need not be given of the adjourned meeting if the time, place, if any, thereof, and the means of remote communications, if any, by which stockholders and proxy holders may be deemed to be present in person and vote at such adjourned meeting are announced at the meeting at which the adjournment is taken. At any adjourned meeting, the Corporation may transact any business which might have been transacted at the original meeting. If the adjournment is for more than thirty (30) days, a notice of the adjourned meeting shall be given to each stockholder of record entitled to vote at the meeting. If after the adjournment a new record date for determination of stockholders entitled to vote is fixed for the adjourned meeting, the Board shall fix as the record date for determining stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for determination of stockholders entitled to vote at the adjourned meeting, and shall give notice of the adjourned meeting to each stockholder of record entitled to vote at such meeting as of the record date so fixed for notice of such adjourned meeting.

2.10 Conduct of Business.

The date and time of the opening and the closing of the polls for each matter upon which the stockholders will vote at a meeting shall be announced at the meeting by the person presiding over the meeting. The Board may adopt by resolution such rules and regulations for the conduct of the meeting of stockholders as it shall deem appropriate. Except to the extent inconsistent with such rules and regulations as adopted by the Board, the person presiding over any meeting of stockholders shall have the right and authority to convene and (for any or no reason) to recess and/or adjourn the meeting, to prescribe such rules, regulations and procedures (which need not be in writing) and to do all such acts as, in the judgment of such presiding person, are appropriate for the proper conduct of the meeting. Such rules, regulations or procedures, whether adopted by the Board or prescribed by the person presiding over the meeting, may include, without limitation, the following: (i) the establishment of an agenda or order of business for the meeting; (ii) rules and procedures for maintaining order at the meeting and the safety of those present (including, without limitation, rules and procedures for removal of disruptive persons from the meeting); (iii) limitations on attendance at or participation in the meeting to stockholders entitled to vote at the meeting, their duly authorized and constituted proxies or such other persons as the person presiding over the meeting shall determine; (iv) restrictions on entry to the meeting after the time fixed for the commencement thereof; and (v) limitations on the time allotted to questions or comments by participants. The presiding person at any meeting of stockholders, in addition to making any other determinations that may be appropriate to the conduct of the meeting (including, without limitation, determinations with respect to the administration and/or interpretation of any of the rules, regulations or procedures of the meeting, whether adopted by the Board or prescribed by the person presiding over the meeting), shall, if the facts warrant, determine and declare to the meeting that a matter of business was not properly brought before the meeting and if such presiding person should so determine, such presiding person shall so declare to the meeting and any such matter or business not properly brought before the meeting shall not be transacted or considered. Unless and to the extent determined by the Board or the person presiding over the meeting, meetings of stockholders shall not be required to be held in accordance with the rules of parliamentary procedure.

2.11 <u>Voting</u>.

Except as may be otherwise provided in the Certificate of Incorporation, these Bylaws or the DGCL, each stockholder shall be entitled to one (1) vote for each share of capital stock held by such stockholder.

Except as otherwise provided by the Certificate of Incorporation, at all duly called or convened meetings of stockholders at which a quorum is present, for the election of directors, a plurality of the votes cast shall be sufficient to elect a director. Except as otherwise provided by the Certificate of Incorporation, these Bylaws, the rules or regulations of any stock exchange applicable to the Corporation, or applicable law or pursuant to any regulation applicable to the Corporation or its securities, each other matter presented to the stockholders at a duly called or convened meeting at which a quorum is present shall be decided by the affirmative vote of the holders of a majority in voting power of the votes cast (excluding abstentions and broker non-votes) on such matter.

2.12 Record Date for Stockholder Meetings and Other Purposes.

In order that the Corporation may determine the stockholders entitled to notice of or to vote at any meeting of stockholders or any adjournment thereof, the Board may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted by the Board, and which record date shall, unless otherwise required by law, not be more than sixty (60) days nor less than ten (10) days before the date of such meeting. If the Board so

fixes a date, such date shall also be the record date for determining the stockholders entitled to vote at such meeting unless the Board determines, at the time it fixes such record date, that a later date on or before the date of the meeting shall be the date for making such determination. If no record date is fixed by the Board, the record date for determining stockholders entitled to notice of or to vote at a meeting of stockholders shall be the close of business on the next day preceding the day on which notice is first given, or, if notice is waived, at the close of business on the day next preceding the day on which the meeting is held. A determination of stockholders of record entitled to notice of or to vote at a meeting of stockholders shall apply to any adjournment of the meeting; provided, however, that the Board may fix a new record date for determination of stockholders entitled to vote at the adjourned meeting; and in such case shall also fix as the record date for stockholders entitled to notice of such adjourned meeting the same or an earlier date as that fixed for determination of stockholders entitled to vote in accordance herewith at the adjourned meeting.

In order that the Corporation may determine the stockholders entitled to receive payment of any dividend or other distribution or allotment or any rights or the stockholders entitled to exercise any rights in respect of any change, conversion or exchange of capital stock, or for the purposes of any other lawful action, the Board may fix a record date, which record date shall not precede the date upon which the resolution fixing the record date is adopted, and which record date shall be not more than sixty (60) days prior to such action. If no record date is fixed, the record date for determining stockholders for any such purpose shall be at the close of business on the day on which the Board adopts the resolution relating thereto.

2.13 Proxies.

Each stockholder entitled to vote at a meeting of stockholders may authorize another person or persons to act for such stockholder by proxy authorized by an instrument in writing or by a transmission permitted by law filed in accordance with the procedure established for the meeting, but no such proxy shall be voted or acted upon after three (3) years from its date, unless the proxy provides for a longer period. The revocability of a proxy that states on its face that it is irrevocable shall be governed by the provisions of Section 212 of the DGCL. A proxy may be in the form of an electronic transmission which sets forth or is submitted with information from which it can be determined that the transmission was authorized by the stockholder.

2.14 <u>List of Stockholders Entitled to Vote</u>.

The Corporation shall prepare, at least ten (10) days before every meeting of stockholders, a complete list of the stockholders entitled to vote at the meeting (provided, however, that if the record date for determining the stockholders entitled to vote is less than ten (10) days before the date of the meeting, the list shall reflect the stockholders entitled to vote as of the tenth day before the meeting date), arranged in alphabetical order, and showing the address of each stockholder and the number of shares registered in the name of each stockholder. The Corporation shall not be required to include electronic mail addresses or other electronic contact information on such list. Such list shall be open to the examination of any stockholder, for any purpose germane to the meeting for a period of at least ten (10) days prior to the meeting: (i) on a reasonably accessible electronic network, provided that the information required to gain access to such list is provided with the notice of the meeting, or (ii) during ordinary business hours, at the Corporation's principal executive office. In the event that the Corporation determines to make the list available on an electronic network, the Corporation may take reasonable steps to ensure that such information is available only to stockholders of the Corporation. If the meeting is to be held at a place, then the list shall be produced and kept at the time and place of the meeting during the whole time thereof, and may be inspected by any stockholder who is present. If the meeting is to be held solely by means of remote communication, then the list shall also be open to the examination of any stockholder during the whole time of the meeting on a reasonably

accessible electronic network, and the information required to access such list shall be provided with the notice of the meeting. Such list shall presumptively determine the identity of the stockholders entitled to vote at the meeting and the number of shares held by each of them. Except as otherwise provided by law, the stock ledger shall be the only evidence as to who are the stockholders entitled to examine the list of stockholders required by this Section 2.14 or to vote in person or by proxy at any meeting of stockholders.

2.15 <u>Inspectors of Election</u>.

Before any meeting of stockholders, the Corporation shall appoint an inspector or inspectors of election to act at the meeting or its adjournment and make a written report thereof. The Corporation may designate one or more persons as alternate inspectors to replace any inspector who fails to act. If any person appointed as inspector or any alternate fails to appear or fails or refuses to act, then the person presiding over the meeting shall appoint a person to fill that vacancy.

Such inspectors shall:

- (i) determine the number of shares outstanding and the voting power of each, the number of shares represented at the meeting and the validity of any proxies and ballots;
 - (ii) count all votes or ballots;
 - (iii) count and tabulate all votes;
- (iv) determine and retain for a reasonable period a record of the disposition of any challenges made to any determination by the inspector(s); and
- (v) certify its or their determination of the number of shares represented at the meeting and its or their count of all votes and ballots.

Each inspector, before entering upon the discharge of the duties of inspector, shall take and sign an oath faithfully to execute the duties of inspection with strict impartiality and according to the best of such inspector's ability. Any report or certificate made by the inspectors of election is prima facie evidence of the facts stated therein. The inspectors of election may appoint such persons to assist them in performing their duties as they determine.

2.16 <u>Delivery to the Corporation</u>.

Whenever this Article II requires one or more persons (including a record or beneficial owner of stock) to deliver a document or information to the Corporation or any officer, employee or agent thereof (including any notice, request, questionnaire, revocation, representation or other document or agreement), such document or information shall be in writing exclusively (and not in an electronic transmission) and shall be delivered exclusively by hand (including, without limitation, overnight courier service) or by certified or registered mail, return receipt requested, and the Corporation shall not be required to accept delivery of any document not in such written form or so delivered. For the avoidance of doubt, the Corporation expressly opts out of Section 116 of the DGCL with respect to the delivery of information and documents to the Corporation required by this Article II.

Article III - Directors

3.1 Powers.

Except as otherwise provided by the Certificate of Incorporation or the DGCL, the business and affairs of the Corporation shall be managed by or under the direction of the Board.

3.2 <u>Number of Directors</u>.

Subject to the Certificate of Incorporation, the total number of directors constituting the Board shall be determined from time to time by resolution of the Board. No reduction of the authorized number of directors shall have the effect of removing any director before that director's term of office expires.

3.3 Election, Qualification and Term of Office of Directors.

Except as provided in Section 3.4 of these Bylaws, and subject to the Certificate of Incorporation, each director, including a director elected to fill a vacancy or newly created directorship, shall hold office until the expiration of the term of the class, if any, for which elected and until such director's successor is elected and qualified or until such director's earlier death, resignation, disqualification or removal. Directors need not be stockholders to be qualified for election or service as a director of the Corporation. The Certificate of Incorporation or these Bylaws may prescribe qualifications for directors.

3.4 Resignation; Vacancies; Removal.

Any director may resign at any time upon notice given in writing or by electronic transmission to the Corporation. The resignation shall take effect at the time specified therein or upon the happening of an event specified therein, and if no time or event is specified, at the time of its receipt. When one or more directors so resigns and the resignation is effective at a future date or upon the happening of an event to occur on a future date, a majority of the directors then in office, including those who have so resigned, shall have power to fill such vacancy or vacancies, the vote thereon to take effect when such resignation or resignations shall become effective, and each director so chosen shall hold office as provided in Section 3.3.

Unless otherwise provided in the Certificate of Incorporation or these Bylaws, vacancies resulting from the death, resignation, disqualification or removal of any director, and newly created directorships resulting from any increase in the authorized number of directors shall be filled only by a majority of the directors then in office, although less than a quorum, or by a sole remaining director.

Directors of the Corporation may be removed only as expressly provided in the Certificate of Incorporation.

3.5 Place of Meetings; Meetings by Telephone.

The Board may hold meetings, both regular and special, either within or outside the State of Delaware.

Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, members of the Board, or any committee designated by the Board, may participate in a meeting of the Board, or any committee, by means of conference telephone or other communications equipment by means of which all persons participating in the meeting can hear each other, and such

participation in a meeting pursuant to this bylaw shall constitute presence in person at the meeting.

3.6 <u>Regular Meetings</u>.

Regular meetings of the Board may be held within or outside the State of Delaware and at such time and at such place as which has been designated by the Board and publicized among all directors, either orally or in writing, by telephone, including a voice-messaging system or other system designed to record and communicate messages, facsimile, telegraph or telex, or by electronic mail or other means of electronic transmission. No further notice shall be required for regular meetings of the Board.

3.7 Special Meetings; Notice.

Special meetings of the Board for any purpose or purposes may be called at any time by the Chairperson of the Board, the Chief Executive Officer, the President, the Secretary or a majority of the total number of directors constituting the Board.

Notice of the time and place of special meetings shall be:

- (i) delivered personally by hand, by courier or by telephone;
- (ii) sent by United States first-class mail, postage prepaid;
- (iii) sent by facsimile or electronic mail; or
- (iv) sent by other means of electronic transmission,

directed to each director at that director's address, telephone number, facsimile number or electronic mail address, or other address for electronic transmission, as the case may be, as shown on the Corporation's records.

If the notice is (i) delivered personally by hand, by courier or by telephone, (ii) sent by facsimile or electronic mail, or (iii) sent by other means of electronic transmission, it shall be delivered or sent at least twenty-four (24) hours before the time of the holding of the meeting. If the notice is sent by U.S. mail, it shall be deposited in the U.S. mail at least four (4) days before the time of the holding of the meeting. The notice need not specify the place of the meeting (if the meeting is to be held at the Corporation's principal executive office) nor the purpose of the meeting.

3.8 Quorum.

At all meetings of the Board, unless otherwise provided by the Certificate of Incorporation, a majority of the total number of directors shall constitute a quorum for the transaction of business; *provided* that, solely for the purposes of filling vacancies pursuant to Section 3.4 of these Bylaws, a meeting of the Board may be held if a majority of the directors then in office participate in such meeting. The vote of a majority of the directors present at any meeting at which a quorum is present shall be the act of the Board, except as may be otherwise specifically provided by statute, the Certificate of Incorporation or these Bylaws. If a quorum is not present at any meeting of the Board, then the directors present thereat may adjourn the meeting from time to time, without notice other than announcement at the meeting, until a quorum is present.

3.9 Board Action Without a Meeting.

Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, any action required or permitted to be taken at any meeting of the Board, or of any committee thereof, may be taken without a meeting if all members of the Board or committee, as the case may be, consent thereto in writing or by electronic transmission. After an action is taken, the consent or consents relating thereto shall be filed with the minutes of the proceedings of the Board, or the committee thereof, in the same paper or electronic form as the minutes are maintained. Such action by written consent or consent by electronic transmission shall have the same force and effect as a unanimous vote of the Board.

3.10 Fees and Compensation of Directors.

Unless otherwise restricted by the Certificate of Incorporation or these Bylaws, the Board shall have the authority to fix the compensation, including fees and reimbursement of expenses, of directors for services to the Corporation in any capacity.

Article IV - Committees

4.1 Committees of Directors.

The Board may designate one (1) or more committees, each committee to consist, of one (1) or more of the directors of the Corporation. The Board may designate one (1) or more directors as alternate members of any committee, who may replace any absent or disqualified member at any meeting of the committee. In the absence or disqualification of a member of a committee, the member or members thereof present at any meeting and not disqualified from voting, whether or not such member or members constitute a quorum, may unanimously appoint another member of the Board to act at the meeting in the place of any such absent or disqualified member. Any such committee, to the extent provided in the resolution of the Board or in these Bylaws, shall have and may exercise all the powers and authority of the Board in the management of the business and affairs of the Corporation, and may authorize the seal of the Corporation to be affixed to all papers that may require it; but no such committee shall have the power or authority to (i) approve or adopt, or recommend to the stockholders, any action or matter expressly required by the DGCL to be submitted to stockholders for approval, or (ii) adopt, amend or repeal any bylaw of the Corporation.

4.2 Committee Minutes.

Each committee shall keep regular minutes of its meetings and report the same to the Board when required.

4.3 <u>Meetings and Actions of Committees</u>.

Meetings and actions of committees shall be governed by, and held and taken in accordance with, the provisions of:

- (i) Section 3.5 (place of meetings; meetings by telephone);
- (ii) Section 3.6 (regular meetings);
- (iii) Section 3.7 (special meetings; notice);
- (iv) Section 3.9 (board action without a meeting); and

(v) Section 7.13 (waiver of notice),

with such changes in the context of those Bylaws as are necessary to substitute the committee and its members for the Board and its members. *However*:

- (i) the time of regular meetings of committees may be determined either by resolution of the Board or by resolution of the committee;
- (ii) special meetings of committees may also be called by resolution of the Board or the chairperson of the applicable committee; and
- (iii) the Board may adopt rules for the governance of any committee to override the provisions that would otherwise apply to the committee pursuant to this Section 4.3, provided that such rules do not violate the provisions of the Certificate of Incorporation or applicable law.

4.4 Subcommittees.

Unless otherwise provided in the Certificate of Incorporation, these Bylaws or the resolutions of the Board designating the committee, a committee may create one (1) or more subcommittees, each subcommittee to consist of one (1) or more members of the committee, and delegate to a subcommittee any or all of the powers and authority of the committee.

Article V - Officers

5.1 Officers.

The officers of the Corporation shall include a Chief Executive Officer, Chief Operating Officer, Chief Financial Officer, a President and a Secretary. The Corporation may also have, at the discretion of the Board, a Chairperson of the Board, a Vice Chairperson of the Board, a Treasurer, one (1) or more Vice Presidents, one (1) or more Executive Vice Presidents, one (1) or more Assistant Treasurers, one (1) or more Assistant Secretaries, and any such other officers as may be appointed in accordance with the provisions of these Bylaws. Any number of offices may be held by the same person. No officer need be a stockholder or director of the Corporation.

5.2 Appointment of Officers.

The Board shall appoint the officers of the Corporation, except such officers as may be appointed in accordance with the provisions of Section 5.3 of these Bylaws.

5.3 Subordinate Officers.

The Board may appoint, or empower the Chief Executive Officer or, in the absence of a Chief Executive Officer, the Chief Financial Officer, to appoint, such other officers and agents as the business of the Corporation may require. Each of such officers and agents shall hold office for such period, have such authority, and perform such duties as are provided in these Bylaws or as the Board may from time to time determine.

5.4 Removal and Resignation of Officers.

Subject to the rights, if any, of an officer under any contract of employment, any officer may be removed, either with or without cause, by the Board or, except in the case of an officer chosen by the Board, by any officer upon whom such power of removal may be conferred by the Board.

Any officer may resign at any time by giving notice in writing or by electronic transmission to the Corporation. Any resignation shall take effect at the date of the receipt of that notice or at any later time specified in that notice. Unless otherwise specified in the notice of resignation, the acceptance of the resignation shall not be necessary to make it effective. If a resignation is made effective at a later date and the Corporation accepts the future effective date, the Board may fill the pending vacancy before the effective date if the Board provides that the successor shall not take office until the effective date. Any resignation is without prejudice to the rights, if any, of the Corporation under any contract to which the officer is a party.

5.5 <u>Vacancies in Offices</u>.

Any vacancy occurring in any office of the Corporation shall be filled by the Board or as provided in Section 5.2 or Section 5.3, as applicable.

5.6 Representation of Shares of Other Corporations.

The Chairperson of the Board, the Chief Executive Officer, or the President of this Corporation, or any other person authorized by the Board, the Chief Executive Officer, the President or the Chief Financial Officer, is authorized to vote, represent and exercise on behalf of this Corporation all rights incident to any and all shares or voting securities of any other corporation or other person standing in the name of this Corporation. The authority granted herein may be exercised either by such person directly or by any other person authorized to do so by proxy or power of attorney duly executed by such person having the authority.

5.7 <u>Authority and Duties of Officers</u>.

All officers of the Corporation shall respectively have such authority and perform such duties in the management of the business of the Corporation as may be provided herein or designated from time to time by the Board and, to the extent not so provided, as generally pertain to their respective offices, subject to the oversight of the Board.

5.8 <u>Compensation</u>.

The compensation of the officers of the Corporation for their services as such shall be fixed from time to time by or at the direction of the Board. An officer of the Corporation shall not be prevented from receiving compensation by reason of the fact that he or she is also a director of the Corporation.

Article VI - Records

A stock ledger consisting of one or more records in which the names of all of the Corporation's stockholders of record, the address and number of shares registered in the name of each such stockholder, and all issuances and transfers of stock of the Corporation are recorded in accordance with Section 224 of the DGCL shall be administered by or on behalf of the Corporation. Any records administered by or on behalf of the Corporation in the regular course of its business, including its stock ledger, books of account, and minute books, may be kept on, or by means of, or be in the form of, any information storage device, or method, or one or more electronic networks or databases (including one or more distributed electronic networks or databases), provided that the records so kept can be converted into clearly legible paper form within a reasonable time and, with respect to the stock ledger, that the records so kept (i) can be used to prepare the list of stockholders specified in Sections 219 and 220 of the DGCL, (ii) record the information specified in Sections 156, 159, 217(a) and 218 of the DGCL, and (iii) record transfers of stock as governed by Article 8 of the Uniform Commercial Code as adopted in the State of Delaware.

Article VII - General Matters

7.1 <u>Execution of Corporate Contracts and Instruments.</u>

The Board, except as otherwise provided in these Bylaws, may authorize any officer or officers, or agent or agents, to enter into any contract or execute any instrument in the name of and on behalf of the Corporation; such authority may be general or confined to specific instances.

7.2 Stock Certificates.

The shares of the Corporation may be represented by certificates, provided that the Board by resolution may provide that some or all of the shares of any class or series of stock of the Corporation shall be uncertificated. Certificates for the shares of stock, if any, shall be in such form as is consistent with the Certificate of Incorporation and applicable law. Every holder of stock represented by a certificate shall be entitled to have a certificate signed by, or in the name of the Corporation by, any two officers authorized to sign stock certificates representing the number of shares registered in certificate form. The Chairperson or Vice Chairperson of the Board, Chief Executive Officer, the President, Chief Financial Officer, any Executive Vice President, the Treasurer, any Assistant Treasurer, the Secretary or any Assistant Secretary of the Corporation shall be specifically authorized to sign stock certificates. Any or all of the signatures on the certificate may be a facsimile. In case any officer, transfer agent or registrar who has signed or whose facsimile signature has been placed upon a certificate has ceased to be such officer, transfer agent or registrar before such certificate is issued, it may be issued by the Corporation with the same effect as if he or she were such officer, transfer agent or registrar at the date of issue.

The Corporation may issue the whole or any part of its shares as partly paid and subject to call for the remainder of the consideration to be paid therefor. Upon the face or back of each stock certificate issued to represent any such partly paid shares, or upon the books and records of the Corporation in the case of uncertificated partly paid shares, the total amount of the consideration to be paid therefor and the amount paid thereon shall be stated. Upon the declaration of any dividend on fully paid shares, the Corporation shall declare a dividend upon partly paid shares of the same class, but only upon the basis of the percentage of the consideration actually paid thereon.

7.3 Special Designation of Certificates.

If the Corporation is authorized to issue more than one class of stock or more than one series of any class, then the powers, the designations, the preferences and the relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights shall be set forth in full or summarized on the face or on the back of the certificate that the Corporation shall issue to represent such class or series of stock (or, in the case of uncertificated shares, set forth in a notice provided pursuant to Section 151 of the DGCL); provided, however, that except as otherwise provided in Section 202 of the DGCL, in lieu of the foregoing requirements, there may be set forth on the face of back of the certificate that the Corporation shall issue to represent such class or series of stock (or, in the case of any uncertificated shares, included in the aforementioned notice) a statement that the Corporation will furnish without charge to each stockholder who so requests the powers, the designations, the preferences and the relative, participating, optional or other special rights of each class of stock or series thereof and the qualifications, limitations or restrictions of such preferences and/or rights.

7.4 Lost Certificates.

Except as provided in this Section 7.4, no new certificates for shares shall be issued to replace a previously issued certificate unless the latter is surrendered to the Corporation and cancelled at the same time. The Corporation may issue a new certificate of stock or uncertificated shares in the place of any certificate theretofore issued by it, alleged to have been lost, stolen or destroyed, and the Corporation may require the owner of the lost, stolen or destroyed certificate, or such owner's legal representative, to give the Corporation a bond sufficient to indemnify it against any claim that may be made against it on account of the alleged loss, theft or destruction of any such certificate or the issuance of such new certificate or uncertificated shares.

7.5 Shares Without Certificates.

The Corporation may adopt a system of issuance, recordation and transfer of its shares of stock by electronic or other means not involving the issuance of certificates, provided the use of such system by the Corporation is permitted in accordance with applicable law.

7.6 Construction; Definitions.

Unless the context requires otherwise, the general provisions, rules of construction and definitions in the DGCL shall govern the construction of these Bylaws. Without limiting the generality of this provision, the singular number includes the plural and the plural number includes the singular.

7.7 Dividends.

The Board, subject to any restrictions contained in either (i) the DGCL or (ii) the Certificate of Incorporation, may declare and pay dividends upon the shares of its capital stock. Dividends may be paid in cash, in property or in shares of the Corporation's capital stock.

The Board may set apart out of any of the funds of the Corporation available for dividends a reserve or reserves for any proper purpose and may abolish any such reserve. Such purposes shall include but not be limited to equalizing dividends, repairing or maintaining any property of the Corporation, and meeting contingencies.

7.8 Fiscal Year.

The fiscal year of the Corporation shall be fixed by resolution of the Board and may be changed by the Board.

7.9 <u>Seal</u>.

The Corporation may adopt a corporate seal, which shall be adopted and which may be altered by the Board. The Corporation may use the corporate seal by causing it or a facsimile thereof to be impressed or affixed or in any other manner reproduced.

7.10 Transfer of Stock.

Shares of the Corporation shall be transferable in the manner prescribed by law, the Certificate of Incorporation and in these Bylaws. Shares of stock of the Corporation shall be transferred on the books of the Corporation only by the holder of record thereof or by such holder's attorney duly authorized in writing, upon surrender to the Corporation of the certificate or certificates representing such shares endorsed by the appropriate person or persons (or by

delivery of duly executed instructions with respect to uncertificated shares), with such evidence of the authenticity of such endorsement or execution, transfer, authorization and other matters as the Corporation may reasonably require, and accompanied by all necessary stock transfer stamps. No transfer of stock shall be valid as against the Corporation for any purpose until it shall have been entered in the stock records of the Corporation by an entry showing the names of the persons from and to whom it was transferred.

7.11 Stock Transfer Agreements.

The Corporation shall have power to enter into and perform any agreement with any number of stockholders of any one or more classes or series of stock of the Corporation to restrict the transfer of shares of stock of the Corporation of any one or more classes owned by such stockholders in any manner not prohibited by the DGCL.

7.12 <u>Registered Stockholders</u>.

The Corporation:

- (i) shall be entitled to recognize the exclusive right of a person registered on its books as the owner of shares to receive dividends and to vote as such owner; and
- (ii) shall not be bound to recognize any equitable or other claim to or interest in such share or shares on the part of another person, whether or not it shall have express or other notice thereof, except as otherwise provided by the laws of the State of Delaware.

7.13 Waiver of Notice.

Whenever notice is required to be given under any provision of the DGCL, the Certificate of Incorporation or these Bylaws, a written waiver, signed by the person entitled to notice, or a waiver by electronic transmission by the person entitled to notice, whether before or after the time of the event for which notice is to be given, shall be deemed equivalent to notice. Attendance of a person at a meeting shall constitute a waiver of notice of such meeting, except when the person attends a meeting for the express purpose of objecting at the beginning of the meeting, to the transaction of any business because the meeting is not lawfully called or convened. Neither the business to be transacted at, nor the purpose of, any regular or special meeting of the stockholders need be specified in any written waiver of notice or any waiver by electronic transmission unless so required by the Certificate of Incorporation or these Bylaws.

Article VIII - Notice

8.1 <u>Delivery of Notice; Notice by Electronic Transmission.</u>

Without limiting the manner by which notice otherwise may be given effectively to stockholders, any notice to stockholders given by the Corporation under any provisions of the DGCL, the Certificate of Incorporation, or these Bylaws may be given in writing directed to the stockholder's mailing address (or by electronic transmission directed to the stockholder's electronic mail address, as applicable) as it appears on the records of the Corporation and shall be given (1) if mailed, when the notice is deposited in the U.S. mail, postage prepaid, (2) if delivered by courier service, the earlier of when the notice is received or left at such stockholder's address or (3) if given by electronic mail, when directed to such stockholder's electronic mail address unless the stockholder has notified the Corporation in writing or by electronic transmission of an objection to receiving notice by electronic mail. A notice by

electronic mail must include a prominent legend that the communication is an important notice regarding the Corporation.

Without limiting the manner by which notice otherwise may be given effectively to stockholders, any notice to stockholders given by the Corporation under any provision of the DGCL, the Certificate of Incorporation or these Bylaws shall be effective if given by a form of electronic transmission consented to by the stockholder to whom the notice is given. Any such consent shall be revocable by the stockholder by written notice or electronic transmission to the Corporation. Notwithstanding the provisions of this paragraph, the Corporation may give a notice by electronic mail in accordance with the first paragraph of this section without obtaining the consent required by this paragraph.

Any notice given pursuant to the preceding paragraph shall be deemed given:

- (i) if by facsimile telecommunication, when directed to a number at which the stockholder has consented to receive notice;
- (ii) if by a posting on an electronic network together with separate notice to the stockholder of such specific posting, upon the later of (A) such posting and (B) the giving of such separate notice; and
 - (iii) if by any other form of electronic transmission, when directed to the stockholder.

Notwithstanding the foregoing, a notice may not be given by an electronic transmission from and after the time that (1) the Corporation is unable to deliver by such electronic transmission two (2) consecutive notices given by the Corporation and (2) such inability becomes known to the Secretary or an Assistant Secretary of the Corporation or to the transfer agent, or other person responsible for the giving of notice, provided, however, the inadvertent failure to discover such inability shall not invalidate any meeting or other action.

An affidavit of the Secretary or an Assistant Secretary or of the transfer agent or other agent of the Corporation that the notice has been given shall, in the absence of fraud, be prima facie evidence of the facts stated therein.

Article IX - Indemnification

9.1 <u>Indemnification of Directors and Officers.</u>

The Corporation shall indemnify and hold harmless, to the fullest extent permitted by the DGCL as it presently exists or may hereafter be amended, any current or former director or officer of the Corporation who was or is made or is threatened to be made a party or is otherwise involved in any action, suit or proceeding, whether civil, criminal, administrative or investigative (a "Proceeding") by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was a director or officer of the Corporation or, while serving as a director or officer of the Corporation, is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, enterprise or non-profit entity (a "covered person"), including service with respect to employee benefit plans, out of the assets of the Corporation, against all liability, claims, damages, costs, and loss suffered and expenses (including attorneys' fees, judgments, fines ERISA excise taxes or penalties and amounts paid in settlement) reasonably incurred by such person in connection with any such Proceeding other than such liability (if any) that they may incur by reason of their own actual fraud, willful neglect, or willful default. Notwithstanding the preceding sentence, except as otherwise provided in Section 9.4, the Corporation shall be required to indemnify a

person in connection with a Proceeding initiated by such person only if the Proceeding was authorized in the specific case by the Board. No covered person shall be found to have committed actual fraud, willful neglect or willful default under this Article unless or until a court of competent jurisdiction shall have made a finding to that effect. The directors, on behalf of the Corporation, may purchase and maintain insurance for the benefit of any director or officer against any liability which, by virtue of any rule of law, would otherwise attach to such person in respect of any negligence, default, breach of duty or breach of trust of which such person may be guilty in relation to the Corporation.

9.2 Indemnification of Others.

The Corporation shall have the power to indemnify and hold harmless, to the fullest extent permitted by applicable law as it presently exists or may hereafter be amended, any employee or agent of the Corporation who was or is made or is threatened to be made a party or is otherwise involved in any Proceeding by reason of the fact that he or she, or a person for whom he or she is the legal representative, is or was an employee or agent of the Corporation or is or was serving at the request of the Corporation as a director, officer, employee or agent of another corporation or of a partnership, joint venture, trust, enterprise or non-profit entity, including service with respect to employee benefit plans, against all liability and loss suffered and expenses reasonably incurred by such person in connection with any such Proceeding.

9.3 <u>Prepayment of Expenses.</u>

The Corporation shall to the fullest extent not prohibited by applicable law pay or advance the expenses (including attorneys' fees) incurred by any covered person, and may pay or advance the expenses incurred by any employee or agent of the Corporation, in defending any Proceeding in advance of its final disposition; *provided, however*, that such payment or advance of expenses in advance of the final disposition of the Proceeding shall be made only upon receipt of an undertaking by the person to repay all amounts paid or advanced (without interest) if it should be ultimately determined by final judgment or other final adjudication that the person is not entitled to be indemnified under this Article IX or otherwise.

9.4 <u>Determination; Claim.</u>

If a claim for indemnification (following the final judgment or other final adjudication of such Proceeding) under this Article IX is not paid in full within sixty (60) days, or a claim for advancement of expenses under this Article IX is not paid in full within thirty (30) days, after a written claim therefor has been received by the Corporation the claimant may thereafter (but not before) file suit to recover the unpaid amount of such claim and, if successful in whole or in part, shall be entitled to be paid the expense of prosecuting such claim to the fullest extent permitted by law. In any such action the Corporation shall have the burden of proving that the claimant was not entitled to the requested indemnification or payment of expenses under applicable law.

9.5 <u>Non-Exclusivity of Rights</u>.

The rights conferred on any person by this Article IX shall not be exclusive of any other rights which such person may have or hereafter acquire under any statute, provision of the Certificate of Incorporation, these Bylaws, agreement, vote of stockholders or disinterested directors or otherwise.

9.6 <u>Insurance</u>.

The Corporation may purchase and maintain insurance on behalf of any person who is or was a director, officer, employee or agent of the Corporation, or is or was serving at the request

of the Corporation as a director, officer, employee or agent of another corporation, partnership, joint venture, trust enterprise or non-profit entity against any liability asserted against him or her and incurred by him or her in any such capacity, or arising out of his or her status as such, whether or not the Corporation would have the power to indemnify him or her against such liability under the DGCL.

9.7 Other Indemnification.

The Corporation's obligation, if any, to indemnify or advance expenses to any person who was or is serving at its request as a director, officer, employee or agent of another corporation, partnership, joint venture, trust, enterprise or non-profit entity shall be reduced by any amount such person may collect as indemnification or advancement of expenses from such other corporation, partnership, joint venture, trust, enterprise or non-profit enterprise.

9.8 <u>Continuation of Indemnification</u>.

The rights to indemnification and to prepayment of expenses provided by, or granted pursuant to, this Article IX shall continue notwithstanding that the person has ceased to be a director or officer of the Corporation and shall inure to the benefit of the estate, heirs, executors, administrators, legatees and distributees of such person.

9.9 Amendment or Repeal; Interpretation.

The provisions of this Article IX shall constitute a contract between the Corporation, on the one hand, and, on the other hand, each individual who serves or has served as a director or officer of the Corporation (whether before or after the adoption of these Bylaws), in consideration of such person's performance of such services, and pursuant to this Article IX the Corporation intends to be legally bound to each such current or former director or officer of the Corporation. With respect to current and former directors and officers of the Corporation, the rights conferred under this Article IX are present contractual rights and such rights are fully vested, and shall be deemed to have vested fully, immediately upon adoption of these Bylaws. With respect to any directors or officers of the Corporation who commence service following adoption of these Bylaws, the rights conferred under this provision shall be present contractual rights and such rights shall fully vest, and be deemed to have vested fully, immediately upon such director or officer commencing service as a director or officer of the Corporation. Any repeal or modification of the foregoing provisions of this Article IX shall not adversely affect any right or protection (i) hereunder of any person in respect of any act or omission occurring prior to the time of such repeal or modification or (ii) under any agreement providing for indemnification or advancement of expenses to an officer or director of the Corporation in effect prior to the time of such repeal or modification.

Any reference to an officer of the Corporation in this Article IX shall be deemed to refer exclusively to the Chief Executive Officer, Chief Financial Officer, and Secretary, or other officer of the Corporation appointed by (x) the Board pursuant to Article V of these Bylaws or (y) an officer to whom the Board has delegated the power to appoint officers pursuant to Article V of these Bylaws, and any reference to an officer of any other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise shall be deemed to refer exclusively to an officer appointed by the board of directors (or equivalent governing body) of such other entity pursuant to the certificate of incorporation and Bylaws (or equivalent organizational documents) of such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise. The fact that any person who is or was an employee of the Corporation or an employee of any other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise has been given or has used the title of "Vice President" or any other title that could be construed to suggest or imply that such person is or may be an officer of the

Corporation or of such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise shall not result in such person being constituted as, or being deemed to be, an officer of the Corporation or of such other corporation, partnership, joint venture, trust, employee benefit plan or other enterprise for purposes of this Article IX.

Article X - Amendments

The Board is expressly empowered to adopt, amend or repeal the Bylaws of the Corporation. The stockholders also shall have power to adopt, amend or repeal the Bylaws of the Corporation; *provided, however*, that such action by stockholders shall require, in addition to any other vote required by the Certificate of Incorporation or applicable law, the affirmative vote of the holders of at least two-thirds of the voting power of all the then-outstanding shares of voting stock of the Corporation entitled to vote generally in an election of directors, voting together as a single class.

Article XI - Definitions

As used in these Bylaws, unless the context otherwise requires, the following terms shall have the following meanings:

An "<u>electronic transmission</u>" means any form of communication, not directly involving the physical transmission of paper, including the use of, or participation in, one or more electronic networks or databases (including one or more distributed electronic networks or databases), that creates a record that may be retained, retrieved and reviewed by a recipient thereof, and that may be directly reproduced in paper form by such a recipient through an automated process.

An "<u>electronic mail</u>" means an electronic transmission directed to a unique electronic mail address (which electronic mail shall be deemed to include any files attached thereto and any information hyperlinked to a website if such electronic mail includes the contact information of an officer or agent of the Corporation who is available to assist with accessing such files and information).

An "<u>electronic mail address</u>" means a destination, commonly expressed as a string of characters, consisting of a unique user name or mailbox (commonly referred to as the "local part" of the address) and a reference to an internet domain (commonly referred to as the "domain part" of the address), whether or not displayed, to which electronic mail can be sent or delivered.

The term "person" means any individual, general partnership, limited partnership, limited liability company, corporation, trust, business trust, joint stock company, joint venture, unincorporated association, cooperative or association or any other legal entity or organization of whatever nature, and shall include any successor (by merger or otherwise) of such entity.

Innovid Corp.

Certificate of Amendment and Restatement of Bylaws

The undersigned hereby certifies that he is the duly elected, qualified, and acting Secretary of Innovid Corp., a Delaware corporation (the "<u>Corporation</u>"), and that the foregoing Bylaws were approved on November 30, 2021, effective as of November 30, 2021 by the Corporation's board of directors.

IN WITNESS WHEREOF, the undersigned has hereunto set his hand this 30th day of November, 2021.

Anthony Reich Chief Financial Officer

[Signature Page to Amended & Restated Bylaws]

NUMBER C-	SHARES
SEE REVERSE FOR CERTAIN DEFINITIONS	
	CUSIP
	INNOVID CORP.
	COMMON STOCK
	par value 0.0001 per share (each, a " $Common\ Stock$ "), of Innovid Corp., a Delaware corporation (the by duly authorized attorney upon surrender of this certificate properly endorsed.
This certificate is not valid unless countersigned by the Transfe	er Agent and registered by the Registrar of the Company.
Witness the facsimile signature of a duly authorized signatory of	of the Company.
Authorized Signatory	Transfer Agent
	·

INNOVID CORP.

The Company will furnish without charge to each shareholder who so requests, a statement of the powers, designations, preferences and relative, participating, optional or other special rights of each class of equity or series thereof of the Company and the qualifications, limitations, or restrictions of such preferences and/or rights. This certificate and the shares represented thereby are issued and shall be held subject to all the provisions of the Amended and Restated Certificate of Incorporation of the Company and all amendments thereto and resolutions of the Board of Directors providing for the issue of securities (copies of which may be obtained from the secretary of the Company), to all of which the holder of this certificate by acceptance hereof assents.

The following abbreviations, when used in the inscription on the face of this certificate, shall be construed as though they were written out in full according to applicable laws or regulations:

TEN COM	— as tenants in commo		INIF GIFT MIN ACT		Custodian
				(Cust)	(Minor)
TEN ENT	 as tenants by the ent 	ireties			
				Under Uniform	m Gifts to Minors Act
JT TEN	 as joint tenants with survivorship and not common 				(State)
		Additional ab	obreviations may also be used	d though not in the	e above list.
For va	ılue received,	hereby sells, assigns and trai	nsfers unto		
		(PLEASE INSERT SOCIAL S	SECURITY OR OTHER IDE	NTIFYING NUM	IBER(S) OF ASSIGNEE(S))
(PLEASE PRINT OR TYPEWRITE NAME(S) AND ADDRESS(ES), INCLUDING ZIP CODE, OF ASSIGNEE(S))					
Common Stock 1		certificate, and do hereby irrevo se said Common Stock on the bo			wer of substitution in the premises.
Dated					

Notice:

The signature to this assignment must correspond with the name as written upon the face of the certificate in every particular, without alteration or enlargement or any change whatever.

Signature(s) Guaranteed:

THE SIGNATURE(S) MUST BE GUARANTEED BY AN ELIGIBLE GUARANTOR INSTITUTION (BANKS, STOCKBROKERS, SAVINGS AND LOAN ASSOCIATIONS AND CREDIT UNIONS WITH MEMBERSHIP IN AN APPROVED SIGNATURE GUARANTEE MEDALLION PROGRAM, PURSUANT TO S.E.C. RULE 17Ad-15 UNDER THE SECURITIES EXCHANGE ACT OF 1934, AS AMENDED (OR ANY SUCCESSOR RULE)).

Form of War	rant Certificate
-------------	------------------

[FACE]

Number	1 1	

Warrants

THIS WARRANT SHALL BE VOID IF NOT EXERCISED PRIOR TO THE EXPIRATION OF THE EXERCISE PERIOD PROVIDED FOR IN THE WARRANT AGREEMENT DESCRIBED BELOW

INNOVID CORP.

Incorporated Under the Laws of the State of Delaware

CUSIP [

1

Warrant Certificate

] warrants evidenced hereby (the"Warrants" and each, a "Warrant") to], or registered assigns, is the registered holder of [purchase Common Stock, \$0.0001 par value per share (the "Common Stock"), of Innovid Corp., a Delaware corporation (the "Company"). Each Warrant entitles the holder, upon exercise during the Exercise Period set forth in the Warrant Agreement referred to below, to receive from the Company that number of fully paid and non-assessable Common Stock as set forth below, at the exercise price (the "Warrant Price") as determined pursuant to the Warrant Agreement, payable in lawful money (or through "cashless exercise" as provided for in the Warrant Agreement) of the United States of America upon surrender of this Warrant Certificate and payment of the Warrant Price at the office or agency of the Warrant Agent referred to below, subject to the conditions set forth herein and in the Warrant Agreement. Defined terms used in this Warrant Certificate but not defined herein shall have the meanings given to them in the Warrant Agreement.

Each whole Warrant is initially exercisable for one fully paid and non-assessable Common Stock. No fractional shares will be issued upon exercise of any Warrant. If, upon the exercise of Warrants, a holder would be entitled to receive a fractional interest in a Common Stock, the Company will, upon exercise, round down to the nearest whole number the number of Common Stock to be issued to the Warrant holder. The number of Common Stock issuable upon exercise of the Warrants is subject to adjustment upon the occurrence of certain events set forth in the Warrant Agreement.

The initial Warrant Price per Common Stock for any Warrant is equal to \$11.50 per share. The Warrant Price is subject to adjustment upon the occurrence of certain events set forth in the Warrant Agreement.

Subject to the conditions set forth in the Warrant Agreement, the Warrants may be exercised only during the Exercise Period and to the extent not exercised by the end of such Exercise Period, such Warrants shall become void.

Reference is hereby made to the further provisions of this Warrant Certificate set forth on the reverse hereof and such further provisions shall for all purposes have the same effect as though fully set forth at this place.

This Warrant Certificate shall not be valid unless countersigned by the Warrant Agent, as such term is used in the Warrant Agreement.

This Warrant Certificate shall be governed by and construed in accordance with the internal laws of the State of New York.

[Signature Page Follows]

INNOVID CORP.
By:
Name: Tanya Andreev-Kaspin
Title: Chief Financial Officer
CONTINENTAL STOCK TRANSFER & TRUST
COMPANY as Warrant Agent
Ву:
Name:
Title:

[Form of Warrant Certificate]

[Reverse]

The Warrants evidenced by this Warrant Certificate are part of a duly authorized issue of Warrants entitling the holder on exercise to receive Common Stock and are issued or to be issued pursuant to a Warrant Agreement dated as of [] (the "Warrant Agreement"), duly executed and delivered by the Company to Continental Stock Transfer & Trust Company, a New York corporation, as warrant agent (the "Warrant Agent"), which Warrant Agreement is hereby incorporated by reference in and made a part of this instrument and is hereby referred to for a description of the rights, limitation of rights, obligations, duties and immunities thereunder of the Warrant Agent, the Company and the holders (the words "holders" or "holder" meaning the Registered Holders or Registered Holder, respectively) of the Warrants. A copy of the Warrant Agreement may be obtained by the holder hereof upon written request to the Company. Defined terms used in this Warrant Certificate but not defined herein shall have the meanings given to them in the Warrant Agreement.

Warrants may be exercised at any time during the Exercise Period set forth in the Warrant Agreement. The holder of Warrants evidenced by this Warrant Certificate may exercise them by surrendering this Warrant Certificate, with the form of election to purchase set forth hereon properly completed and executed, together with payment of the Warrant Price as specified in the Warrant Agreement (or through "cashless exercise" as provided for in the Warrant Agreement) at the principal corporate trust office of the Warrant Agent. In the event that upon any exercise of Warrants evidenced hereby the number of Warrants exercised shall be less than the total number of Warrants evidenced hereby, there shall be issued to the holder hereof or his, her or its assignee, a new Warrant Certificate evidencing the number of Warrants not exercised.

Notwithstanding anything else in this Warrant Certificate or the Warrant Agreement, no Warrant may be exercised unless at the time of exercise (i) a registration statement covering the Common Stock to be issued upon exercise is effective under the Securities Act of 1933, as amended, and (ii) a prospectus thereunder relating to the Common Stock is current, except through "cashless exercise" as provided for in the Warrant Agreement.

The Warrant Agreement provides that upon the occurrence of certain events the number of Common Stock issuable upon the exercise of the Warrants set forth on the face hereof may, subject to certain conditions, be adjusted. If, upon exercise of a Warrant, the holder thereof would be entitled to receive a fractional interest in a Common Stock, the Company shall, upon exercise, round down to the nearest whole number of Common Stock to be issued to the holder of the Warrant.

Warrant Certificates, when surrendered at the principal corporate trust office of the Warrant Agent by the Registered Holder thereof in person or by legal representative or attorney duly authorized in writing, may be exchanged, in the manner and subject to the limitations provided in the Warrant Agreement, but without payment of any service charge, for another Warrant Certificate or Warrant Certificates of like tenor evidencing in the aggregate a like number of Warrants.

Upon due presentation for registration of transfer of this Warrant Certificate at the office of the Warrant Agent, a new Warrant Certificate or Warrant Certificates of like tenor and evidencing in the aggregate a like number of Warrants shall be issued to the transferee(s) in exchange for this Warrant Certificate, subject to the limitations provided in the Warrant Agreement, without charge except for any tax or other governmental charge imposed in connection therewith.

The Company and the Warrant Agent may deem and treat the Registered Holder(s) hereof as the absolute owner(s) of this Warrant Certificate (notwithstanding any notation of ownership or other writing hereon made by anyone), for the purpose of any exercise hereof, of any distribution to the holder(s) hereof, and for all other purposes, and neither the Company nor the Warrant Agent shall be affected by any notice to the contrary. Neither the Warrants nor this Warrant Certificate entitles any holder hereof to any rights of a shareholder of the Company.

Election to Purchase

(To Be Executed Upon Exercise of Warrant)

The undersigned hereby irrevocably elects to exercise the right, represented by this Warrant Certificate, to receive [] Common Stock and herewith tenders payment for such Common Stock to the order of Innovid Corp. (the "Company") in the amount of \$[] in accordance with the terms hereof. The undersigned requests that a certificate for such Common Stock be registered in the name of [] whose address is [] and that such Common Stock be delivered to [] whose address is [] and that such Common Stock be delivered to [] whose address is [] and that such Warrant Certificate representing the remaining balance of such Common Stock be registered in the name of [], whose address is [] and that such Warrant Certificate be delivered to [], whose address is [] and that such Warrant Certificate be delivered to [], whose address is [].
In the event that the Warrant has been called for redemption by the Company pursuant to <u>Section 6</u> of the Warrant Agreement and the Company has required cashless exercise pursuant to <u>Section 6.3</u> of the Warrant Agreement, the number of Common Stock that this Warrant is exercisable for shall be determined in accordance with <u>subsection 3.3.1(b)</u> and <u>Section 6.3</u> of the Warrant Agreement.
In the event that the Warrant is a Private Placement Warrant that is to be exercised on a "cashless" basis pursuant to <u>subsection 3.3.1(c)</u> of the Warrant Agreement, the number of Common Stock that this Warrant is exercisable for shall be determined in accordance with <u>subsection 3.3.1(c)</u> of the Warrant Agreement.
In the event that the Warrant is to be exercised on a "cashless" basis pursuant to Section 7.4 of the Warrant Agreement, the number of Common Stock that this Warrant

is exercisable for shall be determined in accordance with Section 7.4 of the Warrant Agreement.

In the event that the Warrant may be exercised, to the extent allowed by the Warrant Agreement, through cashless exercise (i) the number of Common Stock that this

Warrant is exercised for would be determined in accordance with the relevant section of the Warrant Agreement, thick allows for such cashless exercise and (ii) the holder hereof shall complete the following: The undersigned hereby irrevocably elects to exercise the right, represented by this Warrant Certificate, through the cashless exercise provisions of the Warrant Agreement, to receive shares of Common Stock. If said number of shares is less than all of the Common Stock purchasable hereunder (after giving effect to the cashless exercise), the undersigned requests that a new Warrant Certificate representing the remaining balance of such Common Stock be registered in the name of [], whose address is [] and that such Warrant Certificate be delivered to [], whose address is [].

[Signature Page Follows]

Date:	,		
		(Signature)	
		(Address)	
		(Tax Identification Number)	
Signature Guarante	eed:		

THE SIGNATURE(S) SHOULD BE GUARANTEED BY AN ELIGIBLE GUARANTOR INSTITUTION (BANKS, STOCKBROKERS, SAVINGS AND LOAN ASSOCIATIONS AND CREDIT UNIONS WITH MEMBERSHIP IN AN APPROVED SIGNATURE GUARANTEE MEDALLION PROGRAM, PURSUANT TO SEC RULE 17Ad-15 UNDER THE SECURITIES EXCHANGE ACT OF 1934, AS AMENDED (OR ANY SUCCESSOR RULE).

INVESTOR RIGHTS AGREEMENT

THIS INVESTOR RIGHTS AGREEMENT (this "Agreement"), dated as of November 30, 2021, is made and entered into by and among:

- (1) Innovid Corp., a Delaware corporation (the "Company"), which was formerly named ION Acquisition Corp 2 Ltd. ("ION");
- (2) certain equityholders of ION listed on Schedule A hereto (the "SPAC Holders"); and
- (3) certain former equityholders of Innovid, Inc., a Delaware corporation ("Innovid"), listed on Schedule B hereto (the "Innovid Equityholders" and, together with the SPAC Holders and any Person who hereafter becomes a party to this Agreement pursuant to Section 7.2 of this Agreement, a "Holder" and collectively the "Holders").

RECITALS

WHEREAS, Innovid, ION, Inspire Merger Sub 1, Inc. ("Merger Sub 1"), and Inspire Merger Sub 2 LLC ("Merger Sub 2") have entered into that certain Agreement and Plan of Merger, dated as of June 24, 2021 (the "Merger Agreement"), pursuant to which, among other things, (i) ION migrated to and domesticated as a Delaware corporation, (ii) (x) Merger Sub 1 merged with and into Innovid (the "First Merger"), with Innovid continuing as the surviving corporation of the First Merger and becoming a wholly owned Subsidiary of the Company (Innovid, as the surviving corporation in the First Merger, is sometimes referred to herein as the "Surviving Corporation"), and (y) the Surviving Corporation merged with and into Merger Sub 2 (the "Second Merger"), with Merger Sub 2 continuing as the surviving entity of the Second Merger and (iii) ION changed its name to "Innovid Corp.";

WHEREAS, Innovid and certain of the Innovid Equityholders are parties to that certain Amended and Restated Investors Rights Agreement, dated January 7, 2019 (the "Prior Innovid Agreement");

WHEREAS, ION and certain of the SPAC Holders are parties to that certain Registration Rights Agreement, dated February 10, 2021 (the "Prior ION Agreement");

WHEREAS, Innovid and the Innovid Equityholders party to the Prior Innovid Agreement desire to terminate the Prior Innovid Agreement in its entirety and to accept the rights created pursuant to this Agreement in lieu of the rights granted to them under the Prior Innovid Agreement;

WHEREAS, ION and the SPAC Holders desire to terminate the Prior ION Agreement in its entirety and to accept the rights created pursuant to this Agreement in lieu of the rights granted to them under the Prior ION Agreement; and

WHEREAS, in connection with the consummation of the transactions described above, the Company and the Holders desire to enter into this Agreement, pursuant to which, among other things, the Company shall grant the Holders certain registration rights with respect to the Registrable Securities (as defined below) on the terms and conditions set forth in this Agreement.

NOW, **THEREFORE**, in consideration of the representations, covenants and agreements contained herein, and certain other good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, the parties hereto, intending to be legally bound, hereby agree as follows:

Article I. **DEFINITIONS**

- Section 1.1 Definitions. The terms defined in this Article I shall, for all purposes of this Agreement, have the respective meanings set forth below:
- "Adverse Disclosure" means any public disclosure of material non-public information, which disclosure, in the good faith judgment of the Chief Executive Officer of the Company, Chief Financial Officer of the Company or the Board, after consultation with counsel to the Company, (i) would be required to be made in any Registration Statement or Prospectus in order for the applicable Registration Statement or Prospectus not to contain any untrue statement of a material fact or omit to state a material fact necessary to make the statements contained therein (in the case of any prospectus and any preliminary prospectus, in the light of the circumstances under which they were made) not misleading, (ii) would not be required to be made at such time if the Registration Statement were not being filed, declared effective or used, as the case may be, and (iii) the Company has a *bona fide* business purpose for not making such information public.
- "Action" means any claim, action, suit, audit, examination, assessment, arbitration, mediation or inquiry, or any proceeding or investigation, by or before any Governmental Authority.
- "affiliate" of any particular person means any other person controlling, controlled by or under common control with such person, where "control" means the possession, directly or indirectly, of the power to direct the management and policies of a person whether through the ownership of voting securities, its capacity as a sole or managing member or otherwise.
 - "Agreement" shall have the meaning given in the Preamble hereto.
 - "Board" means the board of directors of the Company.
 - "Block Trade" shall have the meaning given in Section 2.4.1.
- "Business Day" means a day other than a Saturday, Sunday or other day on which commercial banks in New York, New York are authorized or required by Law to close.
- "Certificate of Incorporation" means the certificate of incorporation of the Company (as amended, modified or supplemented from time to time).
 - "Closing" shall have the meaning given in the Merger Agreement.
 - "Closing Date" shall have the meaning given in the Merger Agreement.
 - "Commission" means the Securities and Exchange Commission.
- "Company" shall have the meaning given in the Recitals hereto and includes the Company's successors by recapitalization, merger, consolidation, spin-off, reorganization or similar transaction.
 - "Company Common Stock" means the shares of common stock, par value \$0.0001, per share of the Company.
 - "Demanding Holder" shall have the meaning given in Section 2.1.4.
 - "Exchange Act" means the Securities Exchange Act of 1934, as it may be amended from time to time.
 - "FINRA" means the Financial Industry Regulatory Authority Inc.

- "First Merger" shall have the meaning given in the Recitals hereto.
- "Form S-1 Shelf" shall have the meaning given in Section 2.1.1.
- "Form S-3 Shelf" shall have the meaning given in Section 2.1.1.
- "Governmental Authority" means any federal, state, provincial, municipal, local or foreign government, governmental authority, regulatory or administrative agency (which for the purposes of this Agreement shall include FINRA and the Commission), governmental commission, department, board, bureau, agency or instrumentality, court or tribunal.
- "Governmental Order" means any order, judgment, injunction, decree, writ, stipulation, determination or award, in each case, entered by or with any Governmental Authority.
 - "Holder Information" shall have the meaning given in Section 4.1.2.
 - "Holders" shall have the meaning given in the Preamble hereto, for so long as such Person holds any Registrable Securities.
 - "Indemnification Sources" shall have the meaning given in Section 6.1.4.
 - "Indemnified Liabilities" shall have the meaning given in Section 6.1.1.
 - "Indemnitee-Related Entities" shall have the meaning given in Section 6.1.4.
 - "Independent Committee" means the audit committee of the Board.
 - "Innovid" shall have the meaning given in the Preamble hereto.
 - "Innovid Equityholders" shall have the meaning given in the Preamble hereto.
 - "ION" shall have the meaning given in the Preamble hereto.
 - "Jointly Indemnifiable Claims" shall have the meaning given in Section 6.1.4.
- "Law" means any statute, law, ordinance, rule, regulation or Governmental Order, in each case, of any Governmental Authority and shall include applicable case law interpreting the foregoing.
 - "Maximum Number of Securities" shall have the meaning given in Section 2.1.5.
 - "Merger Agreement" shall have the meaning given in the Recitals hereto.
 - "Merger Sub 1" shall have the meaning given in the Recitals hereto.
 - "Merger Sub 2" shall have the meaning given in the Recitals hereto.
 - "Minimum Takedown Threshold" shall have the meaning given in Section 2.1.4.
- "Misstatement" means an untrue statement of a material fact or an omission to state a material fact required to be stated in a Registration Statement or Prospectus or necessary to make the statements in a Registration Statement or Prospectus, (in the case of a Prospectus, in the light of the circumstances under which they were made) not misleading.
 - "Organizational Documents" shall have the meaning assigned to such term in the Merger Agreement.

"Permitted Transfer" means, with respect to any Holder, any Transfer (i) to any affiliate of such Holder, (ii) in the case of an individual, by gift a member of such individual's immediate family or a trust, the beneficiary of which is a member of such individual's immediate family, an affiliate of such individual, or to a charitable organization; (iii) in the case of an individual, by virtue of laws of descent and distribution upon death of such individual; (iv) in the case of an individual, pursuant to a qualified domestic relations order; (v) to a nominee or custodian of a Person to whom a Transfer would be permitted under clauses (i) through (iv) above; or (vi) to the Company; provided, however, that in each case of clauses (i) through (v), these Permitted Transferees must, before any such Transfer is effected, enter into a written agreement with the Company agreeing to be bound by this Agreement.

"Permitted Transferee" means, with respect to any Holder, any Person to whom such Holder is permitted to Transfer Registrable Securities through a Permitted Transfer prior to the expiration of the Lock-Up Termination Date (as defined in such Holder's Support Agreement).

"Person" means any individual, firm, corporation, partnership, limited liability company, incorporated or unincorporated association, joint venture, joint stock company, governmental authority or instrumentality or other entity of any kind.

"Piggyback Registration" shall have the meaning given in Section 2.2.1.

"Prior Innovid Agreement" shall have the meaning given in the Recitals hereto.

"Prior ION Agreement" shall have the meaning given in the Recitals hereto.

"Prospectus" means the prospectus included in any Registration Statement, as supplemented by any and all prospectus supplements and as amended by any and all post-effective amendments and including all material incorporated by reference in such prospectus.

"Registrable Security" means (a) any outstanding Company Common Stock or warrants to purchase Company Common Stock (including any Company Common Stock issued or issuable upon the exercise of any such warrant) held by a Holder immediately following the Closing (including Company Common Stock distributable pursuant to the Merger Agreement), (b) any Company Common Stock that may be acquired by Holders upon the exercise of a warrant or other right to acquire Company Common Stock held by a Holder immediately following the Closing, (c) any Company Common Stock or warrants to purchase Company Common Stock (including any Company Common Stock issued or issuable upon the exercise of any such warrant) of the Company owned by a Holder or otherwise acquired following the date hereof to the extent that such securities are "restricted securities" (as defined in Rule 144) or are otherwise held by an "affiliate" (as defined in Rule 144) of the Company and for so long as the Holder may be deemed to be an Underwriter pursuant to Rule 145(c), and (d) any other equity security of the Company or any of its subsidiaries issued or issuable with respect to any securities referenced in clause (a), (b) or (c) above by way of a stock dividend or stock split or in connection with a recapitalization, merger, consolidation, spin-off, reorganization or similar transaction; provided, however, that, as to any particular Registrable Security, such securities shall cease to be Registrable Securities upon the earliest to occur of: (A) a Registration Statement with respect to the sale of such securities shall have become effective under the Securities Act and such securities shall have been sold, transferred, disposed of or exchanged in accordance with such Registration Statement by the applicable Holder; (B) (i) such securities shall have been otherwise transferred, (ii) new certificates for such securities not bearing a legend restricting further transfer shall have been delivered by the Company, and (iii) subsequent public distribution of such securities shall not require registration under the Securities Act; (C) such securities shall have ceased to be outstanding; (D) such securities may be sold without registration pursuant to Section 4(a)(1) of the Securities Act or Rule 144 or Rule 145 promulgated under the Securities Act (or any successor rule promulgated under the Securities Act) (but with no volume or other restrictions or limitations including as to manner or timing of sale); and (E) such securities have been sold to, or through, a broker, dealer or underwriter in a public distribution or other public securities transaction.

"Registration" means a registration, including any related Shelf Takedown, effected by preparing and filing a Registration Statement, prospectus or similar document in compliance with the requirements of the Securities Act, and the applicable rules and regulations promulgated thereunder, and such Registration Statement becoming effective.

"Registration Expenses" means the expenses of a Registration, including, without limitation, the following:

- (A) all registration and filing fees (including fees with respect to filings required to be made with the Financial Industry Regulatory Authority, Inc.) and fees of any national securities exchange on which the Company Common Stock are then listed;
- (B) fees and expenses of compliance with securities or blue sky laws (including reasonable fees and disbursements of outside counsel for the Underwriters in connection with blue sky qualifications of Registrable Securities);
 - (C) printing, messenger, telephone and delivery expenses;
 - (D) reasonable fees and disbursements of counsel for the Company;
- (E) reasonable fees and disbursements of all independent registered public accountants of the Company incurred specifically in connection with such Registration; and
- (F) reasonable fees and expenses of one legal counsel selected by the majority-in-interest of the Demanding Holders in an Underwritten Offering (not to exceed \$200,000 without the consent of the Company).

"Registration Statement" means any registration statement that covers Registrable Securities pursuant to the provisions of this Agreement, including the Prospectus included in such registration statement, amendments (including post-effective amendments) and supplements to such registration statement or Prospectus, and all exhibits to and all materials incorporated by reference in such registration statement.

"Restricted Securities" shall have the meaning given in Section 5.1.1.

"Requesting Holders" shall have the meaning given in Section 2.1.5.

"Second Merger" shall have the meaning given in the Recitals hereto.

"Securities Act" means the Securities Act of 1933, as amended from time to time.

"Shelf' means the Form S-1 Shelf, the Form S-3 Shelf or any Subsequent Shelf Registration, as the case may be.

"Shelf Registration" means a registration of securities pursuant to a registration statement filed with the Commission in accordance with and pursuant to Rule 415 promulgated under the Securities Act (or any successor rule then in effect).

"Shelf Takedown" means an Underwritten Shelf Takedown or any proposed transfer or sale using a Registration Statement, including a Piggyback Registration.

"SPAC Holders" shall have the meaning given in the Preamble.

"Sponsor" means ION Holdings 2, LP, a Cayman Islands exempted limited partnership.

"Sponsor Indemnitees" shall have the meaning given in Section 6.1.1.

"Subsequent Shelf Registration" shall have the meaning given in Section 2.1.2.

"Subsidiary" means, with respect to any Person (for purposes of this definition, the "Controlling Company"), any other Person (i) of which a majority of the outstanding voting securities or other voting equity interests, or a majority of any other interests having the power to direct or cause the direction of the management and policies of such other Person, are owned, directly or indirectly, by the Controlling Company and/or (ii) with respect to which the Controlling Company or its Subsidiaries is a general partner or managing member, and, in each case of the foregoing clauses (i) and (ii), any predecessor or successor of such other Person.

"Support Agreement" means, (i) with respect to an Innovid Equityholder, if applicable, the Support Agreement, dated as of June 24, 2021, by and among such Innovid Equityholder, ION and Innovid, and (ii) with respect to a SPAC Holder, if applicable, the Support Agreement, dated as of June 24, 2021, by and among Sponsor, such SPAC Holder and certain other SPAC Holders, ION and Innovid.

"Surviving Corporation" shall have the meaning given in the Recitals hereto.

"Transfer" means the (a) sale or assignment of, offer to sell, contract or agreement to sell, hypothecate, pledge, grant of any option to purchase or otherwise dispose of or agreement to dispose of, directly or indirectly, or establishment or increase of a put equivalent position or liquidation with respect to or decrease of a call equivalent position within the meaning of Section 16 of the Exchange Act with respect to, any security, (b) entry into any swap or other arrangement that transfers to another, in whole or in part, any of the economic consequences of ownership of, or interest in, any security, whether any such transaction is to be settled by delivery of such securities, in cash or otherwise, or (c) public announcement of any intention to effect any transaction specified in clause (a) or (b)

"Underwriter" means a securities dealer who purchases any Registrable Securities as principal and not as part of such dealer's market-making activities.

"Underwritten Offering" means a Registration in which securities of the Company are sold to an Underwriter in a firm commitment underwriting for distribution to the public.

"Underwritten Shelf Takedown" shall have the meaning given in Section 2.1.4.

"Withdrawal Notice" shall have the meaning given in Section 2.1.6.

Article II. REGISTRATIONS AND OFFERINGS

Section 2.1 <u>Shelf Registration.</u>

2.1.1 Filing. The Company shall file within thirty (30) days after the Closing Date, and use commercially reasonable efforts to cause to be declared effective as soon as practicable thereafter, a Registration Statement for a Shelf Registration on Form S-1 (the "Form S-1 Shelf") or, if the Company is eligible to use a Registration Statement on Form S-3, a Shelf Registration on Form S-3 (the "Form S-3 Shelf"), in each case, covering the resale of all the Registrable Securities (determined as of two business days prior to such filing) on a delayed or continuous basis. Such Shelf shall provide for the resale of the Registrable Securities included therein pursuant to any method or combination of methods legally available to, and requested by, any Holder named therein. The Company shall maintain a Shelf in accordance with the terms hereof, and shall prepare and file with the SEC such amendments, including post-effective amendments, and supplements as may be necessary to keep a Shelf continuously effective, available for use and in compliance with the provisions of the Securities Act until such time as there are no longer any Registrable Securities. In the event the Company files a Form

- S-1 Shelf, the Company shall use its commercially reasonable efforts to convert the Form S-1 Shelf (and any Subsequent Shelf Registration) to a Form S-3 Shelf as soon as practicable after the Company is eligible to use Form S-3.
- 2.1.2 <u>Subsequent Shelf Registration</u>. If any Shelf ceases to be effective under the Securities Act for any reason at any time while Registrable Securities are still outstanding, the Company shall, subject to <u>Section 3.4</u>, use its commercially reasonable efforts to as promptly as is reasonably practicable cause such Shelf to again become effective under the Securities Act (including obtaining the prompt withdrawal of any order suspending the effectiveness of such Shelf), and shall use its commercially reasonable efforts to as promptly as is reasonably practicable amend such Shelf in a manner reasonably expected to result in the withdrawal of any order suspending the effectiveness of such Shelf or file an additional registration statement as a Shelf Registration (a "<u>Subsequent Shelf Registration</u>") registering the resale of all Registrable Securities (determined as of two business days prior to such filing), and pursuant to any method or combination of methods legally available to, and requested by, any Holder named therein.

If a Subsequent Shelf Registration is filed, the Company shall use its commercially reasonable efforts to (i) cause such Subsequent Shelf Registration to become effective under the Securities Act as promptly as is reasonably practicable after the filing thereof (it being agreed that the Subsequent Shelf Registration shall be an automatic shelf registration statement (as defined in Rule 405 promulgated under the Securities Act) if the Company is a well-known seasoned issuer (as defined in Rule 405 promulgated under the Securities Act) at the most recent applicable eligibility determination date) and (ii) keep such Subsequent Shelf Registration continuously effective, available for use and in compliance with the provisions of the Securities Act until such time as there are no longer any Registrable Securities. Any such Subsequent Shelf Registration shall be on Form S-3 to the extent that the Company is eligible to use such form. Otherwise, such Subsequent Shelf Registration shall be on another appropriate form.

- 2.1.3 Additional Registerable Securities. In the event that any Holder holds Registrable Securities that are not registered for resale on a delayed or continuous basis, the Company, upon request of any such Holder or Holders that hold in the aggregate at least one percent (1.0%) of the Registrable Securities, shall promptly use its commercially reasonable efforts to cause the resale of such Registrable Securities to be covered by either, at the Company's option, the Shelf (including by means of a post-effective amendment) or a Subsequent Shelf Registration and cause the same to become effective as soon as practicable after such filing and such Shelf or Subsequent Shelf Registration shall be subject to the terms hereof; provided, however, that the Company shall only be required to cause such Registrable Securities to be so covered twice per calendar year for the Innovid Equityholders, and the SPAC Holders, respectively.
- 2.1.4 Requests for Underwritten Shelf Takedowns. At any time and from time to time when an effective Shelf is on file with the Commission, any Innovid Equityholder or SPAC Holder (any of the Innovid_Equityholders or the SPAC Holders being, in such case, individually, jointly or collectively, a "Demanding Holder") may request to sell all or any portion of its Registrable Securities in an Underwritten Offering or Other Coordinated Offering that is registered pursuant to the Shelf (each, an "Underwritten Shelf Takedown"); provided that the Company shall only be obligated to effect an Underwritten Shelf Takedown if such offering shall include Registrable Securities proposed to be sold by the Demanding Holder, individually, jointly or collectively, with a total offering price reasonably expected to exceed, in the aggregate, \$100.0 million (the "Minimum Takedown Threshold"). All requests for Underwritten Shelf Takedowns shall be made by giving written notice to the Company, which shall specify the approximate number of Registrable Securities proposed to be sold in the Underwritten Shelf Takedown. Subject to Section 2.4.4, the Company shall select the Underwriters for such offering (which shall consist of one or more reputable internationally recognized investment banks), subject to the initial Demanding Holder's or Holders' prior approval (which shall not be unreasonably withheld, conditioned or delayed). The Innovid Equityholders or the SPAC Holders may each demand not more than two (2) Underwritten Shelf Takedowns pursuant to this Section 2.1.4 in

any 12-month period, *provided*, *however*, that this number of offerings may be increased by the decision of an Independent Committee. Notwithstanding anything to the contrary in this Agreement, the Company may effect any Underwritten Offering pursuant to any then effective Registration Statement, including a Form S-3, that is then available for such offering.

- 2.1.5 Reduction of Underwritten Offering. If the managing Underwriter or Underwriters in an Underwritten Shelf Takedown, in good faith, advises the Company, the Demanding Holder(s) and the Holders requesting piggy back rights pursuant to this Agreement with respect to such Underwritten Shelf Takedown (the "Requesting Holders") (if any) in writing that the dollar amount or number of Registrable Securities that the Demanding Holder(s) and the Requesting Holders (if any) desire to sell, taken together with all other Company Common Stock or other equity securities that the Company desires to sell and all Company Common Stock or other equity securities, if any, that have been requested to be sold in such Underwritten Offering pursuant to separate written contractual piggy-back registration rights held by any other stockholders, exceeds the maximum dollar amount or maximum number of equity securities that can be sold in the Underwritten Offering without adversely affecting the proposed offering price, the timing, the distribution method, or the probability of success of such offering (such maximum dollar amount or maximum number of such securities, as applicable, the "Maximum Number of Securities"), then the Company shall include in such Underwritten Offering, before including any Company Common Stock or other equity securities proposed to be sold by Company or by other holders of Company Common Stock or other equity securities, the Registrable Securities of the Demanding Holder(s) and the Requesting Holders (pro rata based on the respective number of Registrable Securities held by each Demanding Holder and Requesting Holder) that can be sold without exceeding the Maximum Number of Securities. To facilitate the allocation of Registrable Securities in accordance with the above provisions, the Company or the Underwriters may round the number of shares allocated to any Holder to the nearest 100 shares. The Company shall not be required to include any Registrable Securities in such Underwritten Shelf Takedown unless the
- 2.1.6 Withdrawal. Prior to the filing of the applicable "red herring" prospectus or prospectus supplement used for marketing such Underwritten Shelf Takedown, a majority-in-interest of the Demanding Holders initiating an Underwritten Shelf Takedown shall have the right to withdraw from such Underwritten Shelf Takedown for any or no reason whatsoever upon written notification (a "Withdrawal Notice") to the Company and the Underwriter or Underwriters (if any) of their intention to withdraw from such Shelf Takedown; provided that any Innovid Equityholder or SPAC Holder may elect to have the Company continue an Underwritten Shelf Takedown if the Minimum Takedown Threshold would still be satisfied by the Registrable Securities proposed to be sold in the Underwritten Shelf Takedown by the Innovid Equityholders or the SPAC Holders or any of their respective Permitted Transferees, as applicable. If withdrawn, a demand for an Underwritten Shelf Takedown shall constitute a demand for an Underwritten Shelf Takedown for purposes of Section 2.1.4, unless either (i) the Demanding Holder has not previously withdrawn any Underwritten Shelf Takedown or (ii) the Holder reimburses the Company for all Registration Expenses with respect to such Underwritten Shelf Takedown; provided that, if a Innovid Equityholder or a SPAC Holder elects to continue an Underwritten Shelf Takedown pursuant to the proviso in the immediately preceding sentence, such Underwritten Shelf Takedown shall instead count as an Underwritten Shelf Takedown demanded by the Innovid_Equityholders or the SPAC Holders, as applicable, for purposes of Section 2.1.4. Following the receipt of any Withdrawal Notice, the Company shall promptly forward such Withdrawal Notice to any other Holders that had elected to participate in such Shelf Takedown. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with a Shelf Takedown prior to its withdrawal under this Section 2.1.6, other than if a Demanding Holder elects to pay such Registration Expenses pursuant to clause (ii) of the second sentence of this Section 2.1.6.

Section 2.2 Piggyback Registration.

- 2.2.1 Piggyback Rights. Subject to Section 2.4.3, following the applicable Lock-Up Termination Date, if the Company or any Holder proposes to conduct a registered offering of, or if the Company proposes to file a Registration Statement under the Securities Act with respect to the Registration of, equity securities, or securities or other obligations exercisable or exchangeable for, or convertible into equity securities, for its own account or for the account of stockholders of the Company (or by the Company and by the stockholders of the Company including, without limitation, an Underwritten Shelf Takedown pursuant to Section 2.1 hereof), other than a Registration Statement (or any registered offering with respect thereto) (i) filed in connection with any employee stock option or other benefit plan, (ii) pursuant to a Registration Statement on Form S-4 (or similar form that relates to a transaction subject to Rule 145 under the Securities Act or any successor rule thereto), (iii) for an offering of debt that is convertible into equity securities of the Company or, (iv) for a dividend reinvestment plan or (v) for a rights offering, then the Company shall give written notice of such proposed offering to all of the Holders of Registrable Securities as soon as practicable but not less than five (5) days before the anticipated filing date of such Registration Statement or, in the case of an Underwritten Offering pursuant to a Shelf Registration, the applicable "red herring" prospectus or prospectus supplement used for marketing such offering, which notice shall (A) describe the amount and type of securities to be included in such offering, the intended method(s) of distribution, and the name of the proposed managing Underwriter or Underwriters, if any, in such offering, and (B) offer to all of the Holders of Registrable Securities the opportunity to include in such registered offering such number of Registrable Securities as such Holders may request in writing within two (2) Business Days after receipt of such written notice (such registered offering, a "Piggyback Registration"). Subject to Section 2.2.2, the Company shall, in good faith, cause such Registrable Securities to be included in such Piggyback Registration and, if applicable, shall use its commercially reasonable efforts to cause the managing Underwriter or Underwriters of such Piggyback Registration to permit the Registrable Securities requested by the Holders pursuant to this Section 2.2.1 to be included therein on the same terms and conditions as any similar securities of the Company included in such registered offering and to permit the sale or other disposition of such Registrable Securities in accordance with the intended method(s) of distribution thereof. The inclusion of any Holder's Registrable Securities in a Piggyback Registration shall be subject to such Holder's agreement to enter into an underwriting agreement in customary form with the Underwriter(s) selected for such Underwritten Offering.
- 2.2.2 <u>Reduction of Piggyback Registration</u>. If the managing Underwriter or Underwriters in an Underwritten Offering that is to be a Piggyback Registration, in good faith, advises the Company and the Holders of Registrable Securities participating in the Piggyback Registration in writing that the dollar amount or number of Company Common Stock or other equity securities that the Company desires to sell, taken together with (i) the Company Common Stock or other equity securities, if any, as to which Registration or a registered offering has been demanded pursuant to separate written contractual arrangements with persons or entities other than the Holders of Registrable Securities hereunder, (ii) the Registrable Securities as to which registration has been requested pursuant to <u>Section 2.2</u> hereof, and (iii) the Company Common Stock or other equity securities, if any, as to which Registration or a registered offering has been requested pursuant to separate written contractual piggy-back registration rights of other stockholders of the Company, exceeds the Maximum Number of Securities, then:
 - (a) If the Registration or registered offering is undertaken for the Company's account, the Company shall include in any such Registration or registered offering (A) first, the Company Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to Section 2.2.1, pro rata, based on the respective number of Registrable Securities held by each Holder that has requested to include Registrable Securities in such Underwritten Offering, which can be sold without exceeding the Maximum Number of Securities; and (C) third, to the extent that the Maximum Number of Securities has not been

reached under the foregoing clauses (A) and (B), the Company Common Stock or other equity securities, if any, as to which Registration or a registered offering has been requested pursuant to written contractual piggy-back registration rights of other stockholders of the Company, which can be sold without exceeding the Maximum Number of Securities;

- (b) If the Registration or registered offering is pursuant to a request by persons or entities other than the Holders of Registrable Securities, then the Company shall include in any such Registration or registered offering (A) first, the Company Common Stock or other equity securities, if any, of such requesting persons or entities, other than the Holders of Registrable Securities, which can be sold without exceeding the Maximum Number of Securities; (B) second, to the extent that the Maximum Number of Securities has not been reached under the foregoing clause (A), the Registrable Securities of Holders exercising their rights to register their Registrable Securities pursuant to Section 2.2.1, pro rata, based on the respective number of Registrable Securities held by each Holder that has requested to include Registrable Securities in such Underwritten Offering, which can be sold without exceeding the Maximum Number of Securities; (C) third, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A) and (B), the Company Common Stock or other equity securities that the Company desires to sell, which can be sold without exceeding the Maximum Number of Securities; and (D) fourth, to the extent that the Maximum Number of Securities has not been reached under the foregoing clauses (A), (B) and (C), the Company Common Stock or other equity securities for the account of other persons or entities that the Company is obligated to register pursuant to separate written contractual arrangements with such persons or entities, which can be sold without exceeding the Maximum Number of Securities; and
- (c) If the Registration or registered offering is pursuant to a request by Holder(s) of Registrable Securities pursuant to Section 2.1 hereof, then the Company shall include in any such Registration or registered offering securities pursuant to Section 2.1.5.
- 2.2.3 <u>Piggyback Registration Withdrawal</u>. Any Holder of Registrable Securities (other than a Demanding Holder, whose right to withdrawal from an Underwritten Shelf Takedown, and related obligations, shall be governed by <u>Section 2.1.6</u>) shall have the right to withdraw from a Piggyback Registration for any or no reason whatsoever upon written notification to the Company and the Underwriter or Underwriters (if any) of his, her or its intention to withdraw from such Piggyback Registration prior to the effectiveness of the Registration Statement filed with the Commission with respect to such Piggyback Registration or, in the case of a Piggyback Registration pursuant to a Shelf Registration, the filing of the applicable "red herring" prospectus or prospectus supplement with respect to such Piggyback Registration used for marketing such transaction. The Company (whether on its own good faith determination or as the result of a request for withdrawal by persons pursuant to separate written contractual obligations) may withdraw a Registration Statement filed with the Commission in connection with a Piggyback Registration (which, in no circumstance, shall include the Shelf) at any time prior to the effectiveness of such Registration Statement. Notwithstanding anything to the contrary in this Agreement (other than Section 2.1.6), the Company shall be responsible for the Registration Expenses incurred in connection with the Piggyback Registration prior to its withdrawal under this Section 2.2.3.
- 2.2.4 <u>Unlimited Piggyback Registration Rights</u>. For purposes of clarity, subject to <u>Section 2.1.6</u>, any Piggyback Registration effected pursuant to <u>Section 2.2</u> hereof shall not be counted as a demand for an Underwritten Shelf Takedown under <u>Section 2.1.4</u> hereof.

Section 2.3 <u>Market Stand-off.</u>

2.3.1 In connection with any Underwritten Offering of equity securities of the Company (other than a Block Trade), each Holder given an opportunity to participate in the Underwritten Offering pursuant to the terms of this Agreement agrees that it shall not Transfer any

Company Common Stock or other equity securities of the Company (other than those included in such offering pursuant to this Agreement), subject to customary exceptions including without limitation transfers to affiliates, gifts, exercise of options, without the prior written consent of the Company, during the 90-day period beginning on the date of pricing of such offering or such shorter period during which the Company agrees not to conduct an underwritten primary offering of Company Common Stock or other equity securities, except in the event the Underwriters managing the offering otherwise agree by written consent. Each Holder agrees to execute a customary lock-up agreement in favor of the Underwriters to such effect (in each case on substantially the same terms and conditions as all such Holders).

Section 2.4 <u>Block Trades.</u>

- 2.4.1 Notwithstanding the foregoing, at any time and from time to time when an effective Shelf is on file with the Commission and effective, if a Demanding Holder wishes to engage in (a) an underwritten block trade or similar transaction or other transaction with a two (2)-day or less marketing period (a "Block Trade") or (b) an "at the market" or similar registered offering through a broker, sales agent or distribution agent, whether as agent or principal, (an "Other Coordinated Offering"), in each case, with a total offering price reasonably expected to exceed, in the aggregate, either (x) \$50 million or (y) all remaining Registrable Securities held by the Demanding Holder, then notwithstanding the time periods provided for in Section 2.1.4, such Demanding Holder need only to notify the Company of the Block Trade at least five (5) days prior to the day such offering is to commence and the Company shall as expeditiously as possible use its commercially reasonable efforts to facilitate such Block Trade; provided that the Demanding Holders representing a majority of the Registrable Securities wishing to engage in the Block Trade shall use commercially reasonable efforts to work with the Company and any Underwriters prior to making such request in order to facilitate preparation of the registration statement, prospectus and other offering documentation related to the Block Trade.
- 2.4.2 Prior to the filing of the applicable "red herring" prospectus or prospectus supplement used in connection with a Block Trade, a majority-in-interest of the Demanding Holders initiating such Block Trade shall have the right to submit a Withdrawal Notice to the Company and the Underwriter or Underwriters (if any) of their intention to withdraw from such Block Trade. Notwithstanding anything to the contrary in this Agreement, the Company shall be responsible for the Registration Expenses incurred in connection with a block trade prior to its withdrawal under this Section 2.4.2.
- 2.4.3 Notwithstanding anything to the contrary in this Agreement, <u>Section 2.2</u> hereof shall not apply to a Block Trade initiated by a Demanding Holder pursuant to this Agreement.
- 2.4.4 The Demanding Holder in a Block Trade shall have the right to select the Underwriters for such Block Trade (which shall consist of one or more reputable internationally recognized investment banks).
- 2.4.5 A Holder in the aggregate may demand no more than two (2) Block Trades or Other Coordinated Offerings pursuant to this Section 2.4 in any twelve (12) month period.

Article III. COMPANY PROCEDURES

Section 3.1 <u>General Procedures.</u> In connection with any Shelf and/or Shelf Takedown, the Company shall use its commercially reasonable efforts to effect such Registration to permit the sale of such Registrable Securities in accordance with the intended plan of distribution thereof, and pursuant thereto the Company shall:

- 3.1.1 prepare and file with the Commission a Registration Statement with respect to such Registrable Securities and use its reasonable best efforts to cause such Registration Statement to become effective and remain effective until all Registrable Securities covered by such Registration Statement have ceased to be Registrable Securities;
- 3.1.2 prepare and file with the Commission such amendments and post-effective amendments to the Registration Statement, and such supplements to the Prospectus, as may be reasonably requested by any Holder or Holders that together with its or their respective affiliates hold at least one (1%) percent of the Registrable Securities registered on such Registration Statement or any Underwriter of Registrable Securities or as may be required by the rules, regulations or instructions applicable to the registration form used by the Company or by the Securities Act or rules and regulations thereunder to keep the Registration Statement effective until all Registrable Securities covered by such Registration Statement are sold in accordance with the intended plan of distribution set forth in such Registration Statement or supplement to the Prospectus and either (i) any underwriter overallotment option has terminated by its terms or (ii) the underwriters have advised the Company that they will not exercise such option or any remaining portion thereof;
- 3.1.3 prior to filing a Registration Statement or Prospectus, or any amendment or supplement thereto, furnish without charge to the Underwriters, if any, and the Holders of Registrable Securities included in such Registration, and such Holders' legal counsel, copies of such Registration Statement as proposed to be filed, each amendment and supplement to such Registration Statement (in each case including all exhibits thereto and documents incorporated by reference therein), the Prospectus included in such Registration Statement (including each preliminary Prospectus), and such other documents as the Underwriters and the Holders of Registrable Securities included in such Registration or the legal counsel for any such Holders may request in order to facilitate the disposition of the Registrable Securities owned by such Holders;
- 3.1.4 prior to any public offering of Registrable Securities (i) register or qualify the Registrable Securities covered by the Registration Statement under such securities or "blue sky" laws of such jurisdictions in the United States as the Holders of Registrable Securities included in such Registration Statement (in light of their intended plan of distribution) may reasonably request (or provide evidence satisfactory to such Holders that the Registrable Securities are exempt from registration or qualification) and (ii) take such action necessary to cause such Registrable Securities covered by the Registration Statement to be registered with or approved by such other governmental authorities as may be reasonably necessary by virtue of the business and operations of the Company and do any and all other acts and things that may be necessary or advisable to enable the Holders of Registrable Securities included in such Registration Statement to consummate the disposition of such Registrable Securities in such jurisdictions; provided, however, that the Company shall not be required to qualify generally to do business in any jurisdiction where it would not otherwise be required to qualify or take any action to which it would be subject to general service of process or taxation in any such jurisdiction where it is not then otherwise so subject;
- 3.1.5 cause all such Registrable Securities to be listed on each national securities exchange on which similar securities issued by the Company are then listed;
- 3.1.6 provide a transfer agent or warrant agent, as applicable, and registrar for all such Registrable Securities no later than the effective date of such Registration Statement;
- 3.1.7 advise each seller of such Registrable Securities, promptly after it shall receive notice or obtain knowledge thereof, of the issuance of any stop order by the Commission suspending the effectiveness of such Registration Statement or the initiation or threatening of any proceeding for such purpose and promptly use its reasonable best efforts to prevent the issuance of any stop order or to obtain its withdrawal if such stop order should be issued:

- 3.1.8 at least five (5) days prior to the filing of any Registration Statement or Prospectus or any amendment or supplement to such Registration Statement or Prospectus (or such shorter period of time as may be necessary in order to comply with the Securities Act, the Exchange Act, and the rules and regulations promulgated under the Securities Act or Exchange Act, as applicable), furnish a copy thereof to each seller of such Registrable Securities or its counsel (excluding any exhibits thereto and any filing made under the Exchange Act that is to be incorporated by reference therein);
- 3.1.9 notify the Holders at any time when a Prospectus relating to such Registration Statement is required to be delivered under the Securities Act, of the happening of any event as a result of which the Prospectus included in such Registration Statement, as then in effect, includes a Misstatement, and then to correct such Misstatement as set forth in Section 3.4 hereof;
- 3.1.10 permit a representative of any Holder, the Underwriters, if any, and any attorney or accountant retained by such Holder(s) or Underwriter to participate, at each such person's own expense, in the preparation of the Registration Statement, and cause the Company's officers, directors and employees to supply all information reasonably requested by any such representative, Underwriter, attorney or accountant in connection with the Registration; provided, however, that such representatives or Underwriters agree to confidentiality arrangements reasonably satisfactory to the Company, prior to the release or disclosure of any such information;
- 3.1.11 obtain a "comfort" letter from the Company's independent registered public accountants in the event of an Underwritten Offering or Other Coordinated Offering that is registered pursuant to a Registration Statement, in customary form and covering such matters of the type customarily covered by "comfort" letters as the managing Underwriter or other similar type of sales agent or placement agent may reasonably request, and reasonably satisfactory to a majority-in-interest of the participating Holders;
- 3.1.12 on the date the Registrable Securities are delivered for sale pursuant to such Registration, obtain an opinion, dated such date, of counsel representing the Company for the purposes of such Registration, addressed to the Holders, the placement agent or sales agent, if any, and the Underwriters, if any, covering such legal matters with respect to the Registration in respect of which such opinion is being given as the Holders, placement agent, sales agent, or Underwriter may reasonably request and as are customarily included in such opinions and negative assurance letters, and reasonably satisfactory to a majority-in-interest of the participating Holders;
- 3.1.13 in the event of any Underwritten Offering or Other Coordinated Offering that is registered pursuant to a Registration Statement, enter into and perform its obligations under an underwriting agreement, sales agreement or placement agreement, in usual and customary form, with the managing Underwriter, sales agent or placement agent of such offering;
- 3.1.14 make available to its security holders, as soon as reasonably practicable, an earnings statement covering the period of at least twelve (12) months beginning with the first day of the Company's first full calendar quarter after the effective date of the Registration Statement which satisfies the provisions of Section 11(a) of the Securities Act and Rule 158 thereunder (or any successor rule then in effect), and which requirement will be deemed to be satisfied if the Company timely files complete and accurate information on Forms 10-Q, 10-K and 8-K under the Exchange Act and otherwise complies with Rule 158 under the Securities Act;
- 3.1.15 if the Registration involves the Registration of Registrable Securities involving gross proceeds in excess of \$50 million with respect to an Underwritten Offering pursuant to Section 2.1.4, use its reasonable efforts to make available senior executives of the Company to participate in customary "road show" presentations that may be reasonably requested by the Underwritter in such Underwritten Offering; and

3.1.16 otherwise, in good faith, cooperate reasonably with, and take such customary actions as may reasonably be requested by the Holders, in connection with such Registration.

Notwithstanding the foregoing, the Company shall not be required to provide any documents or information to an Underwriter or other sales agent or placement agent has not then been named with respect to the applicable Underwritten Offering or Other Coordinated Offering that is registered pursuant to a Registration Statement.

- Section 3.2 <u>Registration Expenses</u>. The Registration Expenses of all Registrations shall be borne by the Company. It is acknowledged by the Holders that the Holders shall bear all incremental selling expenses relating to the sale of Registrable Securities, such as Underwriters' or agents' commissions and discounts, brokerage fees, Underwriter marketing costs and, other than as set forth in the definition of "*Registration Expenses*," all fees and expenses of any legal counsel representing the Holders.
- Section 3.3 Requirements for Participation in Underwritten Offerings. Notwithstanding anything in this Agreement to the contrary, if any Holder does not timely provide the Company with its requested Holder Information, the Company may exclude such Holder's Registrable Securities from the applicable Registration Statement or Prospectus if the Company determines, based on the advice of counsel, that such information is necessary to effect the registration and such Holder continues thereafter to withhold such information. No Person may participate in any Underwritten Offering or Other Coordinated Offering for equity securities of the Company pursuant to a Registration initiated by the Company hereunder unless such Person (i) agrees to sell such Person's securities on the basis provided in any arrangements approved by the Company and (ii) timely completes and executes all customary questionnaires, powers of attorney, indemnities, lock-up agreements, underwriting or other agreements and other customary documents as may be reasonably required under the terms of such arrangements. The exclusion of a Holder's Registrable Securities as a result of this Section 3.3 shall not affect the registration of the other Registrable Securities to be included in such Registration.

Section 3.4 <u>Suspension of Sales; Adverse Disclosure; Restrictions on Registration Rights.</u>

- 3.4.1 Upon receipt of written notice from the Company that a Registration Statement or Prospectus contains a Misstatement, each of the Holders shall forthwith discontinue disposition of Registrable Securities until it has received copies of a supplemented or amended Prospectus correcting the Misstatement (it being understood that the Company hereby covenants to prepare and file such supplement or amendment as soon as practicable after the time of such notice), or until it is advised in writing by the Company that the use of the Prospectus may be resumed.
- 3.4.2 If the filing, initial effectiveness or continued use of a Registration Statement in respect of any Registration at any time would (a) require the Company to make an Adverse Disclosure, (b) require the inclusion in such Registration Statement of financial statements that are unavailable to the Company for reasons beyond the Company's control, or (c) in the good faith judgment of the majority of the Board such Registration, cause serious and irreparable harm to the Company and the majority of the Board concludes as a result that it is essential to defer such filing, initial effectiveness or continued use at such time, the Company may, upon giving prompt written notice of such action to the Holders, delay the filing or initial effectiveness of, or suspend use of, such Registration Statement for the shortest period of time determined in good faith by the Company to be necessary for such purpose. In the event the Company exercises its rights under this Section 3.4.2, the Holders agree to suspend, immediately upon their receipt of the notice referred to above, their use of the Prospectus relating to any Registration in connection with any sale or offer to sell Registrable Securities, provided, however, that the Company shall not exercise its rights under this Section 3.4.2 on more than three (3) occasions or for more than sixty (60) consecutive calendar days, or more than one hundred and twenty (120) total calendar days, in each case during any twelve (12) month period.

- 3.4.3 (a) During the period starting with the date sixty (60) days prior to the Company's good faith estimate of the date of the filing of, and ending on a date one hundred and twenty (120) days after the effective date of, a Company-initiated Registration and provided that the Company continues to actively employ, in good faith, all commercially reasonable efforts to maintain the effectiveness of the applicable Shelf Registration Statement, or (b) if, pursuant to Section 2.1.4, Holders have requested an Underwritten Shelf Takedown and the Company and such Holders are unable to obtain the commitment of underwriters to firmly underwrite such offering, the Company may, upon giving prompt written notice of such action to the Holders, delay any other registered offering pursuant to Section 2.1.4 or 2.4.
- Section 3.5 Reporting Obligations. As long as any Holder shall own Registrable Securities, the Company, at all times while it shall be a reporting company under the Exchange Act, covenants to use commercially reasonable efforts to file timely (or obtain extensions in respect thereof and file within the applicable grace period) all reports required to be filed by the Company after the date hereof pursuant to Sections 13(a) or 15(d) of the Exchange Act and to promptly furnish the Holders with true and complete copies of all such filings; provided that any documents publicly filed or furnished with the Commission pursuant to the Electronic Data Gathering, Analysis and Retrieval System shall be deemed to have been furnished or delivered to the Holders pursuant to this Section 3.5. The Company further covenants that it shall take such further action as any Holder may reasonably request, all to the extent required from time to time to enable such Holder to sell Registrable Securities held by such Holder without registration under the Securities Act within the limitation of the exemptions provided by Section 4(a)(1) of the Securities Act or Rule 144 promulgated under the Securities Act (or any successor rule then in effect). Upon the request of any Holder, the Company shall deliver to such Holder a written certification of a duly authorized officer as to whether it has complied with such requirements.

Article IV. REGISTRATION RIGHTS INDEMNIFICATION AND CONTRIBUTION

Section 4.1 <u>Registration Rights Indemnification.</u>

- 4.1.1 The Company agrees to indemnify and hold harmless, to the extent permitted by law, each Holder of Registrable Securities, and each of such Holder's officers, directors, trustees, employees, partners, managers, members, equityholders, beneficiaries, affiliates and agents and each Person, if any, who controls such Holder, within the meaning of either Section 15 of the Securities Act or Section 20 of the Exchange Act, against all losses, claims, damages, liabilities and reasonable and documented out-of-pocket expenses (including without limitation outside attorneys' fees reasonably incurred) to which they may become subject under the Securities Act, the Exchange Act, or other federal or state law arising out of or based on any untrue statement (or alleged untrue statement) of a material fact contained in any Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto, or based on any omission (or alleged omission) to state therein a material fact required to be stated therein or necessary to make the statements therein not misleading in light of the circumstances in which they were made.
- 4.1.2 In connection with any Registration Statement in which a Holder of Registrable Securities is participating, such Holder shall furnish to the Company in writing such information with respect to such Holder as the Company reasonably requests for use in connection with any such Registration Statement or Prospectus (the "Holder Information") and, to the extent permitted by law, shall indemnify and hold harmless the Company, its directors, officers and agents and each Person who controls the Company (within the meaning of the Securities Act) against all losses, claims, damages, liabilities and documented out-of-pocket expenses reasonably incurred (including without limitation documented outside attorneys' fees reasonably incurred) arising out of or resulting from any untrue or alleged untrue statement of material fact contained in any Registration Statement, Prospectus or preliminary Prospectus or any amendment thereof or supplement thereto or any omission or alleged omission of a material fact required to be stated therein or necessary to make the statements therein not misleading, but only to the extent that such untrue or alleged untrue statement or omission or alleged

omission is contained in any information so furnished in writing by such Holder expressly for use therein; provided, however, that the obligation to indemnify shall be several, not joint and several, among such Holders of Registrable Securities, and the liability of each such Holder of Registrable Securities shall be in proportion to and limited to the net proceeds received by such Holder from the sale of Registrable Securities pursuant to such Registration Statement.

- 4.1.3 Any Person entitled to indemnification herein shall (i) give prompt written notice to the indemnifying party of any claim with respect to which it seeks indemnification (provided that the failure to give prompt notice shall not impair any Person's right to indemnification hereunder to the extent such failure has not materially prejudiced the indemnifying party) and (ii) unless in such indemnified party's reasonable judgment a conflict of interest between such indemnified and indemnifying parties may exist with respect to such claim, permit such indemnifying party to assume the defense of such claim with counsel reasonably satisfactory to the indemnified party. If such defense is assumed, the indemnifying party shall not be subject to any liability for any settlement made by the indemnified party without its consent (but such consent shall not be unreasonably withheld). An indemnifying party who is not entitled to, or elects not to, assume the defense of a claim shall not be obligated to pay the fees and expenses of more than one counsel for all parties indemnified by such indemnifying party with respect to such claim, unless in the reasonable judgment of any indemnified party a conflict of interest may exist between such indemnified party and any other of such indemnified parties with respect to such claim. No indemnifying party shall, without the consent of the indemnified party, consent to the entry of any judgment or enter into any settlement which cannot be settled in all respects by the payment of money (and such money is so paid by the indemnifying party pursuant to the terms of such settlement) and which settlement does not include as an unconditional term thereof the giving by the claimant or plaintiff to such indemnified party of a release from all liability in respect to such claim or litigation.
- 4.1.4 The indemnification provided for under this Agreement shall remain in full force and effect regardless of any investigation made by or on behalf of the indemnified party or any officer, director or controlling Person of such indemnified party and shall survive the transfer of securities. The Company and each Holder of Registrable Securities participating in an offering also agrees to make such provisions as are reasonably requested by any indemnified party for contribution to such party in the event the Company's or such Holder's indemnification is unavailable for any reason
- 4.1.5 If the indemnification provided under <u>Section 4.1</u> hereof from the indemnifying party is unavailable or insufficient to hold harmless an indemnified party in respect of any losses, claims, damages, liabilities and out-of-pocket expenses referred to herein, then the indemnifying party, in lieu of indemnifying the indemnified party, shall contribute to the amount paid or payable by the indemnified party as a result of such losses, claims, damages, liabilities and out-of-pocket expenses in such proportion as is appropriate to reflect the relative fault of the indemnifying party and the indemnified party, as well as any other relevant equitable considerations. The relative fault of the indemnifying party and indemnified party shall be determined by reference to, among other things, whether any action in question, including any untrue or alleged untrue statement of a material fact or omission or alleged omission to state a material fact, was made by, or relates to information supplied by, such indemnifying party or indemnified party, and the indemnifying party's and indemnified party's relative intent, knowledge, access to information and opportunity to correct or prevent such action; provided, however, that the liability of any Holder under this <u>Section 4.1.5</u> shall be limited to the amount of the net proceeds received by such Holder in such offering giving rise to such liability. The amount paid or payable by a party as a result of the losses or other liabilities referred to above shall be deemed to include, subject to the limitations set forth in <u>Sections 4.1.1</u>, <u>4.1.2</u> and <u>4.1.3</u> above, any legal or other fees, charges or out-of-pocket expenses reasonably incurred by such party in connection with any investigation or proceeding. The parties hereto agree that it would not be just and equitable if contribution pursuant to this <u>Section 4.1.5</u> were determined by *pro rata* allocation or by any other method of allocation, which does not take account of the equitable considerations referred to in this Sect

11(f) of the Securities Act) shall be entitled to contribution pursuant to this Section 4.1.5 from any Person who was not guilty of such fraudulent misrepresentation.

Article V <u>RESERVED.</u> Article VI. ADDITIONAL AGREEMENTS

Section 6.1 <u>Indemnification</u>; Exculpation.

6.1.1 As an inducement for the Sponsor to enter into this Agreement, subject in each case to restrictions under applicable Law, the Company will, indemnify, exonerate and hold (x) the Sponsor and (y) each of its managers, directors, officers, fiduciaries, managers, controlling Persons, employees, representatives and agents and each of the members, directors, officers, fiduciaries, managers, controlling Persons, employees and agents of each of the foregoing (collectively, the "Sponsor Indemnitees") free and harmless from and against any and all liabilities, losses, damages and out-of-pocket costs and expenses in connection therewith (including reasonable attorneys' fees and expenses) (collectively, "Liabilities") incurred by the Sponsor Indemnitees or any of them before the date of this Agreement (collectively, the "Indemnified Liabilities"), arising out of any Action arising directly or indirectly out of, or in any way relating to, Sponsor's control or management of the Company on or prior to the date of this Agreement, the business of the Company on or prior to the date of this Agreement, services provided by the Sponsor to the Company prior to the date of this Agreement, the Merger Agreement, the Transaction Agreements (as defined in the Merger Agreement) and the transactions and related filings contemplated by the Merger Agreement and the Transaction Agreements (other than, in each and every case, any Liabilities (x) to the extent arise out of any breach by any Sponsor Indemnitee of the Merger Agreement, any Transaction Agreement (as defined in the Merger Agreement) or any written agreement between such Sponsor Indemnitee, on the one hand, and the Company, on the other hand (in each case, to the extent any Sponsor Indemnitee is a party thereto) or, subject to applicable Law, the breach by any Sponsor Indemnitee of any fiduciary or other duty or obligation of any Sponsor Indemnitee to its direct or indirect equityholders, creditors or affiliates, (y) to the extent such Liabilities are directly caused by any Sponsor Indemnitee's gross negligence, bad faith or willful misconduct, or (z) to the extent such Liabilities arise out of any Sponsor Indemnitee's breach of applicable Law (collectively, "Excluded Liabilities"); provided, however, that if and to the extent that the foregoing indemnity undertaking in respect of Indemnified Liabilities may be unavailable or unenforceable for any reason, the Company will make the maximum contribution to the payment and satisfaction of each of the Indemnified Liabilities that is payable pursuant to this Section 6.1.1 to the extent permissible under applicable Law. For the purposes of this Section 6.1. none of the circumstances described in the limitations contained in the proviso in the immediately preceding sentence shall be deemed to apply absent a final non-appealable judgment of a court of competent jurisdiction to such effect, in which case to the extent any such limitation is so determined to apply to any Sponsor Indemnitee as to any previously advanced indemnity payments made by the Company, then such payments shall be promptly repaid by such Sponsor Indemnitee to the Company. Notwithstanding anything herein or otherwise to the contrary (a) for the avoidance of doubt, in no event shall Indemnified Liabilities include any Excluded Liabilities, and (b) in no event shall the Company or any of its affiliates be responsible for, and no Indemnified Liabilities will include, any special, indirect, incidental, consequential or punitive damages; provided that nothing in this clause (b) shall limit the indemnity and contribution obligations of the Company otherwise existing to the extent such special, indirect, incidental, consequential or punitive damages are actually payable to any third party (unaffiliated with any Sponsor Indemnitee). The rights of any Sponsor Indemnitee to indemnification hereunder will be in addition to any other rights any such Person may have under any other agreement or instrument to which such Sponsor Indemnitee is or becomes a party or is or otherwise becomes a beneficiary or under Law or under the Organizational Documents of the Company or its Subsidiaries, provided, however, that (x) each Sponsor Indemnitee shall use commercially reasonable efforts to mitigate the Indemnified Liabilities, including by using commercially reasonable efforts to pursue all applicable rights of

recovery or contribution available to each Sponsor Indemnitee from third parties (excluding Indemnitee-Related Entities (as defined below)) and making applicable claims under all available insurance policies and using commercially reasonable efforts to pursue such claims; and (y) any amount actually received by any Sponsor Indemnitee from any such other source (including the insurance policy) with respect to the Indemnified Liabilities shall be subject to the provisions of Section 6.1.7. Each party hereto agree that each of the Sponsor Indemnitees shall be third-party beneficiaries with respect to this Section 6.1, entitled to enforce this Section 6.1 as though each such Sponsor Indemnitee was a party to this Agreement.

- 6.1.2 If any Action shall be brought or asserted against any Sponsor Indemnitee in respect of which indemnification may be sought pursuant to this Section 6.1, such Sponsor Indemnitee shall promptly notify the Company in writing; provided, that the failure to notify the Company shall not relieve the Company from any liability that the Company may have under this Section 6.1 except to the extent that it has been materially prejudiced (through the loss (in whole or in part) or impairment of substantive rights or defenses) by such failure; and provided, further, that the failure to notify the Company shall not relieve it from any liability that it may have to a Sponsor Indemnitee otherwise than under the preceding paragraphs of this Section 6.1. If any such Action shall be brought or asserted against a Sponsor Indemnitee, the Company shall be entitled to participate therein and, absent the existence of an actual conflict of interests (at the reasonable advice of outside counsel to the Sponsor Indemnitee) between the Company and such Sponsor Indemnitee, to the extent that it shall wish, to assume the defense thereof (by providing notice of such election within thirty (30) days of receipt of notice of such Action from such Sponsor Indemnitee), with counsel reasonably satisfactory to the Sponsor Indemnitee and shall pay the reasonable fees and expenses of such counsel related to such Action, as incurred. After notice from the Company to a Sponsor Indemnitee of its election to assume the defense thereof, except as set forth in Section 6.1.3, the Company shall not be liable to such Sponsor Indemnitee under such subsection for any legal expenses of other counsel or any other expenses, in each case subsequently incurred by such Sponsor Indemnitee, in connection with the defense thereof other than reasonable costs of investigation. Each Sponsor Indemnitee shall reasonably cooperate with the Company and its representatives and advisors and shall give the Company and its representatives and advisors commercially reasonable access to all information, documents and files within such Sponsor Indemnitee's custody and control, and to relevant witnesses with respect to any claim that in respect of which indemnification may be sought pursuant to this Section 6.1, in each case, solely to the extent reasonable and necessary to defend any such Action; provided, that the parties hereto shall use commercially reasonable efforts to avoid the production of any information provided pursuant to this Section 6.1 (consistent with applicable Law), and to cause all communications among employees, counsel and others representing either party to any such Action to be made so as to preserve any applicable attorney-client or work-product privileges.
- 6.1.3 In any such Action, any Sponsor Indemnitee shall have the right to retain its own counsel, but the fees and expenses of such counsel shall be at the expense of such Sponsor Indemnitee unless (i) the Company and the Sponsor Indemnitee shall have mutually agreed to the contrary; (ii) the Company has failed within a reasonable time to retain counsel reasonably satisfactory to the Sponsor Indemnitee; or (iii) there are actual conflicts of interests (at the reasonable advice of outside counsel to the Sponsor Indemnitee) between the Sponsor Indemnitee and the Company or (y) there are one or more different defenses that conflict with respect to such Action that would otherwise not be available to the Company or the Sponsor Indemnitee. It is understood and agreed that the Company shall not, in connection with any Action or related Action in the same jurisdiction, be liable for the reasonable fees and expenses of more than one separate firm (in addition to any local counsel which shall be limited to one firm in each jurisdiction) for all Sponsor Indemnitees, and that all such reasonable fees and expenses shall be paid or reimbursed as they are incurred upon receipt from the Sponsor Indemnitee of a written request for payment thereof accompanied by a written statement with reasonable, non-privileged supporting detail of such fees and expenses. The Company or its Subsidiaries, in the defense of any Action for which a Sponsor Indemnitee would be entitled to indemnification under the terms of this Section 6.1, shall not, without the consent of such Sponsor

Indemnitee, such consent not to be unreasonably conditioned, withheld or delayed, enter into any settlement unless it (a) includes as a term thereof the giving by the claimant or plaintiff or class therein to such Sponsor Indemnitee of an unconditional release from all liability with respect to such Action, (b) does not impose any non-monetary limitations (equitable or otherwise) on such Sponsor Indemnitee, and (c) does not include a statement as to or an admission of fault, culpability or a failure to act by or on behalf of such Sponsor Indemnitee, and provided, that the monetary consideration for such settlement will be paid in full by the Company or its Subsidiaries.

- 6.1.4 The Company acknowledges and agrees that the Company shall be fully and primarily responsible for the payment to any Sponsor Indemnitee in respect of Indemnified Liabilities in connection with any Jointly Indemnifiable Claims (as defined below), pursuant to and in accordance with (as applicable) the terms of, (i) the Laws of the jurisdiction of incorporation of the Company, (ii) any director indemnification agreement, (iii) this Agreement, any other agreement between the Company or any of its Subsidiaries and such Sponsor Indemnitee pursuant to which such Sponsor Indemnitee is indemnified, (iv) the Laws of the jurisdiction of incorporation or organization of any Subsidiary of the Company and/or (v) the Organizational Documents of the Company's Subsidiaries ((i) through (v) collectively, the "<u>Indemnification Sources</u>"), irrespective of any right of recovery such Sponsor Indemnitee (or its affiliates) may have from any corporation, limited liability company, partnership, joint venture, trust, employee benefit plan or other enterprise (other than the Company, any of its Subsidiaries or the insurer under and pursuant to any insurance policy of the Company or any of its Subsidiaries) from whom such Sponsor Indemnitee may be entitled to indemnification with respect to which, in whole or in part, the Company or any of its Subsidiaries may also have an indemnification obligation (collectively, the "Indemnitee-Related Entities"). Under no circumstance, except in connection or as related to Excluded Claims, shall the Company or any of its Subsidiaries be entitled to any right of subrogation or contribution by the Indemnitee-Related Entities and no right of advancement or recovery any Sponsor Indemnitee may have from the Indemnitee-Related Entities shall reduce or otherwise alter the rights of such Sponsor Indemnitee or the obligations of the Company or any of its Subsidiaries under the Indemnification Sources. In the event that any of the Indemnitee-Related Entities shall make any payment to any Sponsor Indemnitee in respect of indemnification with respect to any Jointly Indemnifiable Claim as a result of the Company's failure to comply with its obligations under this Article VI, the Company shall reimburse the Indemnitee-Related Entity making such payment to the extent of such payment promptly upon written demand from such Indemnitee-Related Entity; provided that the Company shall be obligated to reimburse any Indemnitee-Related Entity pursuant to Section 6.1.4 only if, when and to the extent, (i) the Company is required pursuant to one or more Indemnification Sources to make a payment to any Sponsor Indemnitee with respect to a Jointly Indemnifiable Claim pursuant to this Article VI, (ii) the Company has not made such payment to such Sponsor Indemnitee, and (iii) the Indemnitee-Related Entity has made such payment to or on behalf of such Sponsor Indemnitee. For purposes of this Section 6.1.4, the term "Jointly Indemnifiable Claims" shall be broadly construed and shall include, without limitation, any Indemnified Liabilities for which any Sponsor Indemnitee shall be entitled to indemnification from both (1) the Company and/or any of its Subsidiaries pursuant to the Indemnification Sources, on the one hand, and (2) any Indemnitee-Related Entity pursuant to any other agreement between any Indemnitee-Related Entity and such Sponsor Indemnitee (or its affiliates) pursuant to which such Sponsor Indemnitee is indemnified, the Laws of the jurisdiction of incorporation or organization of any Indemnitee-Related Entity and/or the Organizational Documents of any Indemnitee-Related Entity, on the other hand.
- 6.1.5 In no event shall any Sponsor Indemnitee be liable to the Company or any of its Subsidiaries for any act, alleged act, omission or alleged omission that does not constitute a breach of applicable Law, gross negligence, bad faith or willful misconduct of such Sponsor Indemnitee as determined by a final, nonappealable determination of a court of competent jurisdiction.
- 6.1.6 Notwithstanding anything to the contrary contained in this Investor Rights Agreement, for purposes of this Section 6.1, the term Sponsor Indemnitees shall not include any Sponsor or its any of its partners, equityholders, members, affiliates, directors, officers, fiduciaries,

managers, controlling Persons, employees and agents or any of the partners, equityholders, members, affiliates, directors, officers, fiduciaries, managers, controlling Persons, employees and agents of any of the foregoing who, in each case, is an officer or director of the Company or any of its subsidiaries in such capacity as officer or director. Such officers and directors are or will be subject to separate indemnification in such capacity through this Agreement and/or the Organizational Documents and other agreements and instruments of the Company and its Subsidiaries.

- 6.1.7 The rights of any Sponsor Indemnitee to indemnification pursuant to this Section 6.1 will be in addition to any other rights any such Person may have under any other section of this Agreement or any other agreement or instrument to which such Sponsor Indemnitee is or becomes a party or is or otherwise becomes a beneficiary or under Law or under the Organizational Documents of the Company and its Subsidiaries. Notwithstanding the foregoing provisions of this Article VI, all payments to be made by the Company and its Subsidiaries pursuant to the foregoing provisions of this Article VI shall be limited to the amount of any Indemnified Liabilities that remains after deducting therefrom any insurance proceeds and any indemnity, contribution or other similar payment actually received by the Sponsor Indemnitee from any third parties (other than the Company and its Subsidiaries) in respect of any such Action, net of any out-of-pocket costs and expenses of recovery and the amount of any deductibles or retentions. If any Sponsor Indemnitee receives any payment from the Company or its Subsidiaries in respect of any Indemnified Liability and the Sponsor Indemnitee recovers from a third party insurance proceeds or any other amount in respect of the underlying claim or demand asserted pursuant to this Article VI against the Company or such Subsidiary, such Sponsor Indemnitee shall, as soon as reasonably practicable, pay over to the Company or such Subsidiary such insurance proceeds or other amount so recovered (after deducting therefrom the amount of reasonable and documented out-of-pocket costs and expenses incurred by it in procuring such recovery, and the amount of any deductibles or retentions,), but not in excess of the sum of any amount previously paid by the Company and its Subsidiaries to or on behalf of the Sponsor Indemnitee in respect of such claim.
- 6.1.8 Notwithstanding anything herein to the contrary, to the extent that the Company is unable to fulfill its obligations, in whole or in part, under this Section 6.1 as a result of applicable Law, then the Company shall cause its applicable Subsidiaries who are not similarly constrained or restricted by applicable Law to fulfill such obligations of behalf of the Company under this Section 6.1.

Article VII. MISCELLANEOUS

- Section 7.1 Notices. All notices, requests, claims, demands and other communications among the parties shall be in writing and shall be deemed to have been duly given (i) when delivered in person, (ii) when delivered after posting in the United States mail having been sent registered or certified mail return receipt requested, postage prepaid, (iii) when delivered by FedEx or other nationally recognized overnight delivery service or (iv) when e-mailed during normal business hours (and otherwise as of the immediately following Business Day), addressed as follows:
 - (a) If to the Company to:

Innovid Corp.
30 Irving Place, 12th floor
New York, NY 10003
Attention: Nabilah Irshad
Email: nabilah@innovid.com

with a copy (which shall not constitute notice) to:

Latham & Watkins LLP 1271 Avenue of the Americas New York, New York 10020 Attention: Eyal Orgad and Michael Vardanian

Email: Eyal.Orgad@lw.com; Michael.Vardanian@lw.com

(b) If to Sponsor:

ION Holdings 2, LP 89 Medinat Hayehudim Street Herzliya 4676672, Israel Attention: Anthony Reich Email: anthony@ion-am.com

with a copy (which shall not constitute notice) to:

White & Case LLP
1221 Avenue of the Americas
New York, New York 10020
Attn: Colin Diamond, Robert Chung and Jason Rocha
Empile diamond (Nubit tops of the Property of the Prop

E-mail: cdiamond@whitecase.com, robert.chung@whitecase.com, jason.rocha@whitecase.com

- (c) If to a SPAC Holder, to the address or contact information as set forth in the Company's books and records.
- (d) If to an Innovid Equityholder, to the address or contact information as set forth in the Company's books and records.
- Section 7.2 <u>Assignment; No Third Party Beneficiaries.</u>
- 7.2.1 This Agreement and the rights, duties and obligations of the Company hereunder may not be assigned or delegated by the Company in whole or in part.
- 7.2.2 A Holder may assign or delegate such Holder's rights, duties or obligations under this Agreement, in whole or in part, to any Person to whom it transfers Registrable Securities; provided that such Registrable Securities remain Registrable Securities following such transfer and such Person agrees to become bound by the terms and provisions of this Agreement.
- 7.2.3 No assignment by any party hereto of such party's rights, duties and obligations hereunder shall be binding upon or obligate the Company unless and until the Company shall have received (i) written notice of such assignment as provided in <u>Section 7.1</u> hereof and (ii) the written agreement of the assignee, in a form reasonably satisfactory to the Company, to be bound by the terms and provisions of this Agreement (which may be accomplished by the Addendum Agreement set forth in <u>Exhibit A</u>).
- 7.2.4 Subject to the foregoing, this Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective permitted successors and assigns. Any attempted assignment in violation of the terms of this <u>Section 7.2</u> shall be null and void, *ab initio*.
- 7.2.5 This Agreement shall not confer any rights or benefits on any Persons that are not parties hereto, other than as expressly set forth in this Agreement and Section 7.2 hereof.
- Section 7.3 <u>Captions; Counterparts.</u> The headings, subheadings and captions contained in this Agreement are inserted for convenience only and shall not affect in any way the meaning or interpretation of this Agreement. This Agreement and any amendment hereto may be executed in one or more counterparts, each of which shall be deemed to be an original, but all of which shall constitute

one and the same agreement. Delivery of an executed counterpart of a signature page to this Agreement or any amendment hereto by electronic means, including docusign, e-mail, or scanned pages shall be effective as delivery of a manually executed counterpart to this Agreement or any amendment hereto.

Section 7.4 <u>Governing Law.</u> This Agreement, and all claims or causes of action based upon, arising out of, or related to this Agreement, shall be governed by and construed in accordance with the laws of the State of Delaware, without giving effect to any choice of law or conflict of law provision or rule (whether of the State of Delaware or any other jurisdiction) that would cause the application of the law of any jurisdiction other than the State of Delaware.

Section 7.5 Jurisdiction; Waiver of Jury Trial.

- 7.5.1 Each of the parties hereto irrevocably and unconditionally submits to the exclusive jurisdiction of the Chancery Court of the State of Delaware (or, if the Chancery Court of the State of Delaware declines to accept jurisdiction, any state or federal court located within the State of Delaware), for the purposes of any Action, claim, demand, action or cause of action (a) arising under this Agreement or (b) in any way connected with or related or incidental to the dealings of the parties hereto in respect of this Agreement, and irrevocably and unconditionally waives any objection to the laying of venue of any such Action in any such court, and further irrevocably and unconditionally waives and agrees not to plead or claim in any such court that any such Action has been brought in an inconvenient forum. Each party hereby irrevocably and unconditionally waives, and agrees not to assert, by way of motion or as a defense, counterclaim or otherwise, in any Action, claim, demand, action or cause of action against such party (i) arising under this Agreement or (ii) in any way connected with or related or incidental to the dealings of the parties hereto in respect of this Agreement, (A) any claim that such party is not personally subject to the jurisdiction of the courts as described in this Section 7.5 for any reason, (B) that such party or such party's property is exempt or immune from the jurisdiction of any such court or from any legal process commenced in such courts (whether through service of notice, attachment prior to judgment, attachment in aid of execution of judgment, execution of judgment or otherwise) and (C) that (x) the Action, claim, demand, action or cause of action in any such court is brought against such party in an inconvenient forum, (y) the venue of such Action, claim, demand, action or cause of action against such party is improper or (z) this Agreement, or the subject matter hereof, may not be enforced against such party in or by such courts. Each party agrees that service of any process, summons, notice or document by registered mail to such party's respective address set forth in Section 7.5 shall be effective service of process for any such Action, claim, demand, action or cause of action.
- 7.5.2 THE PARTIES HERETO EACH HEREBY WAIVES, TO THE FULLEST EXTENT PERMITTED BY LAW, ANY RIGHT TO TRIAL BY JURY OF ANY PROCEEDING, CLAIM, DEMAND, ACTION, OR CAUSE OF ACTION (I) ARISING UNDER THIS AGREEMENT OR UNDER ANY ANCILLARY DOCUMENT OR (II) IN ANY WAY CONNECTED WITH OR RELATED OR INCIDENTAL TO THE DEALINGS OF THE PARTIES HERETO IN RESPECT OF THIS AGREEMENT, WHETHER NOW EXISTING OR HEREAFTER ARISING, AND WHETHER IN CONTRACT, TORT, EQUITY, OR OTHERWISE. THE PARTIES HERETO EACH HEREBY AGREES AND CONSENTS THAT ANY SUCH PROCEEDING, CLAIM, DEMAND, ACTION OR CAUSE OF ACTION SHALL BE DECIDED BY COURT TRIAL WITHOUT A JURY AND THAT THE PARTIES HERETO MAY FILE AN ORIGINAL COUNTERPART OF A COPY OF THIS AGREEMENT WITH ANY COURT AS WRITTEN EVIDENCE OF THE CONSENT OF THE PARTIES HERETO TO THE WAIVER OF THEIR RIGHT TO TRIAL BY JURY. EACH PARTY CERTIFIES AND ACKNOWLEDGES THAT (A) NO REPRESENTATIVE, AGENT OR ATTORNEY OF ANY OTHER PARTY HAS REPRESENTED, EXPRESSLY OR OTHERWISE, THAT SUCH OTHER PARTY WOULD NOT, IN THE EVENT OF LITIGATION, SEEK TO ENFORCE THE FOREGOING WAIVER, (B) EACH SUCH PARTY UNDERSTANDS AND HAS CONSIDERED THE IMPLICATIONS OF THIS WAIVER, (C) EACH SUCH PARTY MAKES THIS WAIVER VOLUNTARILY AND (D) EACH SUCH PARTY HAS BEEN INDUCED TO ENTER

INTO THIS AGREEMENT BY, AMONG OTHER THINGS, THE MUTUAL WAIVERS AND CERTIFICATIONS IN THIS SECTION 7.5.

- Section 7.6 Amendments and Modifications. Upon the written consent of (a) the Company and (b) the Holders of a majority of the total Registrable Securities, compliance with any of the provisions, covenants and conditions set forth in this Agreement may be waived, or any of such provisions, covenants or conditions may be amended or modified; provided, however, that in the event any such waiver, amendment or modification (including with respect to any defined term as used therein, whether or not such defined term is defined therein) would be adverse in any material respect to the material rights or obligations hereunder of a Holder of at least five (5.0%) percent of the Registrable Securities, the written consent of such Holder will also be required; provided further that in the event any such waiver, amendment or modification (including with respect to any defined term as used therein, whether or not such defined term is defined therein) would be disproportionate and adverse in any material respect to the material rights or obligations hereunder of a Holder, the written consent of such Holder will also be required; provided further that and amendment or modification to, or waiver of, Article VI (including with respect to any defined term as used therein, whether or not such defined term is defined therein) that adversely affects any right granted to the Sponsor shall require the prior written consent of the Sponsor. No course of dealing between any Holder or the Company and any other party hereto or any failure or delay on the part of a Holder or the Company in exercising any rights or remedies under this Agreement shall operate as a waiver of any rights or remedies of any Holder or the Company. No single or partial exercise of any rights or remedies under this Agreement by a party shall operate as a waiver or preclude the exercise of any other rights or remedies hereunder or thereunder by such party.
- Section 7.7 <u>Termination of Existing Registration Rights.</u> The registration rights granted under this Agreement shall supersede any registration, qualification or similar rights of the Holders with respect to any shares or securities of ION or the Company granted under any other agreement, including, but not limited to, the Prior ION Agreement, the Prior Innovid Agreement, and any of such preexisting registration, qualification or similar rights and such agreements shall be terminated and of no further force and effect.
- Section 7.8 <u>Term.</u> This Agreement shall terminate with respect to any Holder on the date that such Holder no longer holds any Registrable Securities. The provisions of <u>Sections 3.5, 7.1, 7.4, 7.5, 7.8, 7.10, 7.11</u>, and <u>Article V</u> and <u>Article VI</u> shall survive any termination. In the event the Merger Agreement is terminated in accordance with its terms, this Agreement shall automatically terminate and be of no further force and effect.
- Section 7.9 <u>Holder Information.</u> Each Holder agrees, if requested in writing, to represent to the Company the total number of Registrable Securities held by such Holder in order for the Company to make determinations hereunder.
- Section 7.10 Severability. This Agreement shall be deemed severable, and the invalidity or unenforceability of any term or provision hereof shall not affect the validity or enforceability of this Agreement or of any other term or provision hereof. Furthermore, in lieu of any such invalid or unenforceable term or provision, the parties hereto intend that there shall be added as a part of this Agreement a provision as similar in terms to such invalid or unenforceable provision as may be possible that is valid and enforceable.
- Section 7.11 <u>Entire Agreement.</u> This Agreement (including all agreements entered into pursuant hereto and all certificates and instruments delivered pursuant hereto and thereto) constitute the entire agreement of the parties with respect to the subject matter hereof and supersede all prior and contemporaneous agreements, representations, understandings, negotiations and discussions between the parties, whether oral or written, including without limitation the Prior Innovid Agreement and the Prior ION Agreement.

[Remainder of the page intentionally left blank]

IN WITNESS WHEREOF, each of the undersigned has caused this Agreement to be executed as of the date first written above.

	COMPANY:
	INNOVID CORP.
	By: Name: Title:
	HOLDERS:
	ION HOLDINGS 2, LP By its General Partner, ION ACQUISITION CORP 2 LTD.
	By: Name: Title:
	Name: Jonathan Kolber
	Name: Gilad Shany
	Name: Avrom Gilbert
	Name: Anthony Reich
	Name: Gabriel Seligsohn
	Name: Rinat Gazit
	Name: Lior Shemesh
Signature Page to	Investor Rights Agreement

NEWSPRING GROWTH CAPITAL III-A, L.P.
Ву:
Name: Title:
SPECIAL SITUATIONS INVESTING GROUP II, LLC
Ву:
Name: Title:
AMOS AND DAUGHTERS INVESTMENTS AND PROPERTIES LTD.
Ву:
Name: Title:
CERCA PARTNERS L.P.
Ву:
Name: Title:
CISCO SYSTEMS INC.
Ву:
Name: Title:
GENESIS PARTNERS III L.P.
Ву:
Name: Title:
IBEX ISRAEL FUND LLLP
Ву:
Signature Page to Investor Rights Agreement

Name: Title:	
LAUDER	DALE GMBH & CO.KG
By:	
Name: Title:	
SEQUOIA	CAPITAL ISRAEL IV HOLDINGS L.P.
By:	
Name: Title:	
VINTAGI	E CO-INVESTMENT FUND I (CAYMAN) L.P.
By:	
Name: Title:	
VINTAGI	C CO-INVESTMENT FUND I (ISRAEL) L.P.
By:	
Name: Title:	
ZOHAR (GILON LTD.
By:	
Name: Title:	
MR. ZVII	KA NETTER

Signature Page to Investor Rights Agreement

Mr. TAL CHALOZIN	
Signature Page to Investor Rights Agreement	

Schedule A

SPAC Holders

<u>Name</u>
ION Holdings 2, LP
Jonathan Kolber
Gilad Shany
Avrom Gilbert
Anthony Reich
Gabriel Seligsohn
Rinat Gazit
Lior Shemesh

Schedule B

Innovid Equityholder

Name Name
NewSpring Growth Capital III-A, L.P.
Special Situations Investing Group II. LLC
Amos and Daughters Investments and Properties Ltd.
Cerca Partners L.P.
Cisco Systems Inc.
Genesis Partners III L.P.
Ibex Israel Fund LLLP
Lauderdale GMBH & CO.KG
Sequoia Capital Israel IV Holdings L.P.
Vintage Co-Investment Fund I (Cayman) L.P.
Vintage Co-Investment Fund I (Israel) L.P.
Zohar Gilon Ltd.
Mr. Zvika Netter
Mr. Tal Chalozin
Tal Chalozin 2021 Family Trust #1
Tal Chalozin 2021 Family Trust #2
Zvika Netter 2021 Family Trust #1
Zvika Netter 2021 Family Trust #2
Zvika Netter 2021 Family Trust #3
Vintage Secondary Fund II (Cayman) L.P.
Vintage Secondary Fund II (Israel) L.P.
Vintage Secondary Fund III (Cayman) L.P.
Vintage Secondary Fund III (Israel) L.P.

EXHIBIT A

Addendum Agreement

This Addendum Agreement ("<u>Addendum Agreement</u>") is executed on [], 20[], by the undersigned (the "<u>New Holder</u>") pursuant to the terms of that certain Investor Rights Agreement dated as of [], 2021 (the "<u>Agreement</u>"), by and among the Company and the other parties thereto, as such Agreement may be amended, supplemented or otherwise modified from time to time. Capitalized terms used but not defined in this Addendum Agreement shall have the respective meanings ascribed to such terms in the Agreement. By the execution of this Addendum Agreement, the New Holder agrees as follows:

- 1. <u>Acknowledgment.</u> New Holder acknowledges that New Holder is acquiring certain Company Common Stock of the Company (the "<u>Shares</u>") as a transferee of such Shares from a party in such party's capacity as a holder of Registrable Securities under the Agreement, and after such transfer, New Holder shall be considered an "Investor" and a holder of Registrable Securities for all purposes under the Agreement.
- 2. <u>Agreement</u>. New Holder hereby (a) agrees that the Shares shall be bound by and subject to the terms of the Agreement and (b) adopts the Agreement with the same force and effect as if the New Holder were originally a party thereto.
- 3. <u>Notice</u>. Any notice required or permitted by the Agreement shall be given to New Holder at the address or facsimile number listed below New Holder's signature below.

NEW HOLDER:	ACCEPTED AND AGREED [Innovid Corp.]
Print Name:	Print Name:
Ву:	Ву:

Exhibit A to Registration Rights Agreement

Innovid Corp.

Non-Employee Director Compensation Program

Non-employee members of the board of directors (the "Board") of Innovid, Inc. (the "Company") shall receive cash and equity compensation as set forth in this Non-Employee Director Compensation Program (this "Program"). The cash and equity compensation described in this Program shall be paid or be made, as applicable, automatically and without further action of the Board, to each member of the Board who is not an employee of the Company or any subsidiary of the Company (each, a "Non-Employee Director") who is entitled to receive such cash or equity compensation unless such Non-Employee Director declines the receipt of such cash or equity compensation by written notice to the Company. This Program shall remain in effect until it is revised or rescinded by further action of the Board. This Program may be amended, modified or terminated by the Board at any time in its sole discretion. The terms and conditions of this Program shall supersede any prior cash and/or equity compensation arrangements for service as a member of the Board between the Company and any of its Non-Employee Directors. This Program shall become effective on the date of approval by the board of directors of the Company (the "Effective Date").

Cash Compensation

The schedule of annual retainers (the "Annual Retainers") for the Non-Employee Directors is as follows:

<u>Position</u>	Amount
Annual Retainer	\$30,000
Chair of Audit Committee	\$20,000
Chair of Compensation Committee	\$20,000
Chair of Nominating and Governance Committee	\$20,000
Member of Audit Committee (non-Chair)	\$10,000
Member of Compensation Committee (non-Chair)	\$10,000
Member of Nominating and Governance Committee (non-Chair)	\$10,000

For the avoidance of doubt, the Annual Retainers in the table above are additive and a Non-Employee Director shall be eligible to earn an Annual Retainer for each position in which he or she serves. The Annual Retainers shall be earned on a quarterly basis based on a calendar quarter and shall be paid in cash by the Company in arrears not later than the fifteenth (15th) day following the end of each calendar quarter. In the event a Non-Employee Director does not serve as a Non-Employee Director, or in the applicable position, for an entire calendar quarter, the retainer paid to such Non-Employee Director shall be prorated for the portion of such calendar quarter actually served as a Non-Employee Director, or in such position, as applicable.

Equity Compensation

Each Non-Employee Director shall be granted the following awards of Restricted Stock Units (each, an "*RSU Award*") or Other Stock or Cash Based Awards (each, a "*Stock Based Award*") under and subject to the terms and provisions of the Equity Plan under and within the meaning of the Company's 2021 Incentive Award Plan or any other applicable Company equity

incentive plan then-maintained by the Company (the "*Equity Plan*"). Each RSU Award or Stock Based Award shall be granted subject to an award agreement in substantially the form previously approved by the Board, and in the following amounts:

Initial RSU Award	A number of Restricted Stock Units (rounded to the nearest whole number) equal to \$350,000 divided by the Reference Price.
Subsequent RSU Award	A Stock Based Award with a Grant Date Value equal to \$175,000, which, unless otherwise determined by the Board will be granted in the form of a number of Restricted Stock Units equal to \$175,000 divided by the Reference Price.

For purposes of this Program, the "Reference Price" shall mean the closing sales price of one share of the Company's common stock on the date of grant or on the last preceding trading day if the date of grant is not a trading day. "Grant Date Value" shall mean the value as determined in accordance with the Company's consolidated financial statements.

- A . <u>Initial RSU Awards</u>. Each Non-Employee Director who is initially elected or appointed to the Board after the Effective Date shall receive the Initial RSU Award on the date of such initial election or appointment. No Non-Employee Director shall be granted more than one Initial RSU Award.
- B. <u>Subsequent RSU Awards</u>. A Non-Employee Director who (i) has been serving as a Non-Employee Director on the Board for at least six (6) months as of the date of any annual meeting of the Company's stockholders after the Effective Date and (ii) will continue to serve as a Non-Employee Director immediately following such meeting, shall be automatically granted a Subsequent RSU Award, in the discretion of the Board, on the date of such annual meeting. For the avoidance of doubt, a Non-Employee Director elected for the first time to the Board at an annual meeting of the Company's stockholders shall only receive an Initial RSU Award in connection with such election, and shall not receive any Subsequent RSU Award on the date of such meeting as well.
- C. <u>Termination of Employment of Employee Directors</u>. Members of the Board who are employees of the Company or any subsidiary of the Company who subsequently terminate their employment with the Company and subsidiary of the Company and remain on the Board will not receive an Initial RSU Award, but to the extent that they are otherwise entitled, will receive, after termination of employment with the Company and any subsidiary of the Company, a Subsequent RSU Award.
 - D. Terms of RSU Awards Granted to Non-Employee Directors.
 - 1. Vesting.

a. *Initial RSU Awards*. Each Initial RSU Award shall vest and become exercisable in substantially equal installments on each of the first three (3) anniversaries of the date of grant, such that the Initial RSU Award shall be fully vested on the

third (3rd) anniversary of the date of grant, subject to the Non-Employee Director continuing in service as a Non-Employee Director through each such vesting date.

- b. Subsequent RSU Awards. Each Subsequent RSU Award shall vest in a single installment on the earlier of the date of the next annual meeting of the Company's stockholders or the first anniversary of the date of grant, subject to the Non-Employee Director continuing in service as a Non-Employee Director through such vesting date.
- c. Forfeiture of RSU Awards; Change in Control Vesting. Unless the Board otherwise determines, any portion of an Initial RSU Award or Subsequent RSU Award which is unvested at the time of a Non-Employee Director's termination of service on the Board as a Non-Employee Director shall be immediately forfeited upon such termination of service and shall not thereafter become vested. All of a Non-Employee Director's Initial RSU Awards and Subsequent RSU Awards shall vest in full immediately prior to the occurrence of a Change in Control (as defined in the Equity Plan), to the extent outstanding at such time.

* * * * *

INDEMNIFICATION AND ADVANCEMENT AGREEMENT

This Indemnification and Advancement Agreement ("Agree	ment") is made as of	, 2021 by and between Innovid Corp., a
Delaware corporation (the "Company"), and,	[a member of the Board of D	irectors] [an officer] of the Company
("Indemnitee"). This Agreement supersedes and replaces any and	all previous Agreements between	the Company and Indemnitee covering
indemnification and advancement.		

RECITALS

WHEREAS, the Board of Directors of the Company (the "Board") believes that highly competent persons have become more reluctant to serve publicly-held corporations as directors, officers, or in other capacities unless they are provided with adequate protection through insurance or adequate indemnification and advancement of expenses against inordinate risks of claims and actions against them arising out of their service to and activities on behalf of the corporation;

WHEREAS, the Board has determined that, in order to attract and retain qualified individuals, the Company will use commercially reasonable efforts to maintain on an ongoing basis, at its sole expense, liability insurance to protect persons serving the Company and its subsidiaries from certain liabilities. Although the furnishing of such insurance has been a customary and widespread practice among United States-based corporations and other business enterprises, the Company believes that, given current market conditions and trends, such insurance may be available to it in the future only at higher premiums and with more exclusions. At the same time, directors, officers, and other persons in service to corporations or business enterprises are being increasingly subjected to expensive and time-consuming litigation relating to, among other things, matters that traditionally would have been brought only against the Company or business enterprise itself. The Amended and Restated Certificate of Incorporation of the Company permits and the Amended and Restated Bylaws of the Company require indemnification of the officers and directors of the Company. Indemnitee may also be entitled to indemnification pursuant to the General Corporation Law of the State of Delaware (the "DGCL"). The Bylaws, Certificate of Incorporation, and the DGCL expressly provide that the indemnification provisions set forth therein are not exclusive, and thereby contemplate that contracts may be entered into between the Company and members of the board of directors, officers and other persons with respect to indemnification and advancement of expenses;

WHEREAS, the uncertainties relating to such insurance, to indemnification, and to advancement of expenses may increase the difficulty of attracting and retaining such persons;

WHEREAS, the Board has determined that the increased difficulty in attracting and retaining such persons is detrimental to the best interests of the Company and its stockholders and that the Company should act to assure such persons that there will be increased certainty of such protection in the future;

WHEREAS, it is reasonable, prudent and necessary for the Company contractually to obligate itself to indemnify, and to advance expenses on behalf of, such persons to the fullest

extent permitted by applicable law so that they will serve or continue to serve the Company free from undue concern that they will not be so indemnified;

WHEREAS, this Agreement is a supplement to and in furtherance of the Bylaws, Certificate of Incorporation and any resolutions adopted pursuant thereto, and is not a substitute therefor, nor diminishes or abrogates any rights of Indemnitee thereunder; and

WHEREAS, Indemnitee does not regard the protection available under the Bylaws, Certificate of Incorporation, DGCL and insurance as adequate in the present circumstances, and may not be willing to serve or continue to serve as an officer or director without adequate additional protection, and the Company desires Indemnitee to serve or continue to serve in such capacity. Indemnitee is willing to serve, continue to serve and to take on additional service for or on behalf of the Company on the condition that Indemnitee be so indemnified and be advanced expenses.

NOW, THEREFORE, in consideration of the premises and the covenants contained herein, the Company and Indemnitee do hereby covenant and agree as follows:

Section 1. <u>Services to the Company.</u> Indemnitee agrees to serve as [a/an] [director] [officer] of the Company. Indemnitee may at any time and for any reason resign from such position (subject to any other contractual obligation or any obligation imposed by operation of law). This Agreement does not create any obligation on the Company to continue Indemnitee in such position and is not an employment contract between the Company (or any of its subsidiaries or any Enterprise) and Indemnitee.

Section 2. <u>Definitions.</u> As used in this Agreement:

- (a) "Agent" means any person who is authorized by the Company or an Enterprise to act for or represent the interests of the Company or an Enterprise, respectively.
 - (b) A "Change in Control" occurs upon the earliest to occur after the date of this Agreement of any of the following events:
- i. Acquisition of Stock by Third Party. Any Person (as defined below) is or becomes the Beneficial Owner (as defined below), directly or indirectly, of securities of the Company representing fifteen percent (15%) or more of the combined voting power of the Company's then outstanding securities unless the change in relative beneficial ownership of the Company's securities by any Person results solely from a reduction in the aggregate number of outstanding shares of securities entitled to vote generally in the election of directors;
- ii. Change in Board of Directors. During any period of two (2) consecutive years (not including any period prior to the execution of this Agreement), individuals who at the beginning of such period constitute the Board, and any new director (other than a director designated by a person who has entered into an agreement with the Company to effect a transaction described in Sections 2(b)(i), 2(b)(iii) or 2(b)(iv)) whose election by the Board or nomination for election by the Company's stockholders was approved by a vote of at least two-thirds of the directors then still in office who either were directors at the beginning of

the period or whose election or nomination for election was previously so approved, cease for any reason to constitute at least a majority of the members of the Board;

- iii. Corporate Transactions. The effective date of a merger or consolidation of the Company with any other entity, other than a merger or consolidation which would result in the voting securities of the Company outstanding immediately prior to such merger or consolidation continuing to represent (either by remaining outstanding or by being converted into voting securities of the surviving entity) more than 50% of the combined voting power of the voting securities of the surviving entity outstanding immediately after such merger or consolidation and with the power to elect at least a majority of the board of directors or other governing body of such surviving entity;
- iv. Liquidation. The approval by the stockholders of the Company of a complete liquidation of the Company or an agreement for the sale or disposition by the Company of all or substantially all of the Company's assets; and
- v. Other Events. There occurs any other event of a nature that would be required to be reported in response to Item 6(e) of Schedule 14A of Regulation 14A (or a response to any similar item on any similar schedule or form) promulgated under the Exchange Act (as defined below), whether or not the Company is then subject to such reporting requirement.
 - vi. For purposes of this Section 2(b), the following terms have the following meanings:
 - 1 "Exchange Act" means the Securities Exchange Act of 1934, as amended from time to time.
 - 2 "Person" has the meaning as set forth in Sections 13(d) and 14(d) of the Exchange Act; provided, however, that Person excludes (i) the Company, (ii) any trustee or other fiduciary holding securities under an employee benefit plan of the Company, and (iii) any corporation owned, directly or indirectly, by the stockholders of the Company in substantially the same proportions as their ownership of stock of the Company.
 - 3 "Beneficial Owner" has the meaning given to such term in Rule 13d-3 under the Exchange Act; provided, however, that Beneficial Owner excludes any Person otherwise becoming a Beneficial Owner by reason of the stockholders of the Company approving a merger of the Company with another entity.
- (c) "Corporate Status" describes the status of a person who is or was acting as a director, officer, employee, fiduciary, or Agent of the Company or an Enterprise.
- (d) "Disinterested Director" means a director of the Company who is not and was not a party to the Proceeding in respect of which indemnification is sought by Indemnitee.

- (e) "Enterprise" means any other corporation, limited liability company, partnership, joint venture, trust, employee benefit plan or other entity for which Indemnitee is or was serving at the request of the Company as a director, officer, employee, or Agent.
- (f) "Expenses" includes all reasonable attorneys' fees, retainers, court costs, transcript costs, fees of experts and other professionals, witness fees, travel expenses, duplicating costs, printing and binding costs, telephone charges, postage, delivery service fees, any federal, state, local or foreign taxes imposed on Indemnitee as a result of the actual or deemed receipt of any payments under this Agreement, ERISA excise taxes and penalties, and all other disbursements or expenses of the types customarily incurred in connection with prosecuting, defending, preparing to prosecute or defend, investigating, being or preparing to be a witness in, or otherwise participating in, a Proceeding. Expenses also include (i) Expenses incurred in connection with any appeal resulting from any Proceeding, including without limitation the premium, security for, and other costs relating to any cost bond, supersedeas bond, or other appeal bond or its equivalent, and (ii) for purposes of Section 15(d) only, Expenses incurred by Indemnitee in connection with the interpretation, enforcement or defense of Indemnitee's rights under this Agreement, by litigation or otherwise. Expenses, however, do not include amounts paid in settlement by Indemnitee or the amount of judgments or fines against Indemnitee.
- (g) "Independent Counsel" means a law firm, or a member of a law firm, that is experienced in matters of corporation law and neither presently is, nor in the past five years has been, retained to represent: (i) the Company or Indemnitee in any matter material to either such party (other than with respect to matters concerning the Indemnitee under this Agreement, or of other indemnitees under similar indemnification agreements), or (ii) any other party to the Proceeding giving rise to a claim for indemnification hereunder. Notwithstanding the foregoing, the term "Independent Counsel" does not include any person who, under the applicable standards of professional conduct then prevailing, would have a conflict of interest in representing either the Company or Indemnitee in an action to determine Indemnitee's rights under this Agreement.
- (h) The term "Proceeding" includes any threatened, pending or completed action, suit, claim, counterclaim, cross claim, arbitration, mediation, alternate dispute resolution mechanism, investigation, inquiry, administrative hearing or any other actual, threatened or completed proceeding, whether brought in the right of the Company or otherwise and whether of a civil, criminal, administrative, legislative, or investigative (formal or informal) nature, including any appeal therefrom, in which Indemnitee was, is or will be involved as a party, potential party, non-party witness or otherwise by reason of Indemnitee's Corporate Status or by reason of any action taken by Indemnitee (or a failure to take action by Indemnitee) or of any action (or failure to act) on Indemnitee's part while acting pursuant to Indemnitee's Corporate Status, in each case whether or not serving in such capacity at the time any liability or Expense is incurred for which indemnification, reimbursement, or advancement of Expenses can be provided under this Agreement. A Proceeding also includes a situation the Indemnitee believes in good faith may lead to or culminate in the institution of a Proceeding.
- Section 3. <u>Indemnity in Third-Party Proceedings.</u> The Company will indemnify Indemnitee in accordance with the provisions of this Section 3 if Indemnitee is, or is threatened to be made, a party to or a participant in any Proceeding, other than a Proceeding by or in the

right of the Company to procure a judgment in its favor. Pursuant to this Section 3, the Company will indemnify Indemnitee to the fullest extent permitted by applicable law against all Expenses, judgments, fines and amounts paid in settlement (including all interest, assessments and other charges paid or payable in connection with or in respect of such Expenses, judgments, fines and amounts paid in settlement) actually and reasonably incurred by Indemnitee or on Indemnitee's behalf in connection with such Proceeding or any claim, issue or matter therein, if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in or not opposed to the best interests of the Company and, in the case of a criminal Proceeding had no reasonable cause to believe that Indemnitee's conduct was unlawful.

Section 4. <u>Indemnity in Proceedings by or in the Right of the Company.</u> The Company will indemnify Indemnitee in accordance with the provisions of this Section 4 if Indemnitee is, or is threatened to be made, a party to or a participant in any Proceeding by or in the right of the Company to procure a judgment in its favor. Pursuant to this Section 4, the Company will indemnify Indemnitee to the fullest extent permitted by applicable law against all Expenses actually and reasonably incurred by Indemnitee or on Indemnitee's behalf in connection with such Proceeding or any claim, issue or matter therein, if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in or not opposed to the best interests of the Company. The Company will not indemnify Indemnitee for Expenses under this Section 4 related to any claim, issue or matter in a Proceeding for which Indemnitee has been finally adjudged by a court to be liable to the Company, unless, and only to the extent that, the Delaware Court of Chancery or any court in which the Proceeding was brought determines upon application by Indemnitee that, despite the adjudication of liability but in view of all the circumstances of the case, Indemnitee is fairly and reasonably entitled to indemnification.

Section 5. <u>Indemnification for Expenses of a Party Who is Wholly or Partly Successful.</u> To the fullest extent permitted by applicable law, the Company will indemnify Indemnitee against all Expenses actually and reasonably incurred by Indemnitee in connection with any Proceeding to the extent that Indemnitee is successful, on the merits or otherwise. If Indemnitee is not wholly successful in such Proceeding but is successful, on the merits or otherwise, as to one or more but less than all claims, issues or matters in such Proceeding, the Company will indemnify Indemnitee against all Expenses actually and reasonably incurred by Indemnitee or on Indemnitee's behalf in connection with or related to each successfully resolved claim, issue or matter to the fullest extent permitted by law. For purposes of this Section 5 and without limitation, the termination of any claim, issue or matter in such a Proceeding by dismissal, with or without prejudice, will be deemed to be a successful result as to such claim, issue or matter.

Section 6. <u>Indemnification For Expenses of a Witness.</u> To the fullest extent permitted by applicable law, the Company will indemnify Indemnitee against all Expenses actually and reasonably incurred by Indemnitee or on Indemnitee's behalf in connection with any Proceeding to which Indemnitee is not a party but to which Indemnitee is a witness, deponent, interviewee, or otherwise asked to participate.

Section 7. <u>Partial Indemnification.</u> If Indemnitee is entitled under any provision of this Agreement to indemnification by the Company for some or a portion of Expenses, but not,

however, for the total amount thereof, the Company will indemnify Indemnitee for the portion thereof to which Indemnitee is entitled.

- Section 8. <u>Additional Indemnification</u>. Notwithstanding any limitation in Sections 3, 4, or 5, the Company will indemnify Indemnitee to the fullest extent permitted by applicable law (including but not limited to, the DGCL and any amendments to or replacements of the DGCL adopted after the date of this Agreement that expand the Company's ability to indemnify its officers and directors) if Indemnitee is a party to or threatened to be made a party to any Proceeding (including a Proceeding by or in the right of the Company to procure a judgment in its favor).
- Section 9. <u>Exclusions.</u> Notwithstanding any provision in this Agreement, the Company is not obligated under this Agreement to make any indemnification payment to Indemnitee in connection with any Proceeding:
- (a) for which payment has actually been made to or on behalf of Indemnitee under any insurance policy or other indemnity provision, except to the extent provided in Section 17(b) and except with respect to any excess beyond the amount paid under any insurance policy or other indemnity provision; or
- (b) for (i) an accounting of profits made from the purchase and sale (or sale and purchase) by Indemnitee of securities of the Company within the meaning of Section 16(b) of the Exchange Act (as defined in Section 2(b) hereof) or similar provisions of state statutory law or common law, (ii) any reimbursement of the Company by the Indemnitee of any bonus or other incentive-based or equity-based compensation or of any profits realized by the Indemnitee from the sale of securities of the Company, as required in each case under the Exchange Act (including any such reimbursements that arise from an accounting restatement of the Company pursuant to Section 304 of the Sarbanes-Oxley Act of 2002 (the "Sarbanes-Oxley Act"), or the payment to the Company of profits arising from the purchase and sale by Indemnitee of securities in violation of Section 306 of the Sarbanes-Oxley Act) or (iii) any reimbursement of the Company by Indemnitee of any compensation pursuant to any compensation recoupment or clawback policy adopted by the Board or the compensation committee of the Board, including but not limited to any such policy adopted to comply with stock exchange listing requirements implementing Section 10D of the Exchange Act; or
- (c) initiated by Indemnitee, including any Proceeding (or any part of any Proceeding) initiated by Indemnitee against the Company or its directors, officers, employees or other indemnitees, unless (i) the Proceeding or part of any Proceeding is to enforce Indemnitee's rights to indemnification or advancement, of Expenses, including a Proceeding (or any part of any Proceeding) initiated pursuant to Section 15 of this Agreement, (ii) the Board authorized the Proceeding (or any part of any Proceeding) prior to its initiation or (iii) the Company provides the indemnification, in its sole discretion, pursuant to the powers vested in the Company under applicable law.

Section 10. Advances of Expenses.

- (a) The Company will advance, to the extent not prohibited by law, the Expenses incurred by Indemnitee in connection with (x) any Proceeding (or any part of any Proceeding) not initiated by Indemnitee or (y) any Proceeding (or any part of any Proceeding) initiated by Indemnitee if, in the case of sub-clause (y), (i) the Proceeding or part of any Proceeding is to enforce Indemnitee's rights to obtain indemnification or advancement of Expenses from the Company or Enterprise, including a proceeding initiated pursuant to Section 15, or (ii) the Board authorized the Proceeding (or any part of any Proceeding) prior to its initiation. The Company will advance the Expenses within thirty (30) days after the receipt by the Company of a statement or statements requesting such advances from time to time, whether prior to or after final disposition of any Proceeding.
- (b) Advances will be unsecured and interest free. Indemnitee undertakes to repay the amounts advanced (without interest) to the extent that it is ultimately determined that Indemnitee is not entitled to be indemnified by the Company, thus Indemnitee qualifies for advances upon the execution of this Agreement and delivery to the Company. No other form of undertaking is required other than the execution of this Agreement. The Company will make advances without regard to Indemnitee's ability to repay the Expenses and without regard to Indemnitee's ultimate entitlement to indemnification under the other provisions of this Agreement.

Section 11. Procedure for Notification of Claim for Indemnification or Advancement.

- (a) Indemnitee will notify the Company in writing of any Proceeding with respect to which Indemnitee intends to seek indemnification or advancement of Expenses hereunder as soon as reasonably practicable following the receipt by Indemnitee of written notice thereof. Indemnitee will include in the written notification to the Company a description of the nature of the Proceeding and the facts underlying the Proceeding and provide such documentation and information as is reasonably available to Indemnitee and is reasonably necessary to determine whether and to what extent Indemnitee is entitled to indemnification following the final disposition of such Proceeding. Indemnitee's failure to notify the Company will not relieve the Company from any obligation it may have to Indemnitee under this Agreement, and any delay in so notifying the Company will not constitute a waiver by Indemnitee of any rights under this Agreement. The Secretary of the Company will, promptly upon receipt of such a request for indemnification or advancement, advise the Board in writing that Indemnitee has requested indemnification or advancement.
 - (b) The Company will be entitled to participate in the Proceeding at its own expense.
 - Section 12. Procedure Upon Application for Indemnification.
 - (a) Unless a Change of Control has occurred, the determination of Indemnitee's entitlement to indemnification will be made:
 - i. by a majority vote of the Disinterested Directors, even though less than a quorum of the Board;

- ii. by a committee of Disinterested Directors designated by a majority vote of the Disinterested Directors, even though less than a quorum of the Board;
- iii. if there are no such Disinterested Directors or, if such Disinterested Directors so direct, by written opinion provided by Independent Counsel selected by the Board; or
 - iv. if so directed by the Board, by the stockholders of the Company.
- (b) If a Change in Control has occurred, the determination of Indemnitee's entitlement to indemnification will be made by written opinion provided by Independent Counsel selected by Indemnitee (unless Indemnitee requests such selection be made by the Board).
- (c) The party selecting Independent Counsel pursuant to subsection (a)(iii) or (b) of this Section 13 will provide written notice of the selection to the other party. The notified party may, within ten (10) days after receiving written notice of the selection of Independent Counsel, deliver to the selecting party a written objection to such selection; provided, however, that such objection may be asserted only on the ground that the Independent Counsel so selected does not meet the requirements of "Independent Counsel" as defined in Section 2 of this Agreement, and the objection will set forth with particularity the factual basis of such assertion. Absent a proper and timely objection, the person so selected will act as Independent Counsel. If such written objection is so made and substantiated, the Independent Counsel so selected may not serve as Independent Counsel unless and until such objection is withdrawn or the Delaware Court has determined that such objection is without merit. If, within thirty (30) days after the later of submission by Indemnitee of a written request for indemnification pursuant to Section 12(a) hereof and the final disposition of the Proceeding, Independent Counsel has not been selected or, if selected, any objection to has not been resolved, either the Company or Indemnitee may petition the Delaware Court for the appointment as Independent Counsel of a person selected by such court or by such other person as such court designates. Upon the due commencement of any judicial proceeding or arbitration pursuant to Section 15(a) of this Agreement, Independent Counsel will be discharged and relieved of any further responsibility in such capacity (subject to the applicable standards of professional conduct then prevailing).
- (d) Indemnitee will cooperate with the person, persons or entity making the determination with respect to Indemnitee's entitlement to indemnification, including providing to such person, persons or entity upon reasonable advance request any documentation or information which is not privileged or otherwise protected from disclosure and which is reasonably available to Indemnitee and reasonably necessary to such determination. The Company will advance and pay any Expenses incurred by Indemnitee in so cooperating with the person, persons or entity making the indemnification determination irrespective of the determination as to Indemnitee's entitlement to indemnification and the Company hereby indemnifies and agrees to hold Indemnitee harmless therefrom. The Company promptly will advise Indemnitee in writing of the determination that Indemnitee is or is not entitled to indemnification, including a description of any reason or basis for which indemnification has been denied and providing a copy of any written opinion provided to the Board by Independent Counsel.

- (e) If it is determined that Indemnitee is entitled to indemnification, the Company will make payment to Indemnitee within thirty (30) days after such determination.
- (f) The Company agrees to pay the reasonable fees and expenses of Independent Counsel and to fully indemnify and hold harmless such Independent Counsel against any and all Expenses, claims, liabilities and damages arising out of or relating to this agreement or its engagement pursuant hereto.

Section 13. <u>Presumptions and Effect of Certain Proceedings.</u>

- (a) In making a determination with respect to entitlement to indemnification hereunder, the person or persons or entity making such determination will, to the fullest extent not prohibited by law, presume Indemnitee is entitled to indemnification under this Agreement if Indemnitee has submitted a request for indemnification in accordance with Section 12(a) of this Agreement, and the Company will, to the fullest extent not prohibited by law, have the burden of proof to overcome that presumption. Neither the failure of the Company (including by its directors or Independent Counsel) to have made a determination prior to the commencement of any action pursuant to this Agreement that indemnification is proper in the circumstances because Indemnitee has met the applicable standard of conduct, nor an actual determination by the Company (including by its directors or Independent Counsel) that Indemnitee has not met such applicable standard of conduct, will be a defense to the action or create a presumption that Indemnitee has not met the applicable standard of conduct.
- (b) If the determination of the Indemnitee's entitlement to indemnification has not been made pursuant to Section 13 within sixty (60) days after the later of (i) receipt by the Company of Indemnitee's request for indemnification pursuant to Section 12(a) and (ii) the final disposition of the Proceeding for which Indemnitee requested Indemnification (the "Determination Period"), the requisite determination of entitlement to indemnification will, to the fullest extent not prohibited by law, be deemed to have been made and Indemnitee will be entitled to such indemnification, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statement not materially misleading, in connection with the request for indemnification, or (ii) a prohibition of such indemnification under applicable law. The Determination Period may be extended for a reasonable time, not to exceed an additional thirty (30) days, if the person, persons or entity making the determination with respect to entitlement to indemnification in good faith requires such additional time for the obtaining or evaluating of documentation and/or information relating thereto; and provided, further, the Determination Period may be extended an additional fifteen (15) days if the determination of entitlement to indemnification is to be made by the stockholders pursuant to Section 13(a)(iv) of this Agreement.
- (c) The termination of any Proceeding or of any claim, issue or matter therein, by judgment, order, settlement or conviction, or upon a plea of <u>nolo contendere</u> or its equivalent, will not (except as otherwise expressly provided in this Agreement) of itself adversely affect the right of Indemnitee to indemnification or create a presumption that Indemnitee did not act in good faith and in a manner which Indemnitee reasonably believed to be in or not opposed to the best interests of the Company or, with respect to any criminal Proceeding, that Indemnitee had reasonable cause to believe that Indemnitee's conduct was unlawful.

- (d) For purposes of any determination of good faith, Indemnitee will be deemed to have acted in good faith if Indemnitee acted based on the records or books of account of the Company, its subsidiaries, or an Enterprise, including financial statements, or on information supplied to Indemnitee by the directors or officers of the Company, its subsidiaries, or an Enterprise in the course of their duties, or on the advice of legal counsel for the Company, its subsidiaries, or an Enterprise or on information or records given or reports made to the Company or an Enterprise by an independent certified public accountant or by an appraiser, financial advisor or other expert selected with reasonable care by or on behalf of the Company, its subsidiaries, or an Enterprise. Further, Indemnitee will be deemed to have acted in a manner "not opposed to the best interests of the Company," as referred to in this Agreement if Indemnitee acted in good faith and in a manner Indemnitee reasonably believed to be in the best interests of the participants and beneficiaries of an employee benefit plan. The provisions of this Section 14(d) is not exclusive and does not limit in any way the other circumstances in which the Indemnitee may be deemed to have met the applicable standard of conduct set forth in this Agreement.
- (e) The knowledge and/or actions, or failure to act, of any director, officer, trustee, partner, managing member, fiduciary, agent or employee of the Enterprise may not be imputed to Indemnitee for purposes of determining Indemnitee's right to indemnification under this Agreement.

Section 14. Remedies of Indemnitee.

(a) Indemnitee may commence litigation against the Company in the Delaware Court of Chancery to obtain indemnification or advancement of Expenses provided by this Agreement in the event that (i) a determination is made pursuant to Section 13 of this Agreement that Indemnitee is not entitled to indemnification under this Agreement, (ii) the Company does not advance Expenses pursuant to Section 11 of this Agreement, (iii) the determination of entitlement to indemnification is not made pursuant to Section 13 of this Agreement within the Determination Period, (iv) the Company does not indemnify Indemnitee pursuant to Section 5 or 6 or the second to last sentence of Section 13(d) of this Agreement within thirty (30) days after receipt by the Company of a written request therefor, (v) the Company does not indemnify Indemnitee pursuant to Section 3, 4, 7, or 8 of this Agreement within thirty (30) days after a determination has been made that Indemnitee is entitled to indemnification, or (vi) in the event that the Company or any other person takes or threatens to take any action to declare this Agreement void or unenforceable, or institutes any litigation or other action or Proceeding designed to deny, or to recover from, the Indemnitee the benefits provided or intended to be provided to the Indemnitee hereunder. Alternatively, Indemnitee or the Company, may seek an award in arbitration to be conducted by a single arbitrator pursuant to the Commercial Arbitration Rules of the American Arbitration Association. Indemnitee must commence such Proceeding seeking an adjudication or an award in arbitration within one hundred and eighty (180) days following the date on which Indemnitee first has the right to commence such Proceeding pursuant to this Section 15(a); provided, however, that the foregoing clause does not apply in respect of a Proceeding brought by Indemnitee to enforce Indemnitee's rights under Section 5 of this Agreement. The Company will not oppose Indemnitee's right to seek any such adjudication or award in arb

- (b) If a determination is made pursuant to Section 13 of this Agreement that Indemnitee is not entitled to indemnification, any judicial proceeding or arbitration commenced pursuant to this Section 15 will be conducted in all respects as a *de novo* trial, or arbitration, on the merits and Indemnitee may not be prejudiced by reason of that adverse determination. In any judicial proceeding or arbitration commenced pursuant to this Section 15, the Company will have the burden of proving Indemnitee is not entitled to indemnification or advancement of Expenses, as the case may be, and will not introduce evidence of the determination made pursuant to Section 13 of this Agreement.
- (c) If a determination is made pursuant to Section 13 of this Agreement that Indemnitee is entitled to indemnification, the Company will be bound by such determination in any judicial proceeding or arbitration commenced pursuant to this Section 15, absent (i) a misstatement by Indemnitee of a material fact, or an omission of a material fact necessary to make Indemnitee's statement not materially misleading, in connection with the request for indemnification, or (ii) a prohibition of such indemnification under applicable law.
- (d) The Company is, to the fullest extent not prohibited by law, precluded from asserting in any judicial proceeding or arbitration commenced pursuant to this Section 15 that the procedures and presumptions of this Agreement are not valid, binding and enforceable and will stipulate in any such court or before any such arbitrator that the Company is bound by all the provisions of this Agreement.
- (e) It is the intent of the Company that, to the fullest extent permitted by law, the Indemnitee not be required to incur legal fees or other Expenses associated with the interpretation, enforcement or defense of Indemnitee's rights under this Agreement by litigation or otherwise because the cost and expense thereof would substantially detract from the benefits intended to be extended to the Indemnitee hereunder. The Company, to the fullest extent permitted by law, will (within thirty (30) days after receipt by the Company of a written request therefor) advance to Indemnitee such Expenses which are incurred by Indemnitee in connection with any action concerning this Agreement, Indemnitee's right to indemnification or advancement of Expenses from the Company, or concerning any directors' and officers' liability insurance policies maintained by the Company, and will indemnify Indemnitee against any and all such Expenses unless the court determines that each of the Indemnitee's claims in such action were made in bad faith or were frivolous or are prohibited by law.
- Section 15. <u>Security.</u> Notwithstanding anything herein to the contrary to the extent requested by Indemnitee and approved by the Board, the Company may at any time and from time to time provide security to Indemnitee for the Company's obligations hereunder through an irrevocable bank line of credit, funded trust or other collateral.

Section 16. Non-exclusivity; Survival of Rights; Insurance; Subrogation.

(a) The indemnification and advancement of Expenses provided by this Agreement are not exclusive of any other rights to which Indemnitee may at any time be entitled under applicable law, the Certificate of Incorporation, the Bylaws, any agreement, a vote of stockholders or a resolution of directors, or otherwise. The indemnification and advancement of Expenses provided by this Agreement may not be limited or restricted by any amendment,

alteration or repeal of this Agreement in any way with respect to any action taken or omitted by Indemnitee in Indemnitee's Corporate Status occurring prior to any amendment, alteration or repeal of this Agreement. To the extent that a change in Delaware law, whether by statute or judicial decision, permits greater indemnification or advancement of Expenses than would be afforded currently under the Bylaws, Certificate of Incorporation, or this Agreement, it is the intent of the parties hereto that Indemnitee enjoy by this Agreement the greater benefits so afforded by such change. No right or remedy herein conferred is intended to be exclusive of any other right or remedy, and every other right and remedy given hereunder or now or hereafter existing at law or in equity or otherwise. The assertion or employment of any right or remedy hereunder, or otherwise, will not prevent the concurrent assertion or employment of any other right or remedy.

- (b) The Company hereby acknowledges that Indemnitee may have certain rights to indemnification, advancement of Expenses and/or insurance provided by one or more other Persons with whom or which Indemnitee may be associated. The relationship between the Company and such other Persons, other than an Enterprise, with respect to the Indemnitee's rights to indemnification, advancement of Expenses, and insurance is described by this subsection, subject to the provisions of subsection (d) of this Section 17 with respect to a Proceeding concerning Indemnitee's Corporate Status with an Enterprise.
 - i. The Company hereby acknowledges and agrees:
- 1) the Company is the indemnitor of first resort with respect to any request for indemnification or advancement of Expenses made pursuant to this Agreement concerning any Proceeding;
- 2) the Company is primarily liable for all indemnification and indemnification or advancement of Expenses obligations for any Proceeding, whether created by law, organizational or constituent documents, contract (including this Agreement) or otherwise:
- 3) any obligation of any other Persons with whom or which Indemnitee may be associated to indemnify Indemnitee and/or advance Expenses to Indemnitee in respect of any proceeding are secondary to the obligations of the Company's obligations;
- 4) the Company will indemnify Indemnitee and advance Expenses to Indemnitee hereunder to the fullest extent provided herein without regard to any rights Indemnitee may have against any other Person with whom or which Indemnitee may be associated or insurer of any such Person; and
- ii. the Company irrevocably waives, relinquishes and releases (A) any other Person with whom or which Indemnitee may be associated from any claim of contribution, subrogation, reimbursement, exoneration or indemnification, or any other recovery of any kind in respect of amounts paid by the Company to Indemnitee pursuant to this Agreement and (B) any right to participate in any claim or remedy of Indemnitee against any Person, whether or not such claim, remedy or right arises in equity or under contract, statute or common law, including, without limitation, the right to take or receive from any Person, directly or indirectly, in cash or

other property or by set-off or in any other manner, payment or security on account of such claim, remedy or right.

- iii. In the event any other Person with whom or which Indemnitee may be associated or their insurers advances or extinguishes any liability or loss for Indemnitee, the payor has a right of subrogation against the Company or its insurers for all amounts so paid which would otherwise be payable by the Company or its insurers under this Agreement. In no event will payment by any other Person with whom or which Indemnitee may be associated or their insurers affect the obligations of the Company hereunder or shift primary liability for the Company's obligation to indemnify or advance of Expenses to any other Person with whom or which Indemnitee may be associated.
- iv. Any indemnification or advancement of Expenses provided by any other Person with whom or which Indemnitee may be associated is specifically in excess over the Company's obligation to indemnify and advance Expenses or any valid and collectible insurance (including but not limited to any malpractice insurance or professional errors and omissions insurance) provided by the Company.
- (c) To the extent that the Company maintains an insurance policy or policies providing liability insurance for directors, officers, employees, or agents of the Company, the Company will obtain a policy or policies covering Indemnitee to the maximum extent of the coverage available for any such director, officer, employee or agent under such policy or policies, including coverage in the event the Company does not or cannot, for any reason, indemnify or advance Expenses to Indemnitee as required by this Agreement. If, at the time of the receipt of a notice of a claim pursuant to this Agreement, the Company has director and officer liability insurance in effect, the Company will give prompt notice of such claim or of the commencement of a Proceeding, as the case may be, to the insurers in accordance with the procedures set forth in the respective policies. The Company will thereafter take all necessary or desirable action to cause such insurers to pay, on behalf of the Indemnitee, all amounts payable as a result of such Proceeding in accordance with the terms of such policies. Indemnitee agrees to assist the Company efforts to cause the insurers to pay such amounts and will comply with the terms of such policies, including selection of approved panel counsel, if required.
- (d) The Company's obligation to indemnify or advance Expenses hereunder to Indemnitee for any Proceeding concerning Indemnitee's Corporate Status with an Enterprise will be reduced by any amount Indemnitee has actually received as indemnification or advancement of Expenses from such Enterprise. The Company and Indemnitee intend that any such Enterprise (and its insurers) be the indemnitor of first resort with respect to indemnification and advancement of Expenses for any Proceeding related to or arising from Indemnitee's Corporate Status with such Enterprise. The Company's obligation to indemnify and advance Expenses to Indemnitee is secondary to the obligations the Enterprise or its insurers owe to Indemnitee. Indemnitee agrees to take all reasonably necessary and desirable action to obtain from an Enterprise indemnification and advancement of Expenses for any Proceeding related to or arising from Indemnitee's Corporate Status with such Enterprise.
- (e) In the event of any payment made by the Company under this Agreement, the Company will be subrogated to the extent of such payment to all of the rights of recovery of

Indemnitee from any Enterprise or insurance carrier. Indemnitee will execute all papers required and take all action necessary to secure such rights, including execution of such documents as are necessary to enable the Company to bring suit to enforce such rights.

Section 17. <u>Duration of Agreement</u>. This Agreement continues until and terminates upon the later of: (a) ten (10) years after the date that Indemnitee ceases to have a Corporate Status or (b) one (1) year after the final termination of any Proceeding then pending in respect of which Indemnitee is granted rights of indemnification or advancement of Expenses hereunder and of any Proceeding commenced by Indemnitee pursuant to Section 15 of this Agreement relating thereto. The indemnification and advancement of Expenses rights provided by or granted pursuant to this Agreement are binding upon and be enforceable by the parties hereto and their respective successors and assigns (including any direct or indirect successor by purchase, merger, consolidation or otherwise to all or substantially all of the business or assets of the Company), continue as to an Indemnitee who has ceased to be a director, officer, employee or agent of the Company or of any other Enterprise, and inure to the benefit of Indemnitee and Indemnitee's spouse, assigns, heirs, devisees, executors and administrators and other legal representatives.

Section 18. Severability. If any provision or provisions of this Agreement is held to be invalid, illegal or unenforceable for any reason whatsoever: (a) the validity, legality and enforceability of the remaining provisions of this Agreement (including without limitation, each portion of any Section of this Agreement containing any such provision held to be invalid, illegal or unenforceable, that is not itself invalid, illegal or unenforceable) will not in any way be affected or impaired thereby and remain enforceable to the fullest extent permitted by law; (b) such provision or provisions will be deemed reformed to the extent necessary to conform to applicable law and to give the maximum effect to the intent of the parties hereto; and (c) to the fullest extent possible, the provisions of this Agreement (including, without limitation, each portion of any Section of this Agreement containing any such provision held to be invalid, illegal or unenforceable, that is not itself invalid, illegal or unenforceable) will be construed so as to give effect to the intent manifested thereby.

Section 19. <u>Interpretation</u>. Any ambiguity in the terms of this Agreement will be resolved in favor of Indemnitee and in a manner to provide the maximum indemnification and advancement of Expenses permitted by law. The Company and Indemnitee intend that this Agreement provide to the fullest extent permitted by law for indemnification and advancement in excess of that expressly provided, without limitation, by the Certificate of Incorporation, the Bylaws, vote of the Company stockholders or disinterested directors, or applicable law.

Section 20. Enforcement.

(a) The Company expressly confirms and agrees that it has entered into this Agreement and assumed the obligations imposed on it hereby in order to induce Indemnitee to serve as a director or officer of the Company, and the Company acknowledges that Indemnitee is relying upon this Agreement in serving or continuing to serve as a director or officer of the Company.

- (b) This Agreement constitutes the entire agreement between the parties hereto with respect to the subject matter hereof and supersedes all prior agreements and understandings, oral, written and implied, between the parties hereto with respect to the subject matter hereof; provided, however, that this Agreement is a supplement to and in furtherance of the Certificate of Incorporation, the Bylaws and applicable law, and is not a substitute therefor, nor to diminish or abrogate any rights of Indemnitee thereunder.
- (c) The Company shall require and cause any successor (whether direct or indirect by purchase, merger, consolidation or otherwise) to all, substantially all or a substantial part, of the business and/or assets of the Company, by written agreement in form and substance satisfactory to Indemnitee, expressly to assume and agree to perform this Agreement in the same manner and to the same extent that the Company would be required to perform if no such succession had taken place.
- Section 21. <u>Modification and Waiver.</u> No supplement, modification or amendment of this Agreement is binding unless executed in writing by the parties hereto. No waiver of any of the provisions of this Agreement will be deemed or constitutes a waiver of any other provisions of this Agreement nor will any waiver constitute a continuing waiver.
- Section 22. <u>Notice by Indemnitee.</u> Indemnitee agrees promptly to notify the Company in writing upon being served with any summons, citation, subpoena, complaint, indictment, information or other document relating to any Proceeding or matter which may be subject to indemnification or advancement of Expenses covered hereunder. The failure of Indemnitee to so notify the Company does not relieve the Company of any obligation which it may have to the Indemnitee under this Agreement or otherwise.
- Section 23. <u>Notices.</u> All notices, requests, demands and other communications under this Agreement will be in writing and will be deemed to have been duly given if (a) delivered by hand to the other party, (b) sent by reputable overnight courier to the other party or (c) sent by facsimile transmission or electronic mail, with receipt of oral confirmation that such communication has been received:
- (a) If to Indemnitee, at the address indicated on the signature page of this Agreement, or such other address as Indemnitee provides to the Company.
 - (b) If to the Company to:

Innovid Corp.
30 Irving Place – 12th Floor
New York, NY 10003
Attn: Legal Department
Email: corporate@innovid.com

With a copy to:

Latham & Watkins LLP 1271 Avenue of Americas New York, NY 10020 Attn: Andra Troy

Facsimile: (212) 751-4864

or to any other address as may have been furnished to Indemnitee by the Company.

Section 24. <u>Contribution</u>. To the fullest extent permissible under applicable law, if the indemnification provided for in this Agreement is unavailable to Indemnitee for any reason whatsoever, the Company, in lieu of indemnifying Indemnitee, will contribute to the amount incurred by Indemnitee, whether for judgments, fines, penalties, excise taxes, amounts paid or to be paid in settlement and/or for Expenses, in connection with any claim relating to an indemnifiable event under this Agreement, in such proportion as is deemed fair and reasonable in light of all of the circumstances of such Proceeding in order to reflect (i) the relative benefits received by the Company and Indemnitee as a result of the event(s) and/or transaction(s) giving cause to such Proceeding; and/or (ii) the relative fault of the Company (and its directors, officers, employees and agents) and Indemnitee in connection with such event(s) and/or transaction(s).

Section 25. Applicable Law and Consent to Jurisdiction. This Agreement and the legal relations among the parties are governed by, and construed and enforced in accordance with, the laws of the State of Delaware, without regard to its conflict of laws rules. Except with respect to any arbitration commenced by Indemnitee pursuant to Section 15(a) of this Agreement, the Company and Indemnitee hereby irrevocably and unconditionally (i) agree that any action or Proceeding arising out of or in connection with this Agreement may be brought only in the Delaware Court of Chancery and not in any other state or federal court in the United States of America or any court in any other country, (ii) consent to submit to the exclusive jurisdiction of the Delaware Court for purposes of any action or Proceeding arising out of or in connection with this Agreement, (iii) waive any objection to the laying of venue of any such action or Proceeding in the Delaware Court, and (iv) waive, and agree not to plead or to make, any claim that any such action or Proceeding brought in the Delaware Court has been brought in an improper or inconvenient forum.

Section 26. <u>Identical Counterparts.</u> This Agreement may be executed in one or more counterparts, each of which will for all purposes be deemed to be an original but all of which together constitutes one and the same Agreement. Only one such counterpart signed by the party against whom enforceability is sought needs to be produced to evidence the existence of this Agreement.

Section 27. <u>Headings</u>. The headings of this Agreement are inserted for convenience only and do not constitute part of this Agreement or affect the construction thereof.

IN WITNESS WHEREOF, the part	ies have caused this Agreement to be signed as of the day and year first above written.
COMPANY	INDEMNITEE
Ву:	
Name:	Name:
Office:	Address:

Subsidiaries of Innovid Corp.

Name of Subsidiary Jurisdiction of Organization

Innovid LLC Delaware

INNOVID, INC. AND ITS SUBSIDIARIES CONDENSED CONSOLIDATED BALANCE SHEETS (In thousands, except stock and per stock data)

	Sej	ptember 30, 2021	Decei	nber 31, 2020
	J)	U naudited)		
ASSETS				
CURRENT ASSETS:				
Cash and cash equivalents	\$	14,472	\$	15,645
Trade receivables, net (allowance for doubtful accounts of \$83 and \$121 at September 30, 2021 and December 31, 2020, respectively)		34,223		34,804
Prepaid expenses and other current assets		1,966		1,174
Total current assets		50,661		51,623
NON-CURRENT ASSETS:				
Long-term other deposit		317		348
Long-term restricted deposits		445		447
Property and equipment, net		3,298		2,325
Goodwill		4,555		4,555
Intangible assets, net		_		33
Deferred offering cost		3,269		_
Other non-current assets		607		127
Total non-current assets		12,491		7,835
TOTAL ASSETS	\$	63,152	\$	59,458
LIABILITIES, TEMPORARY EQUITY, AND STOCKHOLDERS' DEFICIT				
CURRENT LIABILITIES:				
Trade payables	\$	2,564	\$	1,854
Employees and payroll accruals		6,861		6,506
Accrued expenses and other current liabilities		2,171		1,155
Current portion of long-term debt		_		1,527
Deferred offering cost accrual		2,406		_
Total current liabilities		14,002		11,042
NON-CURRENT LIABILITIES:				
Long-term debt		6,000		7,506
Other non-current liabilities		2,854		3,144
Warrants liability		3,690		499
Total non-current liabilities		12,544		11,149
TOTAL LIABILITIES		26,546		22,191
COMMITMENTS AND CONTINGENT LIABILITIES (Note 6)				
TEMPORARY EQUITY				
Preferred stocks - Authorized: 55,514,480 at September 30, 2021 (unaudited) and December 31, 2020; Issued and Outstanding: 55,105,773 at September 30, 2021 (unaudited) and December 31, 2020		139,990		86,997
STOCKHOLDERS' DEFICIT:				
Common stocks of \$0.001 par value - Authorized: 75,254,333 at September 30, 2021 (unaudited) and December 31, 2020; Issued: 15,704,059 and 13,602,467 stocks at September 30, 2021 (unaudited) and December 31, 2020, respectively, and Outstanding 14,272,551 and 12,170,320 stocks at September 30, 2021 (unaudited) and December 31, 2020, respectively, and		14		12
Outstanding: 14,272,521 and 12,170,929 stocks at September 30, 2021 (unaudited) and December 31, 2020, respectively Treasury stocks, at cost (1,431,538 stocks at September 30, 2021 (unaudited) and December 31, 2020)		(1,629)		(1,629)
Accumulated deficit		(1,629)		(48,113)
Total stockholders' deficit		(101,769)		(49,730)
	\$	63,152	\$	59,458
TOTAL LIABILITIES, TEMPORARY EQUITY AND STOCKHOLDERS' DEFICIT	3	05,152	Ф	39,438

The accompanying notes are an integral part of the interim condensed consolidated financial statements.

INNOVID, INC. AND ITS SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF OPERATIONS

(In thousands, except stock and per stock data)

		Nine months end	ed Sej	otember 30,
	2021			2020
		(Unaudited)		(Unaudited)
Revenues	\$	64,324	\$	45,772
Cost of revenues		12,418		8,544
Gross profit		51,906		37,228
Operating expenses:				
Research and development		16,932		13,673
Sales and marketing		23,534		22,624
General and administrative		10,587		5,622
Total operating expenses	·	51,053		41,919
Operating profit/ (loss)		853		(4,691)
Finance expenses, net		3,878		528
Loss before taxes		(3,025)		(5,219)
Taxes on income		829		899
Net loss	\$	(3,854)	\$	(6,118)
Accretion of preferred stock to redemption value		(52,993)		(3,873)
Net loss attributable to common stockholders	\$	(56,847)	\$	(9,991)
Net loss per stock attributable to common stockholders – basic and diluted	\$	(4.32)	\$	(0.83)
Weighted-average number of stocks used in computing net loss per stock attributable to common stockholders		13,157,022		11,973,921
Weighted-average number of stocks used in computing net loss per stock attributable to common stockholders		13,157,022	_	11,973,921

The accompanying notes are an integral part of the unaudited interim condensed consolidated financial statements.

INNOVID, INC. AND ITS SUBSIDIARIES CONDENSED STATEMENTS OF CHANGES IN TEMPORARY EQUITY AND STOCKHOLDERS' DEFICIT

(In thousands, except stock data)

	Temporar	Temporary equity			Common stocks			Treasury stocks			lditional paid-	Δ	ccumulated	sto	Total ockholders'						
	Number	Am	ount	Number		Amount	Number		Amount										deficit	510	deficit
Balance as of January 1, 2021	55,105,773	\$	86,997	12,170,929	\$	12	1,431,538	\$	(1,629)	\$		\$	(48,113)	\$	(49,730)						
Accretion of preferred stocks to redemption value	_		52,993	_		_	_		_		(3,191)		(49,802)		(52,993)						
Stock-based compensation	_		_	_		_	_		_		2,311		_		2,311						
Stock options exercised	_		_	2,101,592		2	_		_		880		_		882						
Net loss	_		_	_		_	_		_		_		(3,854)		(3,854)						
Balance as of Sentember 30, 2021 (unaudited)	55,105,773	\$	139,990	14,272,521	\$	14	1,431,538	\$	(1,629)	\$		\$	(101,769)	\$	(103,384)						

	Temporary	Equity	Common	on stocks Treasury stocks Additional paid			Additional paid-	Accumulated	Total stockholders'	
	Number	Amount	Number	Amount	Number	Amount	in capital	deficit	deficit	
Balance as of January 1, 2020	55,105,773	\$ 79,700	11,941,841	\$ 12	1,431,538	\$ (1,629)	\$ 3,048	\$ (44,218)	\$ (42,787)	
Accretion of preferred stocks to redemption value	_	3,873	_	_	_	_	(3,873)	_	(3,873)	
Capital contribution	_	_	_	_	_	_	504	_	504	
Stock-based compensation	_	_	_	_	_	_	457	_	457	
Stock options exercised	_	_	47,920	_	_	_	30	_	30	
Net loss	_	_	_	_	_	_	_	(6,118)	(6,118)	
Balance as of September 30, 2020 (unaudited)	55,105,773	\$ 83,573	11,989,761	\$ 12	1,431,538	\$ (1,629)	\$ 166	\$ (50,336)	\$ (51,787)	

The accompanying notes are an integral part of the interim condensed consolidated financial statements.

^{*} Represents an amount less than \$1.

INNOVID, INC. AND ITS SUBSIDIARIES CONDENSED CONSOLIDATED STATEMENTS OF CASH FLOWS

(In thousands, except stock and per stock data)

	Nine months ended			d September 30,		
		2021		2020		
		(Unaudited)		(Unaudited)		
Cash flows from operating activities:						
Net loss	\$	(3,854)	\$	(6,118)		
Adjustments to reconcile net loss to net cash used in operating activities:		(2,02.)	<u> </u>	(0,110)		
Depreciation and amortization		487		475		
Stock-based compensation		2,311		457		
Change in fair value of warrants		3,191		51		
Changes in operating assets and liabilities						
Decrease/ (increase) in trade receivables, net		581		(115)		
(Increase)/ decrease in prepaid expenses and other assets		(1,587)		158		
Increase/ (decrease) in trade payables		710		(753)		
Increase in employees and payroll accruals		355		1,735		
Increase in accrued expenses and other liabilities		852		1,633		
Net cash provided by/ (used in) operating activities		3,046		(2,477)		
Cash flows from investing activities:						
Internal use software capitalization		(1,049)		_		
Founders' note receivable		(459)				
Purchase of property and equipment		(378)		(799)		
(Increase)/ decrease in deposits		(58)		54		
Net cash used in investing activities		(1,944)		(745)		
Cash flows from financing activities:						
Proceeds from loans		_		9,025		
Repayment of loans		(3,033)		_		
Proceeds from exercise of options		882		30		
Capital contribution		_		504		
Repayment of acquisition liability		(126)		_		
Net cash (used in)/ provided by financing activities		(2,277)		9,559		
(Decrease)/ increase in cash, cash equivalents and restricted cash		(1,175)		6,337		
Cash, cash equivalents and restricted cash at the beginning of the period		16,092		12,057		
Cash, cash equivalents and restricted cash at the end of the period	\$	14,917	\$	18,394		
Supplemental disclosure of cash flows activities:						
(1) Cash paid during the year for:						
Income taxes	\$	216	\$	221		
Interest	\$	189	\$	171		
(2) Non-cash transactions:						
Accrued acquisition liability	\$	_	\$	126		
Accretion of preferred stocks to redemption value	\$	52,993	\$	3,873		
Deferred offering cost included in accrued liabilities	\$	2,406	\$			
Reconciliation of cash, cash equivalents, and restricted cash reported within the statement of financial position	Ψ	2,100	<u> </u>			
Cash and cash equivalents	\$	14,472	\$	17,976		
Restricted cash in restricted deposits	Ψ	445	Ψ	418		
Total cash, cash equivalents, and restricted cash shown in the consolidated statements of cash flows	\$	14,917	\$	18,394		
rotal Cash, Cash equivalents, and restricted Cash shown in the consumated statements of Cash Hows	Ψ	17,717	Ψ	10,374		

The accompanying notes are an integral part of the interim condensed consolidated financial statements.

(In thousands, except stock and per stock data)

NOTE 1:- OVERVIEW

(a) Description of Business:

Innovid Inc. ("Innovid", Innovid together with its subsidiaries the "Company") was incorporated on June 21, 2007, under the General Corporation Law of the State of Delaware. The Company is a leading independent software platform that provides ad serving and creative services (together "Advertising Services") for the creation, delivery, and measurement of TV ads across connected TV ("CTV"), mobile TV and desktop TV environments to advertisers, publishers and media agencies.

On July 5, 2007, the Company established a wholly-owned subsidiary in Israel, Innovid Media Ltd. (the "Israeli Subsidiary"), which is mainly engaged in research and development ("R&D").

On November 12, 2012, the Company established a wholly-owned subsidiary in the United Kingdom (U.K.), Innovid EU Limited (the "U.K. Subsidiary"), which is engaged in business development, pre-sale and marketing services.

On October 21, 2013, the Company established a wholly-owned subsidiary in Australia, Innovid AU PTY LTD (the "the Australian Subsidiary"), which is engaged in business development, pre-sale and marketing services.

On September 12, 2019, the Company acquired 100% of the outstanding stocks of Dynamo Creative SRL (the "Argentinian Subsidiary" or "Dynamo Creative"), an Argentinian privately-held company which is engaged in R&D, business development and marketing services. The Argentinian subsidiary provides dynamic creative optimization services, a form of programmatic advertising that allows advertisers to optimize the performance of their creative services using real time technology.

(b) Description of Transaction:

On June 24, 2021, ION Acquisition Corp 2 Ltd., a Cayman Islands exempted company ("ION"), Innovid, Merger Sub 1 and Merger Sub 2 entered into the merger agreement ("Merger Agreement"). The Merger Agreement will be effectuated in the following principal steps:

- Merger Sub 1 will merge with and into Innovid, the separate corporate existence of Merger Sub 1 will cease and Innovid will be the surviving corporation (the "Surviving Corporation"),
- immediately thereafter, the Surviving Corporation will merge with and into Merger Sub 2, with Merger Sub 2 continuing as the surviving entity, which will remain a direct wholly owned subsidiary of ION,
- ION will change its name to "Innovid Corp.", pursuant to the terms and subject to the conditions set forth in the Merger Agreement, as more fully described elsewhere in the accompanying proxy statement/prospectus,
- the domestication of ION ("Domestication") as a Delaware corporation in accordance with the Delaware General Corporation Law ("DGCL"), the Cayman Islands Companies Act (As Revised) and the amended and restated memorandum and articles of association of ION, in which ION will effect a deregistration under the Cayman Islands Companies Act and a domestication under Section 388 of the DGCL (by means of filing a certificate of corporate domestication with the Secretary of State of Delaware), and
- · the consummation of other transactions contemplated by the Merger Agreement and documents related thereto.

Immediately prior to the Domestication, pursuant to the Cayman Constitutional Documents, each ION Class B Ordinary Stock, par value \$0.0001 per stock (each an "ION Class B Ordinary Stock") then issued and outstanding will automatically convert into one ION Class A Ordinary Stock, par value \$0.0001 per stock (each an "ION Class A Ordinary Stock") together with the ION Class B Ordinary Stock, the "ION Stock"). Following such conversion, as a result of the Domestication and the mergers, (a) each ION Unit then issued and outstanding as of immediately prior to the first merger will automatically be separated into the underlying ION Class A Ordinary Stock and ION warrant, (b) each ION Class A Ordinary Stock issued and outstanding immediately prior to the Domestication will remain outstanding and will automatically convert into one stock of Innovid Corp. common stock (provided that each ION Class A Ordinary Stock owned by public shareholders who have validly elected to redeem their ION Class A Ordinary Stocks will be redeemed for cash in an amount

(In thousands, except stock and per stock data)

equal to the redemption Price), (c) each ION warrant will automatically convert into a redeemable warrant exercisable for one stock of Innovid Corp. common stock on the same terms as the ION warrants, and (d) each whole Private Placement Warrant (as defined in the accompanying proxy statement/prospectus) issued and outstanding prior to the Domestication will automatically convert into a warrant exercisable for one stock of Innovid Corp. common stock on the terms and subject to the conditions set forth in the applicable warrant agreement. No fractional Innovid Corp. warrants will be issued upon separation of the ION Units.

As a result of the mergers, among other things, the aggregate consideration to be received in respect of the mergers by all of the stockholders and warrant holders of Innovid prior to the closing of the Transaction will be an aggregate of 76,874,354 stocks of Innovid Corp. Common Stock. In addition, pursuant to the Merger Agreement, at the closing of the Transaction (as defined below), immediately prior to the first merger, ION will purchase, and one or more stockholders of Innovid will sell to ION, in accordance with a stock purchase agreement, an aggregate amount of stock of common stock of Innovid, as determined by Innovid and for an aggregate purchase price determined by Innovid. The secondary sale amount will be determined by Innovid based on the amount of cash ION has on hand at the closing of the Transaction minus \$150,000, except if the amount of cash ION has on hand at the closing is equal to or less than \$150,000, the secondary sale amount will equal zero. The allocation of the secondary sale amount among Innovid equity holders and the amount of the secondary sale amount in excess of \$150,000, to the extent ION's cash on hand exceeds \$150,000, is subject to the discretion of the Innovid Board and compliance with the Innovid Equity Holders Support Agreements and each applicable maximum secondary sale amount.

In addition, ION entered into certain subscription agreements ("Initial PIPE Investment") with certain accredited and institutional investors, pursuant to which such investors have subscribed to purchase an aggregate of 15,000,000 stock of Innovid Corp. common stock, for a purchase price of \$10.00 per stock, to be issued immediately prior to or substantially currently with the closing. The merger and the PIPE Investment are collectively referred to as the "Transaction". On October 18, 2021, ION entered into new subscription agreements with certain PIPE Investors, including funds affiliated with ION, pursuant to which some of the PIPE Investors collectively subscribed for an additional 5,000,000 stocks of Innovid Corp. common stock for an aggregate purchase price equal to \$50,000 (the "Additional PIPE Investment" and together with the Initial PIPE Investment the "PIPE Investment"). This includes an additional 200,000 stocks purchased by funds affiliated with ION. The total anticipated proceeds from the PIPE Investment, after taking into account the Initial PIPE Investment and the Additional PIPE Investment, will total \$200,000. The PIPE Investment will be consummated following the Domestication but immediately prior to the closing of the Transaction.

The Transaction will be accounted for as a reverse recapitalization in accordance with U. S. GAAP. Under this method of accounting, ION will be treated as the "acquired" company for accounting purposes and the Transaction will be treated as the equivalent of Innovid issuing stock for the net assets of ION, accompanied by a recapitalization. Refer to the Note 10 Subsequent events for further information on the closing of the Transaction.

NOTE 2:- SIGNIFICANT ACCOUNTING POLICIES

(a) Basis of presentation:

The unaudited interim condensed consolidated financial statements and accompanying notes have been prepared in accordance with U.S. generally accepted accounting principles ("U.S. GAAP"). In management's opinion, the unaudited condensed consolidated financial statements reflect all adjustments of a normal recurring nature that are necessary for a fair presentation of the results for the interim periods presented. The Company's interim period results do not necessarily indicate the results that may be expected for any other interim period or for the full fiscal year. These unaudited interim condensed consolidated financial statements should be read in conjunction with the Company's audited consolidated financial statements.

The significant accounting policies applied in the annual consolidated financial statements of the Company as of December 31, 2020, have been applied consistently in these unaudited interim condensed consolidated financial statements, unless otherwise stated.

(b) Use of estimates:

The preparation of the consolidated financial statements in conformity with U.S. GAAP requires management to make estimates, judgments and assumptions. The Company's management believes that the estimates,

(In thousands, except stock and per stock data)

judgments and assumptions used are reasonable based upon information available at the time they are made. These estimates, judgments and assumptions can affect the reported amounts of assets and liabilities and disclosure of contingent assets and liabilities at the dates of the financial statements, and the reported amounts of revenues and expenses during the reporting period. Actual results could differ from those estimates.

The novel coronavirus ("COVID-19") pandemic has created, and may continue to create significant uncertainty in macroeconomic conditions, and the extent of its impact on the Company's operational and financial performance will depend on certain developments, including the duration and spread of the outbreak and the impact on Company's customers. Based on public reporting and Company's observations, some advertisers in certain industries, such as the automotive industry, decreased their short-term advertising spending in light of supply chain disruptions and/or labor shortage. This in turn could negatively impact the Company's revenues from such advertisers.

The Company have considered the impact of COVID-19 on its estimates and assumptions and determined that there were no material adverse impacts on the unaudited interim condensed consolidated financial statements for the nine months ended September 30, 2021 and year ended December 31, 2020. As events continue to evolve and additional information becomes available, the Company's estimates and assumptions may change materially in future periods.

The Company obtained an unsecured loan of \$3,516 in April 2020 due to uncertainties related to COVID-19. The loan was obtained through Silicon Valley Bank ("SVB") under the Paycheck Protection Program (the "PPP Loan") pursuant to the Coronavirus Aid, Relief, and Economic Security Act (the "CARES Act") and the Paycheck Protection Program Flexibility Act (the "Flexibility Act"). Under the terms of the CARES Act, recipients can apply for and receive forgiveness for all or a portion of loans granted under the PPP. Such forgiveness will be determined, subject to limitations, based on the use of loan proceeds for certain permissible purposes as set forth in the PPP, including, but not limited to, payroll costs (as defined under the PPP) and mortgage interest, rent or utility costs (collectively, "Qualifying Expenses") and the maintenance of employee and compensation levels ("Other Conditions"). The Company has been using the proceeds of the PPP Loan, for Qualifying Expenses and complied with Other Conditions.

In May 2020, the Company has entered into grant agreement ("Grant Agreement") with Special Situations Investing Group II, LLC ("SSIG") to receive \$504 from SSIG, related party of one of its investors, for the purpose of a partial repayment of the PPP Loan. The PPP loan was partially repaid in May 2020, according to the Grant Agreement.

The Company fully repaid the PPP loan in June, 2021 (unaudited).

(c) Goodwill and intangible assets:

Goodwill and certain other purchased intangible assets have been recorded in the Company's financial statements as a result of acquisitions. Goodwill represents excess of the purchase price in a business combination over the fair value of identifiable tangible and intangible assets acquired. Goodwill is not amortized, but rather is subject to an impairment test.

The Company allocates goodwill to reporting units based on the expected benefit from the business combination. Reporting units are evaluated when changes in the Company's operating structure occur, and if necessary, goodwill is reassigned using a relative fair value allocation approach. The Company currently has one reporting unit.

ASC 350, Intangible—Goodwill and other ("ASC 350") requires goodwill to be tested for impairment at least annually and, in certain circumstances, between annual tests. The accounting guidance gives the option to perform a qualitative assessment to determine whether further impairment testing is necessary. The qualitative assessment considers events and circumstances that might indicate that a reporting unit's fair value is less than its carrying amount. If it is determined, as a result of the qualitative assessment, that it is more likely than not that the fair value of a reporting unit is less than its carrying amount, a quantitative test is performed. The Company operates as one reporting unit. The Company elects to perform an annual impairment test of goodwill as of October 1 of each year, or more frequently if impairment indicators are present. For the nine months ended September 30, 2021 and 2020, no impairments of goodwill were recorded.

(In thousands, except stock and per stock data)

Separately acquired intangible assets are measured on initial recognition at cost including directly attributable costs. Intangible assets acquired in a business combination are measured at fair value at the acquisition date.

Intangible assets with a finite useful life are amortized over their useful life and reviewed for impairment whenever there is an indication that the asset may be impaired. For the nine months ended September 30, 2021 and 2020, no impairments of intangible assets were recorded.

(d) Software development costs:

Software development costs, which are included in property and equipment, net, consists of capitalized costs related to purchase and develop internal-use software. The Company uses it to provide services to its customers. The costs to purchase and develop internal-use software are capitalized from the time that the preliminary project stage is completed, and it is considered probable that the software will be used to perform the function intended. These costs include personnel and related employee benefits for employees directly associated with the software development and external costs of the materials or services consumed in developing or obtaining the software. Any costs incurred during subsequent efforts to upgrade and enhance the functionality of the software are also capitalized. Once this software is ready for use in providing the Company's services, these costs are amortized on a straight-line basis over the estimated useful life of the software, which is 3 years. The amortization will be presented within cost of revenues in the consolidated statements of operations. During the period ended September 30, 2021, the Company capitalized \$1,049 in internal-use software cost.

(e) Fair value of financial instruments:

The Company applies a fair value framework in order to measure and disclose its financial assets and liabilities. Fair value is defined as the exchange price that would be received for an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants on the measurement date. The fair value hierarchy requires an entity to maximize the use of observable inputs, where available, and minimize the use of unobservable inputs when measuring fair value. There are three levels of inputs that may be used to measure fair value:

- Level 1 Observable inputs that reflect quoted prices (unadjusted) for identical assets or liabilities in active markets.
- Level 2 Include other inputs that are directly or indirectly observable in the marketplace.
- Level 3 Unobservable inputs which are supported by little or no market activity.

The Company's financial instruments consist of cash and cash equivalents, restricted deposits, trade receivables, net, and trade payables. Their historical carrying amounts are approximate fair values due to the short-term maturities of these instruments.

The Company measures its investments in money market funds classified as cash equivalents and warrants liability at fair value.

(In thousands, except stock and per stock data)

The following table present information about the Company's financial instruments that are measured at fair value on a recurring basis:

		September 30, 2021	
		(Unaudited)	
	Level 1	Level 2	Level 3
Assets:			
Money market funds	\$ 11,013	\$ _	\$ _
Liabilities:			
Warrants liability	\$ _	\$ _	\$ 3,690
		December 31, 2020	
	Level 1	Level 2	Level 3
Assets:			
Money market funds	\$ 9,009	\$ _	\$ _
Liabilities:			
Warrants liability	\$ _	\$ _	\$ 499

The change in the fair value of the Warrants liability is summarized below:

	Se	September 30,		December 31
		2021		2020
	(Unaudited)		
Beginning of the period	\$	499	\$	413
Change in fair value		3,191		86
End of the period	\$	3,690	\$	499

The warrants were classified as level 3 in the fair value hierarchy because some of the inputs used in the valuation (the stock price) were determined based on management's assumptions. The Company estimates the fair value of the Warrants liability using Black-Scholes option pricing model. Gains and losses from the remeasurement of the warrants liability are recognized in finance expenses, net in the unaudited interim condensed consolidated statements of operations. As of September 30, 2021 (unaudited), and December 2020, the risk-free rate used for the valuation of the warrants was 0.07% and 0.1%, volatility used was 70% and 75%. The time to liquidation were 0.8 years for warrants related to Series A preferred stocks and 0.6 years related to Series C as of September 30, 2021 (unaudited). The time to liquidation were 1.6 years for warrants related to Series A preferred stocks and 1.5 years related to Series C as of December 31, 2020.

Fair value estimates are made at a specific point in time, based on relevant market information and information about the financial instruments. These estimates are subjective in nature and involve uncertainties and matters of significant judgment and, therefore, cannot be determined with precision. Changes in assumptions could significantly affect these estimates.

(f) Concentrations of credit risks:

Financial instruments that potentially subject the Company to concentrations of credit risk consist principally of cash and cash equivalents and trade receivables, net.

The majority of the Company's cash and cash equivalents are invested in deposits with major banks in America and Israel. Generally, these investments may be redeemed upon demand and, therefore, bear minimal risk.

The Company's trade receivables, net are mainly derived from sales to customers located in the U. S., Asia-Pacific region ("APAC"), Europe, the Middle East and Africa region ("EMEA"), and Latin America region ("LATAM"). The Company mitigates its credit risks by performing an ongoing credit evaluations of its customers' financial conditions.

The Company have no off-balance-sheet concentration of credit risk such as foreign exchange contracts, option contracts or other foreign hedging arrangements.

(In thousands, except stock and per stock data)

During the nine months ended September 30, 2021 (unaudited) and 2020 (unaudited), one of the Company's customers accounted for the Company's total revenues as presented below:

	Nine months end	ed September 30,
-	2021	2020
	(Unaudited)	(Unaudited)
Customer A	8 %	10 %

(g) Warrants:

The Company accounts for warrants as either equity-classified or liability-classified instruments based on an assessment of the warrant's specific terms and applicable authoritative guidance. The assessment considers whether the warrants are freestanding financial instruments, meet the definition of a liability under ASC 480, and meet all of the requirements for equity classification, including whether the warrants are indexed to the Company's own common stock and whether the warrant holders could potentially require "net cash settlement" in a circumstance outside of the Company's control, among other conditions for equity classification. This assessment, which requires the use of professional judgment, is conducted at the time of warrant issuance and as of each subsequent reporting period end date while the warrants are outstanding.

Warrants that meet all the criteria for equity classification, are required to be recorded as a component of additional paid-in capital. Warrants that do not meet all the criteria for equity classification, are required to be recorded as liabilities at their initial fair value on the date of issuance and remeasured to fair value at each balance sheet date thereafter. The liability-classified warrants are recorded under non-current liabilities. Changes in the estimated fair value of the warrants are recognized in "Financial expenses, net" in the consolidated statements of operations.

(h) Revenue recognition:

The Company generates revenues from providing Advertising Services to advertisers, publishers and media agencies. The services focus on standard, interactive and data driven digital video advertising. The Company major revenue streams are ad serving and creative services. Ad Serving services relate to utilizing Innovid's platform to serve advertising impressions to various digital publishers across CTV, mobile TV, desktop TV, display, and other channels. Creative services relate to the design and development of interactive data-driven and dynamic ad formats by adding data, interactivity and dynamic features to standard ad units.

The Company adopted ASC 606, Revenue from Contracts with Customers ("ASC 606") with a date of initial application of January 1, 2018, using the modified retrospective transition method, applied to all open contracts.

The Company recognizes revenue when its customer obtains control of promised services in an amount that reflects the consideration that the company expects to receive in exchange for those services. The Company recognizes revenue in accordance with ASC Topic 606, Revenue from contracts with customers ("ASC 606") and determines revenue recognition through the following steps: (1) identify the contract with a customer, (2) identify the performance obligations in the contract, (3) determine the transaction price, (4) allocate the transaction price to the performance obligations in the contract, and (5) recognize revenue when a performance obligation is satisfied.

For arrangements with multiple performance obligations, which represent promises within an arrangement that are capable of being distinct and are separately identifiable, the Company allocates the contract consideration to all distinct performance obligations based on their relative stand-alone selling price ("SSP").

Revenues related to ad serving services are recognized at a point in time. The Company recognizes revenue from the display of impression-based ads in the contracted period in which the impressions are delivered. Impressions are considered delivered when an ad is displayed to users.

Revenues related to creative services are recognized at a point in time, when the Company delivers an ad unit. Creative services projects are usually delivered within a week.

The Company's accounts receivable, consist primarily of receivables related to providing ad serving and creative services, in which the Company's contracted performance obligations have been satisfied, amount

(In thousands, except stock and per stock data)

billed and the Company has an unconditional right to payment. The Company typically bills customers on a monthly basis based on actual delivery. The payment terms vary, mainly with terms of net 60 days or less.

Typical contract term is twelve months or less for ASC 606 purposes. Some of the Company's contracts can be cancelled without a cause. The Company has unconditional right to payment for the services provided as of the date of the termination of the contracts.

The Company applies the practical expedient in ASC 606 and does not adjust the promised amount of consideration for the effects of a significant financing component if the Company expects, at contract inception, that the period between when the Company transfers a promised good or service to a customer and when the customer pays for that good or service will be one year or less.

Ad serving services were 93.8% and 97.0% of the Company's revenues for the nine months ended September 30, 2021 (unaudited) and 2020 (unaudited), respectively. Creative services were 4.8% and 2.3% for the nine months ended September 30, 2021 (unaudited) and 2020 (unaudited), respectively.

Costs to obtain a contract:

Contract costs include commission programs to compensate sales employees for generating sales orders with new customers or for new services with existing customers. The Company elected to apply the practical expedient and recognize incremental costs of obtaining a contract as an expense when incurred if the amortization period of the asset that the Company otherwise would have recognized is one year or less. The Company did not capitalize any contract costs during the nine months ended September 30, 2021 (unaudited) and 2020 (unaudited).

(i) Recently issued accounting pronouncements not yet adopted by the Company:

As an "emerging growth company", the Jumpstart Our Business Startups Act ("JOBS Act") allows the Company to delay adoption of new or revised accounting pronouncements applicable to public companies until such pronouncements are made applicable to private companies. The Company has elected to use this extended transition period under the JOBS Act. The Company have not adopted any new standards in the periods presented.

In August 2020, the FASB issued ASU 2020-06, Debt — Debt with Conversion and Other Options (Subtopic 470-20) and Derivatives and Hedging — Contracts in Entity's Own Equity (Subtopic 815-40): Accounting for Convertible Instruments and Contracts in an Entity's Own Equity ("ASU 2020-06"). The final guidance issued by the FASB for convertible instruments eliminates two of the three models in ASC 470-20 that require separate accounting for embedded conversion features. Separate accounting is still required in certain cases. Additionally, among other changes, the guidance eliminates some of the conditions for equity classification in ASC 815-40-25 for contracts in an entity's own equity. The guidance also requires entities to use the if-converted method for all convertible instruments in the diluted earnings per share calculation and include the effect of share settlement for instruments that may be settled in cash or shares, except for certain liability-classified share-based payment awards. ASU 2020-06 is effective for fiscal years beginning after December 15, 2023, and interim periods within those fiscal years. The Company is currently evaluating the potential impact of this guidance on its condensed consolidated financial statements.

In December 2019, the FASB issued ASU 2019-12, Income Taxes (Topic 740): Simplifying the Accounting for Income Taxes ("ASU 2019-12"). The new guidance eliminates certain exceptions related to the approach for intraperiod tax allocation, the methodology for calculating income taxes in an interim period and the recognition of deferred tax liabilities for outside basis differences. It also clarifies and simplifies other aspects of the accounting for income taxes. ASU 2019-12 is effective for fiscal years beginning after December 15, 2021. The Company is currently evaluating the potential impact of this guidance on its condensed consolidated financial statements.

In June 2016, the FASB issued ASU 2016-13, Financial Instruments – Credit Losses (Topic 326): Measurement of Credit Losses on Financial Instruments ("ASU 2016-13"). The ASU 2016-13 requires the measurement of all expected credit losses for financial assets held at the reporting date based on historical experience, current conditions, and reasonable and supportable forecasts. The ASU 2016-13 requires enhanced qualitative and quantitative disclosures to help investors and other financial statement users better understand significant estimates and judgments used in estimating credit losses, as well as the credit quality and underwriting standards

(In thousands, except stock and per stock data)

of an organization's portfolio. ASU 2016-13 is effective for fiscal years beginning after December 15, 2022. The Company is currently evaluating the potential impact of this guidance on its condensed consolidated financial statements.

In February 2016, the FASB issued an ASU 2016-02, Leases (Topic 842) ("ASU 2016-02"). ASC 842 changes the current lease accounting standard by requiring the recognition of lease assets and lease liabilities for all leases, including those currently classified as operating leases. This new guidance is effective for fiscal years beginning after December 15, 2021. The Company is currently evaluating the potential impact of this guidance on its condensed consolidated financial statements.

Other issued new guidance is not expected to have impact on the Company's consolidated financial statements.

NOTE 3:- OTHER NON-CURRENT LIABILITIES

Other non-current liabilities consist of the following:

	Septeml	ber 30, 2021	December 31, 2020
	Un	audited	
Accrued lease liability	\$	1,102	\$ 1,445
Tax provision		1,752	1,699
Total	\$	2,854	\$ 3,144

NOTE 4:- WARRANTS LIABILITY

In connection with a loan and security agreement entered into on June 29, 2010 with SVB, the Company issued warrants to purchase up to 165,654 of the Company's Series A preferred stocks, \$0.001 par value each, in the conversion ratio of 1:1 and at an exercise price of \$0.3622 subject to adjustments on the occurrence of stock splits, stock dividend, recapitalization, other dividends or distributions. In the event of an acquisition of the Company in which the sole consideration is cash and/or marketable securities, SVB shall have the right to exercise its conversion or purchase right in respect of the warrants. The agreement was amended on September 21, 2020 with exercisable period being extended until June 29, 2022. In lieu of exercising the warrants, SVB may convert the warrants, in whole or in part, into a number of shares determined by dividing (a) the aggregate fair market value of the shares or other securities issuable upon exercise of the warrants minus the aggregate warrant price of such shares by (b) the fair market value of one share. The loan has fully been repaid in 2012. The Company has determined that the loan and the warrants are freestanding financial instruments, as they are legally detachable and separately exercisable. The warrants were classified as a liability and were subsequently measured at fair value through earnings pursuant to ASC 480 "Distinguishing Liabilities from Equity". The loan was accounted for pursuant to ASC 470 "Debt".

In connection with a loan agreement entered into on April 23, 2014 with TriplePoint Capital LLC ("TPC loan agreement"), the Company issued warrants to purchase up to 162,409 of the Company's Series C preferred stocks, \$0.001 par value each, and at an exercise price of \$1.339 per stock or lower, subject to the next financing round stock price and provided that in no event shall the exercise price be lower than \$0.938. The warrants are exercisable for the later of (i) 7 years after date of issuance or (ii) 5 years of the effective date of the Company's initial public offering. The Company has determined that the loan and the warrants are freestanding financial instruments, as they are legally detachable and separately exercisable. The warrants were classified as a liability and were subsequently measured at fair value through earnings pursuant to ASC 480 "Distinguishing Liabilities from Equity". The loan was accounted for pursuant to ASC 470 "Debt".

On May 20, 2015 the Company entered into an amendment of the TPC loan agreement. In connection with the amendment, the Company issued warrants to purchase up to 80,645 of the Company's Series C preferred stocks \$0.001 par value each, and at an exercise price of \$1.339 per stock or lower subject to the next financing round stock price and provided that in no event shall the exercise price be lower than \$0.938. The warrants are exercisable for the later of (i) 7 years after date of issuance or (ii) 5 years of the effective date of the Company's initial public offering. In the event of an acquisition of the Company in which the sole consideration is cash and/or marketable securities, the Lender shall have the right to exercise its conversion or purchase right in respect of the warrants issued to the TPC. The TPC loan has been fully repaid in 2018. The Company has determined that

(In thousands, except stock and per stock data)

the loan and the warrants are freestanding financial instruments, as they are legally detachable and separately exercisable. The warrants were classified as a liability and were subsequently measured at fair value through earnings pursuant to ASC 480 "Distinguishing Liabilities from Equity". The loan was accounted for pursuant to ASC 470 "Debt".

The Warrants' fair value remeasurement for the nine months ended September 30, 2021 (unaudited) and 2020 (unaudited) were \$3,191 and \$51, respectively.

NOTE 5:- CREDIT LINE AND OTHER BORROWINGS

Credit Line:

In 2016, the Company entered into additional modifications to credit line agreement dated 2012 (the "Agreement"), pursuant to which certain conditions were amended, and the Maturity Date had been extended to October 21, 2018 and the line of credit increased from \$6,500 to \$10,000.

On April 7, 2017 the Company utilized \$5,000 of the line of credit. The credit installments bear U.S. dollar denominated interest at an annual rate equal to .75%-1% plus a prime rate on the outstanding principal of each credit installment. The balance owing as of December 31, 2017 was \$5,000.

On October 20, 2018, the Company entered into additional modifications to the Agreement, pursuant to which certain conditions were amended and the Maturity Date was extended to December 31, 2018.

On December 26, 2018, the Company entered into an amended and restated Agreement (the "A&A Agreement"), pursuant to which certain conditions were amended and the Maturity Date was extended to December 26, 2020 and the line of credit was increased to from \$10,000 to \$12,000.

On September 1, 2018 the Company utilized an additional \$1,000 of the line of credit. The credit installments bear U.S. dollar denominated interest at an annual rate equal to .75%-1% plus a prime rate on the outstanding principal of each credit installment. The Maturity Date was December 26, 2020. The balance owing as of December 31, 2018 was \$6,000.

On November 30, 2019, the Company fully repaid the outstanding balance of the credit line in the amount of \$6,000.

During 2020, the Company fully drew down on its \$12,000 credit line. As of December 31, 2020, the Company had repaid \$6,000, leaving a balance of \$6,000. On December 29, 2020, the Company entered into additional modifications to the A&A Agreement, pursuant to which certain conditions were amended and the Maturity Date was extended to December 29, 2022, and the line of credit increased to \$15,000.

As of September 30 2021 (unaudited) the outstanding balance of the credit line was in the amount of \$6,000. The credit installments bear U.S. dollar denominated interest at an annual rate equal to .75%-1% plus a prime rate on the outstanding principal of each credit installment. The Company was in compliance with all the covenants, primarily maintaining an adjusted quick ratio of at least 1.20:1.00. As defined in the A&A Agreement "adjusted quick ratio" is the ratio of (a) quick assets to (b) current liabilities minus the current portion of deferred revenue. "Quick assets" determines as Company's unrestricted cash plus accounts receivable, net, determined according to U.S. GAAP.

PPP Loan:

In April, 2020, the Company obtained an unsecured loan of \$3,516 through SVB under the PPP Loan. For more information see Note 2 (b).

In May, 2020, the Company have received a grant of \$504 from SSIG, related party of one of its investors, for the purpose of repayment of the portion of the PPP Loan. The PPP loan was partially repaid at in May 2020, according to the Grant Agreement.

In June, 2021, the Company has repaid the outstanding balance of PPP loan of \$3,012.

Interest expenses for the Credit Line and PPP Loan for the nine months ended September 30, 2021 (unaudited) and 2020 (unaudited) were \$197 and \$218, respectively and were recorded in finance expenses, net in the consolidated statements of operations.

(In thousands, except stock and per stock data)

NOTE 6:- COMMITMENTS AND CONTINGENT LIABILITIES

(a) Lease commitments

The Company leases office space and motor vehicles, which expire on various dates, the latest of which is in 2025. Future minimum lease commitments under non-cancelable operating leases as of September 30, 2021 (unaudited), are as follows:

Year ended September 30,	R	tental of premises	Lease of mo	otor vehicles
		Unaudited	Unau	udited
2021	\$	653	\$	7
2022		2,299		8
2023		1,801		
2024		796		
2025		742		
Total	\$	6,291	\$	15

Operating lease expenses for the nine months ended September 30, 2021 (unaudited) and 2020 (unaudited) totaled \$1,532 and \$1,820, respectively.

(b) Pledges and bank guarantees:

- 1. In conjunction with the Agreement and its amendments (see Note 5), Innovid pledged 65,000 common stocks of its Israeli Subsidiary, NIS 0.01 par value each.
- 2. Israeli Subsidiary pledged bank deposits in an aggregate amount of \$679 in connection with an office rent agreement and credit cards.
- 3. Innovid obtained bank guarantees in an aggregate amount of \$251 in connection with its office lease agreements.

NOTE 7:- STOCK-BASED COMPENSATION

Under the Company's Stock Option Plan (the "Plan"), options may be granted to officers, directors, employees and non-employee consultants of the Company. Each option granted under the Plan expires no later than 10 years from the date of grant. The options vest usually over four years from commencement of employment or services. Any options, which are forfeited or not exercised before expiration, become available for future grants.

A summary of the employees' stock option activity is as follows:

(Unaudited)	Number of options	Weighted-average exercise price	Remaining contractual term	Aggregate intrinsic value
Outstanding at December 31, 2020	9,112,121	\$ 0.49	7.20	\$ 3,100
Granted	1,109,750	3.76		
Exercised	(1,899,793)	0.29		
Forfeited	(163,330)	1.18		
Expired	(47,943)	0.79		
Outstanding at September 30, 2021	8,110,805	\$ 0.97	5.86	\$ 65,412
Exercisable at September 30, 2021	4,867,399	\$ 0.50	4.50	\$ 41,511

(In thousands, except stock and per stock data)

A summary of the consultants' stock option activity under the Plan is as follows:

(Unaudited)	Number of options	Veighted-average exercise price	Remaining contractual term	Aggre	gate intrinsic value
Outstanding at December 31, 2020	762,248	\$ 0.68	6.32	\$	213
Granted	154,502	3.76			
Exercised	(201,799)	1.67			
Forfeited	(20,625)	0.82			
Outstanding at September 30, 2021	694,326	\$ 1.08	6.46	\$	5,523
Exercisable at September 30, 2021	469,659	\$ 0.78	5.72	\$	3,876

As of September 30, 2021, the Company had approximately \$4,537 of total unrecognized compensation cost related to non-vested stock-based compensation. That cost is expected to be recognized over a weighted-average period of 2.62 years.

A summary of the employees' stock option activity under the Plan for the nine months ending September 30, 2021 (unaudited) and 2020 (unaudited) is as follows:

		Nine months end	led Sep	tember 30,
	<u>-</u>	2021		2020
		Unaudited		Unaudited
Cost of goods sold	\$	34	\$	10
Research and development		319		113
Sales and marketing		400		261
General and administrative		1,285		33
Total	\$	2,038	\$	417

In connection with the options granted to service providers and non-employee consultants, during the nine months ended September 30, 2021 (unaudited) and 2020 (unaudited), the Company recorded stock compensation expenses in the amount of \$273 and \$40, respectively. Majority of these expenses were recorded in general and administrative expenses.

In the nine months ended September 30, 2021, the Company's Board approved an amendment of two awards granted to the Company's founders Mr. Zvika Netter, and Mr. Tal Chalozin ("Founders Awards"). According to amendments the Founders Awards will vest over three years (four years originally), with 75% of the options vesting upon expiration of one year from the original commencement date of April 1, 2020 and the remaining 25% of the options vesting ratably on a quarterly basis over the following 24 months. In addition, upon a consumption of the transaction as defined by the Plan, if the founders are terminated or leave for "good reason" within 12 months, the remaining unvested awards would vest. In addition, the amendment also included a provision in which any termination of employment (whether by the Company or by the founder), 50% of his unvested award will vest immediately. The amendments were accounted for as a modification. The Company determined that the amendments did not result in an increase in the fair value of the award. The modified vesting conditions resulted in an additional expense of \$623.

In April 2021, the Company's Board approved a transaction in which the Company granted \$1,199 and received a secured full recourse promissory note in the total aggregate amount of \$1,199, with Mr. Zvika Netter, and Mr. Tal Chalozin (the "Founders Promissory Note"). On June 7, 2021, Innovid granted Mr. Netter a loan in the amount of \$1,076 pursuant to the Founder Promissory Note ("Zvika Netter Loan"). On June 23, 2021, Innovid granted Mr. Chalozin a loan in the amount of \$123 pursuant to the Found Promissory Note (the "Tal Chalozin Loan", together with Zvika Netter Loan the "Founders Loans"). The principal balances together with accrued interest is due and payable in full on the seventh anniversary of the date of the loans. The rate is 0.89% per annum, compound annually and is not less than the current minimum annual mid-term applicable federate rate established pursuant to Section 1274(d) of the Internal Revenue Code of 1986, as amended. Repayment of principal and interest may be made at any time without penalty. In addition, \$740 of the Founders Loans was immediately used to exercise fully vested options held by the founders.

(In thousands, except stock and per stock data)

The Founders Loans are expected to be repaid in full in connection with Closing. This loan represents a recourse note as the Company has a contractual full recourse right against any real, personal, tangible or intangible assets of the Borrowers and intends to so if the loans amount will not be repaid in full.

The amount of \$459 from the Founders Loans was not used by the founders to exercise stock options.

Under ASC 718, when a grantee purchases shares in exchange for a recourse loan, the exercise is considered to be a substantive exercise. A recourse note receivable for the issuance of equity should be presented in accordance with the guidance in ASC 505-10-45 as a component of equity; Thus, the Company recognized the note receivable for the purchase of shares as a component of additional paid in capital. The amount was discounted to its fair value and additional stock-based compensation expense in the amount of \$47 was recorded predominantly in general and administrative expenses.

The amount of the loan not used to exercise stock options in the amount of \$459 was accounted for as a standard loan and is presented as a non-current asset in these unaudited condensed consolidated financial statements.

NOTE 8: SEGMENT REPORTING

The Company operates as one operating segment, which primarily focuses on advertising and creative services. Our Chief Executive Officer ("CEO"), is the chief operating decision-maker, manages and allocates resources to the operations of the Company on an entity-wide basis. Managing and allocating resources on an entity-wide basis enables the CEO to assess the overall level of resources available and how to best deploy these resources across functions and R&D projects based on needs and, as necessary, reallocate resources among the Company's internal priorities and external opportunities to best support the long-term growth of the business.

Revenue by geographical location are as follows:

	Nine months ended September 30,			
	 2021		2020	
	Unaudited		Unaudited	
U.S.	\$ 58,270	\$	41,853	
Canada	799		381	
APAC	2,182		1,660	
EMEA	1,842		940	
LATAM	1,231		938	
Total	\$ 64,324	\$	45,772	

The Company's property and equipment, net by geographical location are as follows:

	S	eptember 30, 2021	December 31, 2020
		Unaudited	
Israel	\$	1,593	\$ 1,625
U. S.		1,331	595
Rest of the World		374	105
Total	\$	3,298	\$ 2,325

NOTE 9:- BASIC AND DILUTED NET LOSS PER SHARE

Basic and diluted net loss per share attributable to common stockholders was calculated as follows:

(In thousands, except stock and per stock data)

	Nine months ended September 30,			tember 30,
	2021		2020	
		Unaudited		Unaudited
Numerator:				
Net loss	\$	(3,854)	\$	(6,118)
Accretion of preferred stocks to redemption value		(52,993)		(3,873)
Net loss attributable to common stockholders	\$	(56,847)	\$	(9,991)
Denominator:				
Weighted-average number of stocks used in computing net loss per stock attributable to common stockholders		13,157,022		11,973,921
Net loss per stock attributable to common stockholders – basic and diluted	\$	(4.32)	\$	(0.83)

The Company's potentially dilutive securities have been excluded from the computation of diluted net loss per share as the effect would be to reduce the net loss per share attributable to common stockholders. Therefore, the weighted average number of common shares outstanding used to calculate both basic and diluted net loss per share attributable to common stockholders is the same.

The Company excluded the following potential common shares, presented based on amounts outstanding at each period end, from the computation of diluted net loss per share attributable to common stockholders for the periods indicated because including them would have had an anti-dilutive effect:

	September	30,
	2021	2020
	Unaudited	Unaudited
Preferred stocks	55,105,773	55,105,773
Options outstanding	8,805,131	6,325,006
Warrants outstanding	508,708	508,708

NOTE 10:- SUBSEQUENT EVENTS

On October 18, 2021, ION entered into new subscription agreements with certain PIPE Investors, including funds affiliated with ION, pursuant to which some of the PIPE Investors collectively subscribed for an additional 5,000,000 shares of stock of Innovid Corp. Common Stock for an aggregate purchase price equal to \$50,000. The total anticipated proceeds from the PIPE Investment will total \$200,000.

Founders Loans with a total principal amount of \$1,199 and related interest were forgiven in November 2021. \$740 of the Founders Loans principal amount was used to exercise fully vested options held by the founders on the date of the grant of the Founders Loans and the remainder in the amount of \$459 was used for other purposes as described in detail in Note 7 Stock-based compensation.

On November 30, 2021, as contemplated by the Merger Agreement, ION consummated the merger transaction contemplated by the Merger Agreement (the "Closing"), whereby (i) Merger Sub 1 merged with the Company (the "Merger") with the Company continuing as the surviving corporation of the Merger, (ii) following the Merger, ION changed its name to "Innovid Corp." (the "Name Change"), (iii) following the Merger and the Name Change, Innovid Corp. (formally ION) issued 86,901,792 shares of common stock (the "Registered Shares"), par value \$0.0001 per share ("Common Stock") and (iii) Innovid Corp. (formally ION) issued 20,000,000 shares of Common Stock to PIPE Investors.

Pursuant to the Merger Agreement, immediately prior to the merger, each issued and outstanding share of Ion Class B Ordinary Stock automatically converted, on a one-for-one basis, into one (1) share of ION Class A Ordinary Stock in accordance with the terms of ION's organizational documents. Immediately following such conversion, upon the Domestication, (i) each then issued and outstanding share of ION Class A Ordinary Stock automatically converted, on a one-for-one basis, into a share of common stock of Ion (after the Domestication) (the "Company Domesticated Common Stock"), (ii) each issued and outstanding Ion Warrant automatically converted into one corresponding warrant to acquire one (1) share of the Company Domesticated Common Stock (the "Company Domesticated Warrant") and (iii) each issued and outstanding unit representing one (1) share of ION Class A Ordinary Stock and one-eighth (1/8) of an ION Warrant automatically converted into one

(In thousands, except stock and per stock data)

(1) unit of the Company (after the Domestication) representing one (1) Company Domesticated Common Stock and one-eighth (1/8) of an the Company Domesticated Warrant. No fractional Company Domesticated Warrants were issued in connection with such conversion such that if a holder of such units was entitled to receive a fractional Domesticated Acquirer Warrant, the number of Domesticated Acquirer Warrants to be issued to such holder upon such conversion was rounded down to the nearest whole number of Domesticated Acquirer Warrants.

The Company has evaluated subsequent events from the balance sheet date through December 6, 2021, the date at which the consolidated financial statements were available to be issued.

UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

Introduction:

On June 24, 2021, ION Acquisition Corp 2 Ltd., a Cayman Islands exempted company ("ION"), entered into an Agreement and Plan of Merger (the "Merger Agreement") with Inspire Merger Sub 1, Inc., a Delaware corporation and a direct, wholly owned subsidiary of ION ("Merger Sub 1"). The following unaudited pro forma condensed combined financial statements are provided to aid you in your analysis of the financial aspects of the consummation of the Merger and the PIPE Investment (collectively referred to as the "Transaction").

The unaudited pro forma condensed combined financial information is prepared in accordance with Article 11 of Regulation S-X as amended by the final rule, Release No. 33-10786 "Amendments to Financial Disclosures about Acquired and Disposed Businesses" and should be read in conjunction with the accompanying notes. The adjustments presented in the unaudited pro forma condensed combined financial information have been identified and presented to provide relevant information necessary for an understanding of the combined company upon consummation of the transactions described in this proxy statement/prospectus. The unaudited pro forma condensed combined financial information presents the pro forma effects of the acquisition of Innovid Inc. ("Innovid") by ION and other agreements entered into as part of the Merger Agreement, as though such transactions occurred on January 1, 2020.

ION was incorporated on November 23, 2020 (inception) as a Cayman Islands exempted company and a blank check company for the purpose of effecting a merger, share exchange, asset acquisition, share purchase, reorganization or similar transaction with one or more businesses. On December 1, 2020, the Founder Shares (an aggregate of 5,750,000 ION Class B Ordinary Shares) were sold to the Sponsor at a price of approximately \$0.004 per share, for an aggregate price of \$25,000. In January 2021, the Sponsor transferred 25,000 Founder Shares to each of Mr. Seligsohn, Ms. Gazit and Mr. Shemesh at their original purchase price. Prior to the initial investment in the company of \$25,000 by the Sponsor, ION had no assets, tangible or intangible. In January 2021, ION effected a share capitalization of 575,000 shares and, as a result, there are 6,325,000 Founder Shares issued and outstanding. On February 16, 2021, ION completed its initial public offering of 25,300,000 ION Units, including the issuance of 3,300,000 ION Units as a result of the underwriters' exercise in full of their over-allotment option (the "IPO"). Each ION Unit consists of one ION Class A Ordinary Share and one-eighth of one warrant, with each whole warrant entitling the holder thereof to purchase one ION Class A Ordinary Share for \$11.50 per share. The ION Units were sold at a price of \$10.00 per ION Unit, generating gross proceeds to ION of \$253,000,000. In addition, prior to the closing of the IPO, ION completed the sale of the warrants exercisable for one ION Class A Ordinary Share at \$11.50 per share (the "ION Warrants") to ION Holdings 2, LP (the "Private Placement Warrants") (7,060,000 warrants at a price of \$1.00 per warrant) in a private placement to the Sponsor, generating gross proceeds of \$7,060,000. A total of \$253,000,000 of the net proceeds from the IPO and the Private Placement Warrants were deposited in a trust account (the "Trust Account") established for the benefit of the holders of ION Class A Ordinary Shares sold in the IPO and the rem

Founded in 2007, Innovid is an independent software platform that provides critical technology infrastructure for the creation, delivery, and measurement of TV ads across connected TV ("CTV"), mobile TV and desktop TV environments. Innovid's purpose-built CTV infrastructure platform is comprised of three key offerings: Ad Serving Solutions, Creative Personalization Solutions and Measurement Solutions. Innovid's software-based platform provides an open technology infrastructure that tightly integrates with the highly fragmented advertising technology and media ecosystem including Demand Side Platforms (DSPs) such as The Trade Desk and Amobee; Supply Side Platforms (SSP) such as Magnite and Verizon Media; publishers such as Hulu and Peacock; and end user devices such as Amazon Fire and Samsung Smart TV. Innovid's offering encompasses independent global ad serving, data-driven personalization, and new forms of measurement designed to connect all channels in a clean, comparable, and privacy-compliant manner.

The unaudited pro forma condensed combined financial information is based upon, and should be read in conjunction with, the historical financial statements and related notes of ION and Innovid for the applicable periods included in this proxy statement/prospectus. The pro forma financial statements have been presented for information purposes only and are not necessarily indicative of what Innovid Corp.'s balance sheet or statement of operations actually would have been had the Transaction been completed as of the dates indicated, nor do they purport to project the future financial position or operating results of Innovid Corp. The actual financial position and results of operations may differ significantly from the pro forma amounts reflected herein due to a variety of factors. The pro forma financial information is presented for illustrative purposes only and does not reflect the costs of any integration activities or cost savings or synergies that may be achieved as a result of the Transaction.

The unaudited pro forma condensed combined statement of operations for the year ended December 31, 2020 and for the nine months ended September 30, 2021 present the pro forma effect of the Transaction as if the Closing occurred on January 1, 2020.

The unaudited pro forma condensed combined balance sheet combines the Innovid unaudited historical condensed consolidated balance sheet as of September 30, 2021 and the ION unaudited historical consolidated balance sheet as of September 30, 2021, giving effect to the Transaction as if it had been consummated on September 30, 2021.

We refer to the unaudited pro forma condensed combined balance sheet and the unaudited pro forma condensed combined statement of operations as the pro forma financial statements

The unaudited pro forma condensed combined financial information has been prepared based on actual redemptions o 19,585,174 ION outstanding Public Shares for aggregate redemption payments of \$195.9 million out of the trust account at a redemption price of \$10.00 per share on the closing date of the transactions contemplated by the Merger Agreement.

Shares outstanding as presented in the unaudited pro forma condensed combined financial statements include the86,901,792 shares of Company Common Stock (as defined below) issued to Innovid's stockholders, the 12,039,826 shares of common stock of ION, now Company Common Stock (after giving effect to the redemption of 19,585,174 shares of ION common stock), and including the 20,000,000 shares of Company Common Stock issued in connection with the PIPE Transaction.

As a result of the Transaction, including the redemption of 19,585,174 shares of ION common stock, Innovid's stockholders own approximately 73% of the common stock of the combined company, ION public stockholders own approximately 17% of the common stock of the combined company, and investors from the PIPE Transaction own approximately 10% of the common stock of the combined company, based on the number of shares of Company Common Stock outstanding as of September 30, 2021 (in each case, not giving effect to any shares issuable upon exercise of Innovid or ION Options).

Description of the Transaction

On June 24, 2021, ION, Innovid, Merger Sub 1 and Inspire Merger Sub 2, LLC ("Merger Sub 2") entered into the Merger Agreement. On November 30, 2021, as contemplated by the Merger Agreement, ION consummated the merger transaction contemplated by the Merger Agreement pursuant to the following steps (the "Closing"):

- Merger Sub 1 merged with and into Innovid (the "First Merger"), the separate corporate existence of Merger Sub 1 ceased with Innovid continuing as the surviving corporation (the "Surviving Corporation"),
- immediately thereafter, the Surviving Corporation merged with and into Merger Sub 2 (collectively with the First Merger, the "Mergers"), with Merger Sub 2 continuing as the surviving entity and a direct wholly owned subsidiary of ION,
- ION changed its name to "Innovid Corp." (the "Name Change"), pursuant to the terms and subject to the conditions set forth in the Merger Agreement, as more fully described elsewhere in the accompanying proxy statement/prospectus,

- prior to the consummation of the Mergers, ION was redomiciled (the "Domestication") as a Delaware corporation in accordance with the Delaware General Corporation Law, the Cayman Islands Companies Act (As Revised) and the amended and restated memorandum and articles of association of ION (the "Cayman Constitutional Documents"), in connection with which ION effected a deregistration under the Companies Act and a domestication under Section 388 of the DGCL (by means of filing a certificate of corporate domestication with the Secretary of State of Delaware), and
- · the other transactions contemplated by the Merger Agreement and documents related thereto were consummated.

In connection with the Mergers and the Name Change, Innovid Corp. (formally ION) issued (a) 86,901,792 shares of common stock, par value \$0.0001 per share ("Company Common Stock") to former equityholders of Innovid Inc. and (b) 20,000,000 shares of Company Common Stock to PIPE Investors (as defined below)

Pursuant to the Merger Agreement, immediately prior to the Domestication, pursuant to the Cayman Constitutional Documents, each ION Class B Ordinary Share, par value \$0.0001 per share (each an "ION Class B Ordinary Share") then issued and outstanding automatically converted into one ION Class A Ordinary Share, par value \$0.0001 per share (each an "ION Class A Ordinary Share" together with the ION Class B Ordinary Shares, the "ION Shares"). Immediately following such conversion, upon the Domestication and the Mergers, (i) each issued and outstanding unit representing one (1) ION Class A Share and one-eighth (1/8) of the warrant to purchase one (1) ION Class A Share at a price of \$11.50 per share (the "ION Warrants") was automatically separated into the underlying ION Class A Share and one-eighth of an ION Warrant, (ii) each ION Class A Share issued and outstanding immediately prior to the Domestication was automatically converted into one share of Company Common Stock, (iii) each whole ION Warrant was automatically converted into a redeemable warrant exercisable for one share of Company Common Stock on the same terms as the ION Warrants (the "Public Warrants"), and (iv) each whole private placement warrant, exercisable for one ION Class A Share at \$11.50 per share, issued and outstanding prior to the Domestication was automatically converted into a warrant exercisable for one share of Company Common Stock on the terms and subject to the conditions set forth in the applicable warrant agreement (the "Private Placement Warrants") and together with the Public Warrants, the "Company Warrants"). No fractional Company Warrants were issued upon separation of the ION Units.

As a result of the Mergers, among other things, the aggregate consideration to be received in respect of the Mergers by all of the stockholders and warrant holders of Innovid prior to the Closing was an aggregate of 86,901,792 shares of Innovid Corp. Common Stock. In addition, in connection with closing, ION purchasedequity securities of Innovid Stockholders for an aggregate purchase price of \$68.9 million (the "Secondary Sale Amount").

The following summarizes the pro forma ownership of Innovid Corp. Common Stock following the Closing:

	Actual Redempt	Actual Redemptions			
Equity Capitalization Summary (shares in thousands)	Shares	%			
ION Shareholders ⁽¹⁾	12,039,826	10 %			
PIPE Investors ⁽²⁾	20,000,000	17 %			
Innovid Equity Holders' interests in ION ³⁾	86,901,792	73 %			
Total common stock in Innovid Corp.	118.941.618				

- (1) Excluding 4,200,000 shares of Innovid Corp. Common Stock purchased by affiliates of the Sponsor in connection with the PIPE Investment.
- (2) Represents the PIPE Investment, including 4,200,000 shares of Innovid Corp. Common Stock purchased by affiliates of the Sponsor.
- (3) Represents existing Innovid equity holders' interest in shares of Innovid Corp. Common Stock.

Accounting Treatment for Transaction

The Transaction was accounted for as a reverse recapitalization in accordance with U.S. GAAP. Under this method of accounting, ION was treated as the "acquired" company for accounting purposes and the Transaction was treated as the equivalent of Innovid issuing stock for the net assets of ION, accompanied by a recapitalization. The

net assets of ION were stated at historical cost, with no goodwill or other intangible assets recorded. The public warrants and the private placement warrants of ION have been reported as liability-classified instruments that will be subsequently remeasured at fair value in future reporting periods, with changes in fair value recognized in earnings.

Innovid has been determined to be the accounting acquirer based on evaluation of the following facts and circumstances:

- Innovid's existing equity holders will have the greatest voting interest in the combined entity;
- · The largest individual minority equity holder of the combined entity is an existing equity holder of Innovid;
- Innovid's directors will represent the majority Innovid Corp. board of directors;
- · Innovid's senior management will be the senior management of Innovid Corp.; and
- Innovid is the larger entity based on historical revenue and has the larger employee base.

The preponderance of evidence as described above is indicative that Innovid is the accounting acquiror in the Transaction.

UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET

As of September 30, 2021 (in thousands, except stock and per stock amounts)

	Historical	Historical Historical	Actual Redemptions		
	ION	Innovid	Transaction Accounting Adjustments	Pro Forma Balance Sheet	
ASSETS			,		
CURRENT ASSETS					
Cash and cash equivalents	184	14,472	253,043 (a)	162,992	
			200,000 (b)		
			(27,625) (c)		
			(12,375) (d)		
			(195,852) (i)		
			(68,855) (h)		
Trade receivables, net	_	34,223		34,223	
Prepaid expenses and other current assets	526	1,966		2,492	
Total current assets	710	50,661	148,336	199,707	
NON-CURRENT ASSETS:			·		
Cash and marketable securities held in Trust Account	253,043	_	(253,043) (a)	_	
Long-term lease and other deposit	_	317		317	
Long-term restricted deposits	_	445		445	
Property and equipment, net	_	3,298		3,298	
Goodwill	_	4,555		4,555	
Prepaid offering cost	_	3,269	(3,269) (d)	_	
Other non-current assets	_	607		607	
Total non-current assets	253,043	12,491	(256,312)	9,222	
TOTAL ASSETS	253,753	63,152	(107,976)	208,929	
LIABILITIES AND STOCKHOLDERS' EQUITY/(DEFICIT)					
CURRENT LIABILITIES:					
Trade payables	_	2,564		2,564	

UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET (CONT.)

As of September 30, 2021 (in thousands, except stock and per stock amounts)

	Historical	Historical	Actual Redemptions		
	ION	Innovid	Transaction Accounting Adjustments	Pro Forma Balance Sheet	
Employees and payroll accruals	_	6,861		6,861	
Accrued expenses and other current liabilities	3,230	2,171	(3,230) (d)	2,171	
Accrued offering costs	319	_	(319) ^(d)	_	
Deferred offering cost accrual	_	2,406	(2,406) (d)	_	
Total current liabilities	3,549	14,002	(5,955)	11,596	
NON-CURRENT LIABILITIES:					
Long-term debt	_	6,000		6,000	
Other non-current liabilities	_	2,854		2,854	
Warrants liability	29,640	3,690	(3,690) (i)	29,640	
Deferred underwriting fee payable	8,855	_	(8,855) (d)	_	
Total non-current liabilities	38,495	12,544	(12,545)	38,494	
TOTAL LIABILITIES	42,044	26,546	(18,500)	50,090	
COMMITMENTS AND CONTINGENT LIABILITIES					
Innovid preferred stocks	_	139,990	(139,990) (i)	_	
ION Class A ordinary shares subject to possible redemption	253,000	_	(57,148) (f)	_	
			(195,852) ()		
ION preference shares	_	_	_	_	
ION Class B ordinary shares	1	_	(1) (e)	_	
Innovid common stocks	_	14	(14) (i)	_	
Innovid treasury stocks	_	(1,629)	1,629 (i)	_	
Innovid Corp. Class A common Stock	_	_	2 (b)	12	

UNAUDITED PRO FORMA CONDENSED COMBINED BALANCE SHEET (CONT.)

As of September 30, 2021 (in thousands, except stock and per stock amounts)

	Historical	Historical	Actual Redemptions		
	ION	Innovid	Transaction Accounting Adjustments	Pro Forma Balance Sheet	
			1 (f)		
			1 (e)		
			8 (i)		
Additional paid-in capital	_	_	(27,625) (c)	260,596	
			199,998 (b)		
			57,147 ^(f)		
			(834) ^(d)		
			(41,292) (g)		
			(68,855) (h)		
			142,057 (i)		
Accumulated deficit	(41,292)	(101,769)	41,292 (g)	(101,769)	
Total stockholders' equity/(deficit)	(41,291)	(103,384)	303,514	158,839	
TOTAL LIABILITIES AND STOCKHOLDERS' EQUITY /(DEFICIT)	253,753	63,152	(107,976)	208,929	

UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS

For the Year Ended December 31, 2020 (in thousands, except stock and per stock amounts)

	Historical	Historical	Actual Redemptions		
	ION	Innovid	Transaction Accounting Adjustments	Pro Forma Statement of Operations	
Revenues		68,801		68,801	
Cost of revenues	_	12,365		12,365	
Gross profit	_	56,436	_	56,436	
Operating expenses:					
Research and development	_	18,283		18,283	
Sales and marketing	_	28,810		28,810	
General and administrative	_	8,221	6,000 (ac)	14,221	
Operating costs	5			5	
Total operating expenses	5	55,314	6,000	61,319	
Operating (loss) profit	(5)	1,122	(6,000)	(4,883)	
Finance expenses, net	_	734		734	
Income (loss) before taxes	(5)	388	(6,000)	(5,617)	
Taxes on Income	_	1,200		1,200	
Net loss	(5)	(812)	(6,000)	(6,817)	
Net loss per stock attributable to common stockholders - basic and diluted	<u> </u>	(0.07)		(0.06) (ab)	
Weighted average common stock used to compute net loss per stock attributable to common stockholders	5,500,000	11,986,185		118,972,233 (ab)	

UNAUDITED PRO FORMA CONDENSED COMBINED STATEMENT OF OPERATIONS

For the Nine Months Ended September 30, 2021 (in thousands, except stock and per stock amounts)

	Historical	Historical	Actual Redemptions		
	ION	Innovid	Transaction Accounting Adjustments	Pro Forma Statement of Operations	
Revenues		64,324		64,324	
Cost of revenues	_	12,418		12,418	
Gross profit		51,906	_	51,906	
Operating expenses:					
Research and development	_	16,932		16,932	
Sales and marketing	_	23,534		23,534	
General and administrative	_	10,587		10,587	
Operating costs	4,336	_	_	4,336	
Total operating expenses	4,336	51,053	_	55,389	
Operating (loss) profit	(4,336)	853	_	(3,483)	
Finance expenses, net	_	3,878		3,878	
Interest income on marketable securities held in Trust Account	(43)	_	43 (aa)	_	
Underwriting discounts and transactions costs attributed to warrants liability	300	_		300	
Change in fair value of the Warrant Liabilities	(425)	_		(425)	
Loss before taxes	(4,168)	(3,025)	(43)	(7,236)	
Taxes on Income	_	829		829	
Net loss	(4,168)	(3,854)	(43)	(8,065)	
Net loss per stock attributable to common stockholders - basic and diluted	(0.67)	(4.32)		(0.07) (ab)	
Weighted average common stock used to compute net loss per stock attributable to common stockholders	6,182,967	13,157,022		118,972,233 (ab)	

NOTES TO UNAUDITED PRO FORMA CONDENSED COMBINED FINANCIAL INFORMATION

1. Adjustments to Unaudited Pro Forma Condensed Combined Balance Sheet as of September 30, 2021

The pro forma notes and adjustments, based on preliminary estimates that could change materially as additional information is obtained, are as follows:

Pro forma Transaction Accounting Adjustments:

- (a) Reflects the reclassification of the Trust Account to cash and cash equivalents that becomes available at the time of the Transaction.
- (b) Reflects the proceeds of \$200.0 million from the issuance and sale of 20,000,000 stocks of Innovid Corp.'s Common Stock at a par value of \$0.0001 in the PIPE Investment pursuant to the Subscription Agreements.
- (c) Reflects the payment of transaction costs incurred by Innovid and ION for legal, financial advisory and other professional fees reflected as a decrease in additional paid-in capital.
- (d) Reflects the settlement of ION and Innovid's obligations, including \$3.2 million in accrued expenses and other current liabilities, \$0.3 million in accrued offering costs and deferred offering cost of \$2.4 million, and a reclassification of deferred transaction cost of \$3.3 million from prepaid offering cost to additional paid in capital. In addition, reflects payment of deferred underwriting fees of \$6.4 million. The initial estimate of deferred underwriting fees as of September 30, 2021 was \$8.9 million, therefore the difference between the estimate and actual amounts of deferred underwriting fees of \$2.5 million was reversed to additional paid in capital.
- (e) Reflects the conversion of ION's Class B ordinary shares into 6,325,000 Innovid Corp' Class A Common Stock at a par value of \$0.0001.
- (f) Reflects the reclassification of ordinary shares subject to redemption of \$57.2 million to common stock of \$1 thousand with the difference to additional paid in capital.
- (g) Reflects the elimination of ION's historical accumulated deficit.
- (h) Represents the cash consideration of \$68.9 million paid to the Sellers as part of the Secondary Sale Transaction with adjustment to common stock for the par value of the shares assumed to be repurchased and additional paid in capital.
- (i) To reflect the recapitalization of Innovid through the contribution of all outstanding common and preferred stock of Innovid to ION and the issuance of 93.8 million shares of Innovid Corp, subsequently reduced to 86.9 million shares as a result of the Secondary Sale (see note h). As a result of the recapitalization, the carrying value of Innovid's preferred stocks of \$140.0 million, common stock of \$0.014 million, treasury stock of \$1.6 million, and the warrants liability of \$3.7 million were derecognized. Innovid Corp,'s stocks issued as part of the recapitalization were recorded to common stock in the amount of \$0.008 million and additional paid in capital in amount of \$142.1 million.
- (j) Reflects ION's public shareholders exercise of their redemption rights totaling 19,585,174 ION Class A Common Stock prior to the consummation of the Business Combination at a redemption price of approximately \$10.00 per share, or \$195.8 million in cash.

Adjustments to Unaudited Pro Forma Condensed Combined Statement of Operations for the Year Ended December 31, 2020 and for the Nine Months Ended September 30, 2021

The pro forma notes and adjustments, based on preliminary estimates that could change materially as additional information is obtained, are as follows:

Pro forma Transaction Accounting Adjustments:

- (aa) To reflect an adjustment to eliminate the interest earned on marketable securities held in the Trust Account of ION.
- (ab) The pro forma basic and diluted earnings per share amounts presented in the unaudited pro forma condensed combined statement of operations are based upon the number of Innovid Corp. shares outstanding at the closing of the Transaction, assuming the Transaction occurred on January 1, 2020. As the unaudited pro forma condensed combined statements of operation are in a loss position, anti-dilutive instruments were not included in the calculation of diluted weighted average number of common shares outstanding.
- (ac) To reflect an accrual for estimated transaction bonuses of \$6.0 million to be paid to certain executives upon the completion of the Transaction within selling, general and administrative expense. This one-time adjustment was made only to the unaudited pro forma condensed combined statement of operations for the year ended December 31, 2020.

MANAGEMENT'S DISCUSSION AND ANALYSIS OF FINANCIAL CONDITION AND RESULTS OF OPERATIONS OF INNOVID

The following discussion and analysis should be read in conjunction with the unaudited condensed financial statements and audited financials statements and related notes of Innovid either included in the final prospectus and definitive proxy statement/prospectus (the "Proxy") relating to the Business Combination dated November 10, 2021 and filed with the Securities and Exchange Commission, or attached to this Current Report on Form 8-K. This discussion contains forward-looking statements reflecting our current expectations, estimates and assumptions concerning events and financial trends that may affect our future operating results or financial position. Actual results and timing of events may differ materially from those contained in these forward-looking statements due to a number of factors, including those discussed in the sections of the Proxy entitled "Risk Factors" and "General Information-Cautionary Note Regarding Forward-Looking Statements."

Unless the context otherwise requires, all references in this section to "Innovid," "we," "us," "our" or the "Company" refer to Innovid, Inc. and its subsidiaries prior to the consummation of the Business Combination.

Company Overview

We are a leading independent software platform that provides critical technology infrastructure for the creation, delivery, and measurement of TV ads across connected TV ("CTV"), mobile TV and desktop TV environments. As of September 30, 2021, over 40% of the top 200 brands by TV U.S. advertising spend according to Kantar Media and Winmo are utilizing our platform in their advertisement delivery infrastructure. Innovid's revenue has grown alongside the growth of CTV advertising. We believe our open platform and purpose-built technology for CTV, combined with our position as a media-independent provider, has allowed us to win a large and growing market share, while the growth of CTV combined with our usage-based revenue model has further contributed to our rapid growth. CTV accounted for 46%, 40% and 31% of all video impressions served by Innovid during the nine months ended September 30, 2021, the year ended December 31, 2020 and the year ended December 31, 2019. During the year ended December 31, 2020, this represented a year-over-year increase of almost 60%. The balance of video impressions served by Innovid during such periods were attributable to Mobile, 40%, 43% and 48%, respectively, and PC, 14%, 16% and 21%, respectively. In 2020, the year-over-year change in impressions for Mobile TV was an increase of 15% and for Desktop TV was a decrease of 3%. For the nine months ended September 30, 2021, impressions for CTV, Mobile TV and Desktop TV increased 63%, 35% and 29%, respectively compared to the comparable prior year period. An impression is the metric used to quantify the number of views of an advertisement. Impressions are measured by cost per mile (CPM), where mile refers to 1,000 impressions (or cost per thousand). For example, a CTV ad might have a CPM of \$25, meaning that the content owner, receives \$25 every time an ad is displayed 1,000 times within a designated program. Ad servers, such as Innovid, provide a pixel that is implemented within an ad. When an ad weth that pixel loads, an impression is

A key driver of CTV growth has been the evolving preferences of consumers. Consumers are increasingly cutting the cord and streaming TV content "Over the Top" ("OTT") through internet-connected devices rather than traditional broadcast, satellite or cable TV. We believe OTT content, which is typically delivered on-demand, seeks to provide a better user experience, and often saves the consumer money over traditional paid TV services. Advertisers seeking to engage these audiences are rapidly shifting dollars away from traditional TV mediums towards increasing budgets for CTV. Advertisers also can benefit from the shift to CTV as the digitally delivered ads can be personalized and measured in real time, similar to other digital advertising mediums such as internet browser-based formats. As a result, TV advertisers have better transparency, control and ultimately potential return on investment from their CTV advertising.

Innovid's purpose-built CTV infrastructure platform is comprised of three key offerings: Ad Serving Solutions, Creative Personalization Solutions and Measurement Solutions. Our software-based platform provides an open technology infrastructure that tightly integrates with the highly fragmented advertising technology and media ecosystem including Demand Side Platforms (DSPs) such as The Trade Desk and Amobee; Supply Side Platforms

(SSP) such as Magnite and Verizon Media; publishers such as Hulu and Peacock; and end user devices such as Amazon Fire and Samsung Smart TV. Our offering encompasses independent global ad serving, data-driven personalization, and new forms of measurement designed to connect all channels in a clean, comparable, and privacy-compliant manner. Although we work closely with the vendors who buy and sell media, our platform only facilitates the creation, delivery and measurement of advertisements and campaigns and we do not make purchasing decisions or facilitate the purchasing of advertisement inventory. Because we do not make ad buying or selling decisions we are able to maintain our independence and remain free of potential buying conflicts.

We target clients comprised of the largest global TV advertisers. In 2020, our blue-chip advertiser client base includes over 40% of the top 200 brands by TV U.S. advertising spend according to Kantar Media and Winmo. In addition we work closely with the top advertising agency holding companies such as WPP, Publicis Groupe, Omnicom, Interpublic Group of Cos. and Dentsu. Our clients are diversified across all major industry verticals, including consumer packaged goods, pharmaceutical and healthcare, financial services, automotive and technology. We believe Innovid's independence is critical to advertisers seeking an interoperable and open partner that is primarily focused on technology infrastructure. We define a core client as an advertiser that generates at least \$100,000 of annual revenue. We have a history of strong growth in our core client base, with over 90 core clients as of December 31, 2020. No individual core client (brand/advertiser) represented more than 13% of 2020 revenue. Together our core clients have typically generated more than 80% of our annual revenue from 2018 through 2020 demonstrating our continued focus on large enterprise customers. In the years 2018, 2019 and 2020 we had 77, 85, and 95 "core clients", that generated 84%, 86%, and 89% of the total company revenue in the corresponding periods. We do not include the number of core clients or percentage of revenue attributable to core clients for any interim periods as the calculation of customers that constitute core clients is calculated only at year end based on annual revenue.

Innovid serves customers globally through a delivery footprint covering over 70 countries, including the United States (U.S.), United Kingdom (U.K.), Mexico, Argentina, Colombia, Israel, Singapore, Japan and Australia. Approximately 8% of Innovid's revenue was generated by our customers outside of the U.S. for the year ended December 31, 2020. Our non-U.S. customers generated approximately 9% of the total revenue for nine months ended September 30, 2021.

Our revenue model is based on impressions volume and the cost per impression for our various ad serving services. For our core ad serving platform, we generate revenue from our advertising customers based on the volume of advertising impressions delivered, enabling us to grow as our customers increase their digital ad spend and corresponding ad impressions. Additionally, we generate revenue from creative services based on flat fee per projects and measurement solutions based on the volume of advertising impressions measured. As we introduce new products such as advanced measurement and creative capabilities including personalization and interactivity, we expect to be able to charge higher prices per impression volume.

The Transaction

On June 24, 2021, ION, Innovid, Merger Sub 1 and Inspire Merger Sub 2, LLC ("Merger Sub 2") entered into an Agreement and Plan of Merger (the "Merger Agreement"). On November 30, 2021, as contemplated by the Merger Agreement, ION consummated the merger transaction contemplated by the Merger Agreement (the "Business Combination") pursuant to the following steps (the "Closing"):

- Merger Sub 1 merged with and into Innovid (the "First Merger"), the separate corporate existence of Merger Sub 1 ceased with Innovid continuing as the surviving corporation (the "Surviving Corporation"),
- immediately thereafter, the Surviving Corporation merged with and into Merger Sub 2 (collectively with the First Merger, the "Mergers"), with Merger Sub 2 continuing as the surviving entity and a direct wholly owned subsidiary of ION,
- ION changed its name to "Innovid Corp." (the "Name Change"), pursuant to the terms and subject to the conditions set forth in the Merger Agreement,

- prior to the consummation of the Mergers, ION was redomiciled (the "Domestication") as a Delaware corporation in accordance with the Delaware General Corporation Law, the Cayman Islands Companies Act (As Revised) and the amended and restated memorandum and articles of association of ION (the "Cayman Constitutional Documents"), in connection with which ION effected a deregistration under the Companies Act and a domestication under Section 388 of the DGCL (by means of filing a certificate of corporate domestication with the Secretary of State of Delaware), and
- · the other transactions contemplated by the Merger Agreement and documents related thereto were consummated.

In connection with the Mergers and the Name Change, Innovid Corp. (formally ION) issued (a) 86,901,792 shares of common stock, par value \$0.0001 per share ("New Innovid Common Stock") to former equityholders of Innovid Inc. and (b) 20,000,000 shares of New Innovid Common Stock to PIPE Investors (as defined below)

Pursuant to the Merger Agreement, immediately prior to the Domestication, pursuant to the Cayman Constitutional Documents, each ION Class B Ordinary Share, par value \$0.0001 per share (each an "ION Class B Ordinary Share") then issued and outstanding automatically converted into one ION Class A Ordinary Share, par value \$0.0001 per share (each an "ION Class A Ordinary Share" together with the ION Class B Ordinary Shares, the "ION Shares"). Immediately following such conversion, upon the Domestication and the Mergers, (i) each issued and outstanding unit representing one (1) ION Class A Share and one-eighth (1/8) of the warrant to purchase one (1) ION Class A Share at a price of \$11.50 per share (the "ION Warrants") was automatically separated into the underlying ION Class A Share and one-eighth of an ION Warrant, (ii) each ION Class A Share issued and outstanding immediately prior to the Domestication was automatically converted into one share of New Innovid Common Stock, (iii) each whole ION Warrant was automatically converted into a redeemable warrant exercisable for one share of New Innovid Common Stock on the same terms as the ION Warrants (the "Public Warrants"), and (iv) each whole private placement warrant, exercisable for one share of New Innovid Common Stock on the terms and subject to the conditions set forth in the applicable warrant agreement (the "Private Placement Warrants" and together with the Public Warrants, the "Company Warrants"). No fractional Company Warrants were issued upon separation of the ION Units.

As previously announced, on June 24, 2021, concurrently with the execution of the Merger Agreement, ION entered into subscription agreements, pursuant to which certain accredited investors (the "PIPE Investors") agreed to purchase an aggregate of 15,000,000 Shares of New Innovid Common Stock at \$10.00 per share for an aggregate commitment amount of \$150,000,000. On October 18, 2021, ION entered into new subscription agreements with certain PIPE Investors, including funds affiliated with ION, pursuant to which some of the PIPE Investors collectively subscribed for an additional 5,000,000 shares of New Innovid Common Stock at \$10.00 per share for an aggregate commitment amount of \$50,000,000.

The Transaction were accounted for as a reverse recapitalization in accordance with U. S. GAAP. Under this method of accounting, ION is treated as the "acquired" company for accounting purposes and the Transaction will be treated as the equivalent of Innovid issuing stock for the net assets of ION, accompanied by a recapitalization.

Impact of COVID-19

The novel coronavirus ("COVID-19") pandemic has created, and may continue to create significant uncertainty in macroeconomic conditions, and the extent of its impact on our operational and financial performance will depend on certain developments, including the duration and spread of the outbreak and the impact on our customers. Based on public reporting and our observations, some advertisers in certain industries, such as the automotive industry, decreased their short-term advertising spending in light of supply chain disruptions and/or labor shortage. This in turn could negatively impact our revenues from such advertisers.

We have considered the impact of COVID-19 on our estimates and assumptions and determined that there were no material adverse impacts on the consolidated financial statements for the nine months ended September 30, 2021. As events continue to evolve and additional information becomes available, our estimates and assumptions may change materially in future periods.

We obtained an unsecured loan of \$3.5 million in April 2020 due to uncertainties related to COVID-19. The loan was obtained through Silicon Valley Bank ("SVB") under the Paycheck Protection Program (the "PPP Loan") pursuant to the Coronavirus Aid, Relief, and Economic Security Act (the "CARES Act") and the Paycheck Protection Program Flexibility Act (the "Flexibility Act"). In May 2020, we received \$0.5 million from Special Situations Investing Group II, LLC ("SSIG"), a related party of one of our investors, for the purpose of a partial repayment of the PPP Loan. We have since repaid the PPP Loan in the amount of \$3.0 million in June 2021. For more detail refer to our audited consolidated financial statement presented in the Proxy or attached to this Current Report on Form 8-K.

Key Factors Affecting Our Performance

There are a number of factors that have impacted, and we believe will continue to impact, our results of operations and growth. These factors include:

Continued market demand. Our performance is dependent on continued global demand across the advertising ecosystem for independent third-party ad serving and measurement of digital ads. Advertisers, programmatic platforms, social media channels and digital publishers are collectively placing increased emphasis on the quality and effectiveness of digital ad spend across all channels, formats and devices.

Our growth is primarily driven by the fastest growing segments of digital ad spend, mostly CTV, and our results depend on our ability to capture continued market growth.

Growth of volume of CTV ad impressions of existing customers Our results also depend on our ability to retain our existing customers and on our customers' continued investment in CTV advertising. Customer retention will continue to impact our results as TV investment continues to shift from linear to CTV and the volume of CTV impressions grows.

Upsell of additional services. An additional contributor to our efforts in expanding revenue generated by our customers is our investment in cross-selling our solutions. We cross-sell our personalized creative solutions to primary ad serving customers, who, for example, begin using our services with standard TV ads and then introduce personalized formats over time. We also have cross-selling efforts related to our advanced measurement solutions, which provide real-time metrics to inform optimization of TV campaigns while in market. The success of these efforts will impact our results of operations.

Global expansion

The majority of our clients are global advertisers and operate at a significant scale. Innovid serves customers globally through a delivery footprint covering over 70 countries, including the United States, the United Kingdom, Mexico, Argentina, Colombia, Israel, Singapore, Japan, Australia, and China. In 2020, less than 10% of Innovid's revenue was generated outside of the US and Canada.

We intend to continue to grow our footprint in international markets in order to meet the needs of our global customer base and to accelerate new customer acquisition in key geographies outside of North America. Our results of operations will be impacted by the success of our geographic expansion, and whether the expected ad spend growth in these markets materializes.

New client accounts: We intend to continue targeting new brand, media agency and digital publisher customers who are currently utilizing solutions provided by our competitors or point solutions. Our results of operations will be impacted by our ability to attract new customers.

Seasonality: We experience fluctuations in revenues that coincide with seasonal fluctuations in the digital ad spending of our customers, in particular television ad spending patterns. Advertisers often allocate the largest portion of their media budgets to the fourth quarter of the calendar year in order to coincide with increased holiday purchasing. As a result, the fourth quarter of the year typically reflects our highest level of revenues while the first quarter typically reflects our lowest level of revenues. We expect our revenues to continue to fluctuate based on seasonal factors that affect the advertising industry as a whole and for these seasonal fluctuations in ad spend to

impact quarter-over-quarter results. We believe that the year-over-year comparison of results more appropriately reflects the overall performance of our business. However, this traditional seasonality may also be impacted by certain external factors or major events that impact traditional television advertising patterns, such as the COVID-19 pandemic, which led to a lower industry spend in 2020 than we would have expected based on traditional seasonality of television advertising spend. In the first nine months of 2021 we observed more traditional seasonality in television advertising spend relative to the corresponding period of 2020.

Public company costs: We will incur additional legal, accounting and other expenses that we did not previously incur, including costs associated with SEC reporting and corporate governance requirements. These requirements include compliance with the Sarbanes-Oxley Act as well as other rules implemented by the SEC and the NYSE. Our financial statements will reflect the impact of these expenses in the future periods.

Components of Results of Operations

The period to period comparisons of our results of operations have been prepared using the historical periods included in our consolidated financial statements. The following discussion should be read in conjunction with the consolidated financial statements and related notes included in the Proxy or attached to this Current Report on Form 8-K.

Revenues

We generate revenues from providing Advertising Services to our customers: advertisers, media agencies and publishers. We focus on standard, interactive and data driven digital video advertising. Our major revenue streams are ad serving and creative services. Ad Serving services relate to utilizing Innovid's platform to serve advertising impressions to various digital publishers across CTV, mobile TV, desktop TV, display, and other channels. Creative services relate to the design and development of interactive data-driven and dynamic ad formats by adding data, interactivity and dynamic features to standard ad units. We also have a new offering, which is focused on measurement of the efficiency of CTV advertising and in-flight optimizations for TV marketers. We are planning to further develop and scale it in the future.

We generate the majority of our revenues from the sale and delivery of our products within the U. S. For information with respect to sales by geographic markets, refer to "—Revenue Recognition." Our chief operating decision maker (our Chief Executive Officer) does not evaluate the profit or loss from any separate geography.

We anticipate that revenues from our U.S. sales will continue to constitute a substantial portion of our revenues in future periods.

Cost of revenues

Cost of revenues consists primarily of costs to run our ad serving and creative services. These costs include hosting fees, personnel costs including stock-based compensation, professional services costs and facility related costs. We allocate overhead, including rent and other facility related costs, communication costs and depreciation expense, based on headcount.

Research and development

Research and development expenses consist primarily of personnel costs, including stock-based compensation, professional services costs and facility related costs. We allocate overhead including rent and other facility related costs, communication costs and depreciation expenses based on headcount. We expect research and development expenses to increase in future periods to support our growth, including continuing to invest in optimization, accuracy and reliability of our platform and other technology improvements to support and drive efficiency in our operations. These expenses may vary from period to period as a percentage of revenue, depending primarily upon when we choose to make more significant investments.

Sales and marketing

Sales and marketing expenses consist primarily of personnel costs, including stock-based compensation, professional services costs and facility related costs, as well as costs related to advertising, promotional materials, public relations, other sales and marketing programs. We allocate overhead, including rent and other facility related costs, communication costs and depreciation expense, based on headcount.

General and administrative

General and administrative expenses consist primarily of personnel costs, including stock-based compensation, for executive management, finance, accounting, human capital, legal and other administrative functions as well as professional services costs and facility related costs. We allocate overhead, including rent and other facility related costs, communication costs and depreciation expense, based on headcount.

Results of Operations

Nine months ended September 30, 2021 compared to nine months ended September 30, 2020

The following table sets forth our unaudited consolidated results of operations for the nine months ended September 30, 2021 and 2020. Our results of operations for the interim periods have been prepared on the same basis as our audited consolidated financial statements, and we believe reflect all normal recurring adjustments necessary for the fair presentation of our results of operations for these periods. This information should be read in conjunction with our audited consolidated financial statements and related notes included in the Proxy or attached to this Current Report on Form 8-K. These results of operations for the interim periods are not necessarily indicative of our results of operations for a full year or any future period.

	Nine months ended September 30,							
	2	2021	:	2020				
	(in thousands)	% of Revenue	(in thousands)	% of Revenue				
Revenues	\$ 64,324	100 %	\$ 45,772	100 %				
Cost of revenues	12,418	19 %	8,544	19 %				
Gross profit	51,906	81 %	37,228	81 %				
Operating expenses:								
Research and development	16,932	26 %	13,673	30 %				
Sales and marketing	23,534	37 %	22,624	49 %				
General and administrative	10,587	16 %	5,622	12 %				
Total operating expenses	51,053	79 %	41,919	92 %				
Operating profit/ (loss)	853	1 %	(4,691)	(10) %				
Finance expenses, net	3,878	6 %	528	1 %				
Loss before taxes	(3,025)	(5) %	(5,219)	(11) %				
Taxes on income	829	1 %	899	2 %				
Net loss	\$ (3,854)	(6) %	\$ (6,118)	(13) %				

Revenues

The growth and scaling of CTV was the key driver of Innovid's revenue growth. As TV ad spend continues to shift from linear to CTV, we continue to benefit from the natural volume growth of CTV impressions we delivered for our existing and new customers. We have driven consistent positive net revenue retention of our core client base, largely through increased CTV advertising volume, as legacy TV budgets migrate from linear TV to CTV.

Total revenue increased by \$18.5 million, or 41%, from \$45.8 million or the nine months ended September 30, 2020 as compared to \$64.3 million for the nine months ended September 30, 2021, driven primarily by growth in ad

impressions delivered on our platform for both existing and new clients. There was no meaningful impact of changes in average cost per impression on total revenue.

Cost of revenues

	 Nine months ended September 30,										
	2	2021									
				_							
	(in thousands)	% of Revenue	(in thousands)	% of Revenue	\$ Variance	% Variance					
Cost of revenues	\$ 12,418	19 %	\$ 8,544	19 %	\$ 3,874	45 %					

Cost of revenues increased by \$3.9 million, or 45%, from \$8.5 million for nine months ended September 30, 2020 to \$12.4 million for the nine months ended September 30, 2021, primarily driven by a \$1.5 million increase in serving and hosting fees and a \$2.2 million increase in personnel costs due to a higher headcount, both to support our increased volumes.

Research and development

		Nine months ended September 30,											
		202	1										
	t	(in housands)	% of Revenue	(in thousands)	% of Revenue	\$ Variance	% Variance						
Research and development	\$	16,932	26 % 5	13,673	30 %	\$ 3,259	24 %						

Research and development expenses increased by \$3.2 million, or 24%, from \$13.7 million in the nine months ended September 30, 2020 to \$16.9 million for the nine months ended September 30, 2021. The increase was primarily due to an increase of \$3 million in personnel costs as well an increase in technology infrastructure and hosting fees of \$0.8 million, both to support our platform enhancement and maintenance work as well as our product research efforts. The increase was partially offset by a \$1 million capitalization of research and development expenses.

Sales and marketing

				Nine months en	ded September 30,		
	_	20	021	2	2020		
	_	(in thousands)	% of Revenue	(in thousands)	% of Revenue	\$ Variance	% Variance
Sales and marketing	\$	23,534	37 %	\$ 22,624	49 %	\$ 910	4 %

Sales and marketing expenses increased by \$0.9 million, or 4%, from \$22.6 million for the nine months ended September 30, 2020 to \$23.5 million for the nine months ended September 30, 2021. The increase was driven primarily by higher commissions costs as a result of increased revenues.

General and administrative

				Nine months end	ded September 30,		
		20	021	20	020		
	1	(in thousands)	% of Revenue	(in thousands)	% of Revenue	\$ Variance	% Variance
General and administrative	\$	10,587	16 %	\$ 5,622	12 %	\$ 4,965	88 %

General and administrative expenses increased by \$5.0 million, or 88%, from \$5.6 million for the nine months ended September 30, 2020 to \$10.6 million for the nine months ended September 30, 2021. The increase was driven

primarily by a \$1.4 million increase in share-based compensation due to increase in headcount, value of our underlying common stock and vesting acceleration for several awards, an increase in personnel costs of \$1.1 million, and \$2 million increase in professional fees.

Finance expenses, net

	 Nine months ended September 30,										
	 20	021									
	(in thousands)	% of Revenue	(in thousands)	% of Revenue	\$ Variance	% Variance					
Finance expenses, net	\$ 3,878	6 %	\$ 5	28 1 %	\$ 3,350	634 %					

Finance expenses increased by \$3.4 million, or 634%, from \$0.5 million for the nine months ended September 30, 2020 to \$3.9 million for the nine months ended September 30, 2021. The increase was driven primarily by warrants valuation as a result of significantly increased valuation of the Company due to the announced SPAC merger transaction, improved market conditions and growth projections.

Taxes on income

		Nine months ended September 30,										
		2021 2020										
	th	(in thousands) % of Revenue				(in thousands)	% of Revenue		\$ Variance	% Variance		
Taxes on income	\$	829	1	%	\$	899	2	%	\$	(70)	(8) %	

Tax expense decreased by \$0.1 million, or 8% from \$0.9 million for the period ended September 30, 2020 to \$0.8 million for the period ended September 30, 2021. The decrease was primarily due to changes in U.S. state and foreign income taxes and lower discrete impact of changes in uncertain tax positions related to our Israel subsidiary. The changes in U.S. state and foreign income taxes are attributable to decreases in taxable income.

Liquidity and Capital Resources

We have financed our operations and capital expenditures primarily through utilization of cash generated from operations, as well as borrowings under our credit facilities. As of September 30, 2021, we had cash, cash equivalents and restricted cash of \$14.9 million and net working capital, consisting of current assets less current liabilities, of \$36.7 million. As of September 30, 2021, we had accumulated deficit of \$101.8 million, \$49.8 million thereof results from accretion of preferred stock to redemption value driven by an increase of our common stock value due to possible SPAC transaction.

We believe our existing cash and cash equivalents, together with anticipated net cash provided by operating activities and available borrowings under our credit facility, will be sufficient to meet our working capital requirements for at least the next 12 months. However, if our operating performance during the next 12 months is below our expectations, our liquidity and ability to operate our business could be adversely affected. We are closely monitoring the effect that current economic conditions may have on our working capital requirements. To date, the COVID-19 pandemic has not had a material negative impact on our cash flow or liquidity. Our future capital requirements and the adequacy of available funds will depend on many factors, including those set forth under "Risk Factors."

In the future, we may attempt to raise additional capital through the sale of equity securities or through equity-linked or debt financing arrangements. If we raise additional funds by issuing equity or equity-linked securities, the ownership of our existing stockholders will be diluted. If we raise additional financing by the incurrence of additional indebtedness, we may be subject to increased fixed payment obligations and could also be subject to additional restrictive covenants, such as limitations on our ability to incur additional debt, and other operating restrictions that could adversely impact our ability to conduct our business. Any future indebtedness we incur may result in terms that are unfavorable to equity investors. We cannot guarantee that we will be able to raise additional

capital in the future on favorable terms, or at all. Any inability to raise capital could adversely affect our ability to achieve our business objectives.

Revolving Line of Credit

In 2016, we entered into additional modifications to the credit line agreement dated 2012 (the "Agreement"), pursuant to which to which certain conditions were amended, the Maturity Date was extended to October 21, 2018 and the line of credit was increased from \$6.5 million to \$10 million.

On April 7, 2017 we utilized \$5 million of the line of credit. The credit installments bear U.S. dollar denominated interest at an annual rate equal to .75%-1% plus a prime rate on the outstanding principal of each credit installment. The balance owing as of December 31, 2017 was \$5 million.

On October 20, 2018, we entered into additional modifications to the Agreement, pursuant to which certain conditions were amended and the Maturity Date was extended to December 31, 2018.

On December 26, 2018, we entered into an amended and restated Agreement (the "A&R Agreement"), pursuant to which certain conditions were amended, the Maturity Date was extended to December 26, 2020 and the line of credit was increased to from \$10 million to \$12 million.

On September 1, 2018 we utilized an additional \$1 million of the line of credit. The credit installments bore U.S. dollar denominated interest at an annual rate equal to .75%-1% plus a prime rate on the outstanding principal of each credit installment. The Maturity Date was December 26, 2020. The balance owing as of December 31, 2018 was \$6 million

On November 30, 2019, we fully repaid the outstanding balance of the credit line in the amount of \$6 million.

During 2020, the we fully drew down on our \$12 million credit line. As of December 31, 2020, we had repaid \$6 million, leaving a balance of \$6 million. On December 29, 2020, we entered into additional modifications to the A&A Agreement, pursuant to which certain conditions were amended and the Maturity Date was extended to December 29, 2022, and the line of credit increased to \$15 million.

As of September 30, 2021 the outstanding balance of the credit line was in the amount of \$6 million. The credit installments bear U.S. dollar denominated interest at an annual rate equal to .75%-1% plus a prime rate on the outstanding principal of each credit installment. We were in compliance with all the covenants, including by maintaining an adjusted quick ratio of at least 1.20:1.00. As defined in the A&R Agreement "adjusted quick ratio" is the ratio of (a) quick assets to (b) current liabilities minus the current portion of deferred revenue. "Quick assets" determines as our unrestricted cash plus accounts receivable, net, determined according to U.S. GAAP.

During nine months ended September 30, 2021, we continued utilizing \$6 million of a \$15 million credit line which was drawn during 2020. As of September 30, 2021, the covenants under the Agreement were not changed from the amended Agreement. We are in compliance with all the covenants.

PPP Loan

In April 2020, we obtained an unsecured loan of \$3.5 million through SVB under the PPP Loan.

In May 2020, we entered into a grant agreement (the "Grant Agreement") to receive a grant of \$0.5 million from SSIG, a related party of one of our investors, for the purpose of repayment of a portion of the PPP Loan. The PPP loan was partially repaid in May 2020, according to the Grant Agreement.

In June 2021, we repaid the outstanding balance of the PPP Loan of \$3.0 million

Interest expenses for the Credit Line and PPP Loan for the nine months ended September 30, 2021 and 2020 were\$0.2 million and \$0.2 million, respectively. They were recorded in finance expenses, net in the unaudited interim condensed consolidated financial statements of operations.

Cash Flows

Nine months ended September 30, 2021 compared to nine months ended September 30, 2020

The following table summarizes our cash flows for the nine months ended September 30, 2021 and 2020:

	Nine	Nine months ended September 30,					
	2021		2020				
Net cash provided by/ (used in) operating activities	\$	3,046 \$	(2,477)				
Net cash used in investing activities		(1,944)	(745)				
Net cash (used in)/ provided by financing activities		(2,277)	9,559				
(Decrease)/ increase in cash, cash equivalents and restricted cash	\$	(1,175) \$	6,337				

Operating Activities

Our cash flows from operating activities are primarily influenced by growth in our operations, increases or decreases in collections from our customers and payments to our vendors, as well as increases in personnel costs as we scale up our business. The timing of cash receipts from customers and payments to vendors and providers can significantly impact our cash flows from operating activities. In addition, we expect seasonality to impact quarterly cash flows from operating activities.

Cash provided by/ (used in) operating activities is calculated by adjusting our net loss for changes in working capital, as well as by excluding non-cash items such as depreciation and amortization, stock-based compensation and changes in fair value of warrants.

For the nine months ended September 30, 2021, net cash provided by operating activities was \$3.0 million compared to net cash used of \$2.5 million for the nine months ended September 30, 2020. The decrease in net cash used in operating activities during 2021 as compared to 2020 was primarily attributable to a decrease in net loss. Our non-cash adjustments increased by \$5.0 million mostly driven by valuation of warrants and stock options granted in 2021 as a result of significantly increased valuation of the Company due to possible SPAC transaction, improved market conditions and growth projections, together with vesting acceleration for several awards.

The change in our working capital in the amount of \$1.7 million was the result of increases in trade receivables and trade payables related to increased revenue and operating activities, and decrease in accrued liabilities primarily related to timing of payment for personnel cost and leases.

Investing Activities

For the nine months ended September 30, 2021, we used \$1.9 million of net cash in investing activities, primarily driven by the investment in software development work of \$1.0 million and loan in the amount of \$0.5 million issued to our founder. This loan was forgiven in November, 2021.

For the nine months ended September 30, 2020, we used \$0.7 million of net cash in investing activities, primarily consisting of leasehold improvements for our office in Israel.

Financing Activities

For the nine months ended September 30, 2021, net cash used in financing activities of \$2.3 million was primarily due to repayment of PPP loan in the amount of \$3.0 million, partially offset by proceeds received for exercises of options in the amount of \$0.9 million.

For the nine months ended September 30 2020, net cash provided by financing activities of \$9.6 million was primarily due to proceeds received under our credit line agreement.

Contractual Obligations and Future Cash Requirements

Lease commitments:

We rent our facilities and certain motor vehicles under operating lease agreements that expire on various dates, the latest of which is in 2025. The minimum rental payments under operating leases for rental of premises as of September 30, 2021 for the next five years totaled \$6.3 million. Other operating leases are immaterial.

Pledges and bank guarantees:

In connection with the Agreement, we pledged 65,000 shares of common stock of our Israeli Subsidiary, NIS 0.01 par value each.

We have a total of \$0.7 million of pledged bank deposits as of September 30, 2021. We obtained bank guarantees in an aggregate amount of \$0.3 million in connection with our office lease agreements in the U.S as of September 30, 2021.

Key Metrics and Non-GAAP Financial Measures

Adjusted EBITDA

In addition to our results determined in accordance with U. S. GAAP, we believe that certain non-GAAP financial measures, including Adjusted EBITDA and Adjusted EBITDA Margin, are useful in evaluating our business. We calculate Adjusted EBITDA Margin as Adjusted EBITDA divided by total revenue. Non-GAAP financial measures have limitations as analytical tools and should not be considered in isolation or as substitutes for an analysis of our results as reported under U.S. GAAP. In addition, other companies in our industry may calculate non-GAAP financial measures differently than we do, limiting their usefulness as a comparative measure. The following table presents a reconciliation of Adjusted EBITDA, a non-GAAP financial measure, to the most directly comparable financial measure prepared in accordance with GAAP.

		Three months ended September 30,				Nine months end	ed Sep	tember 30,	Year ended December 31,				
		2021		2020		2021	2020		2020			2019	
Net (loss)/income	\$	(259)	\$	2,468	\$	(3,854)	\$	(6,118)	\$	(812)	\$	(7,334)	
Net loss margin		(1)%		13 %	(6)%			(13)%		(1)%		(13)%	
Depreciation and amortization		156		177		487		475		730		431	
Stock-based compensation	1	591		94		2,311	457		584			378	
Finance expense, net (a)		707		175		3,878		528		734		387	
Other (b)		_		_		_		153		153		_	
Taxes on income		304		178		829		899		1,200		902	
Adjusted EBITDA	\$	1,499	\$	3,092	\$	3,651	\$	(3,606)	\$	2,589	\$	(5,236)	
Adjusted EBITDA margin		6 %		17 %		6 %		(8)%		4 %		(9)%	

⁽a) Finance expense, net consists mostly of remeasurement expense related to our Argentinian subsidiary's monetary assets, liabilities and operating results, our interest expense and revaluation of our warrants.

(b) Other consists predominantly of the loss related to a one time loss from sale of fixed assets in our Israel subsidiary.

We use Adjusted EBITDA and Adjusted EBITDA Margin as measures of operational efficiency to understand and evaluate our core business operations. We believe that these non-GAAP financial measures are useful to investors for period to period comparisons of our core business and for understanding and evaluating trends in our operating results on a consistent basis by excluding items that we do not believe are indicative of our core operating performance.

These non-GAAP financial measures have limitations as analytical tools and should not be considered in isolation or as substitutes for an analysis of our results as reported under GAAP. Some of the limitations of these measures are:

- · they do not reflect changes in, or cash requirements for, our working capital needs;
- · Adjusted EBITDA does not reflect our capital expenditures or future requirements for capital expenditures or contractual commitments;
- · they do not reflect income tax expense or the cash requirements to pay income taxes;
- · they do not reflect our interest expense or the cash requirements necessary to service interest or principal payments on our debt; and
- although depreciation and amortization are non-cash charges related mainly to intangible assets, certain assets being depreciated and amortized will have to be replaced in the future, and Adjusted EBITDA does not reflect any cash requirements for such replacements.

In addition, other companies in our industry may calculate these non-GAAP financial measures differently than we do, limiting their usefulness as a comparative measure. You should compensate for these limitations by relying primarily on our U. S. GAAP results and using the non-GAAP financial measures only supplementally. We calculate Adjusted EBITDA Margin as Adjusted EBITDA divided by total revenue.

Operational Metrics

In addition, Innovid's management considers Net Revenue Retention and Core Client Retention in evaluating the performance of the business. Net Revenue Retention is defined as the percentage of revenue retained from existing core platform customers (core customers that use our platform as an ad server of record) as compared to the prior year period. Innovid's management believes that Net Revenue Retention is a useful metric for management and investors in evaluating Innovid's value proposition to customers and its ability to retain customers. For the years ended December 31, 2018, 2019 and 2020, Innovid's Net Revenue Retention was 110%, 114% and 121%, respectively. Core Client Retention is defined as the percentage of core platform clients retained by Innovid compared to the prior year period. Innovid's management believes that Core Client Retention is a useful metric for management and investors in evaluating the strength of core customer relationships. For the years ended December 31, 2018, 2019 and 2020, Innovid's Core Client Retention was 84%, 88% and 94%, respectively.

Off-Balance Sheet Arrangements

As of September 30, 2021, we did not have any relationships with unconsolidated organizations or financial partnerships, such as structured finance or special purpose entities that would have been established for the purpose of facilitating off-balance sheet arrangements or other contractually narrow or limited purposes.

Critical Accounting Policies and Estimates

Our discussion and analysis of our financial condition and results of operations are based on our consolidated financial statements, which have been prepared in accordance with generally accepted accounting principles in the U.S. ("GAAP"). The preparation of these financial statements requires us to make estimates, assumptions and judgments that affect the amounts reported in our consolidated financial statements and the accompanying notes to consolidated financial statements. We base our estimates on historical experience and on various other assumptions

that we believe to be reasonable under the circumstances, including the ongoing and potential impacts of the COVID-19 pandemic and related government mandates and restrictions. Actual results may differ from these estimates.

While our significant accounting policies are described in more detail in Note 2 of our audited consolidated financial statements included in the Proxy, we believe the following accounting policies to be the most critical to the judgments and estimates used in the preparation of our consolidated financial statements.

Revenue Recognition

The Company generates revenues from providing Advertising Services to advertisers, publishers and media agencies. The services focus on standard, interactive and data driven digital video advertising. The Company major revenue streams are ad serving and creative services. Ad Serving services relate to utilizing Innovid's platform to serve advertising impressions to various digital publishers across CTV, mobile TV, desktop TV, display, and other channels. Creative services relate to the creation of interactive or dynamic ad units by adding interactivity and dynamic features to standard ad units.

The Company adopted ASC, Revenue from Contracts with Customers Topic 606 ("ASC 606") with a date of initial application of January 1, 2018, using the modified retrospective transition method, applied to all open contracts.

The Company recognizes revenue when its customer obtains control of promised services in an amount that reflects the consideration that the company expects to receive in exchange for those services. The Company recognizes revenue in accordance with ASC Topic 606, Revenue from contracts with customers ("ASC 606") and determines revenue recognition through the following steps: (1) identify the contract with a customer, (2) identify the performance obligations in the contract, (3) determine the transaction price, (4) allocate the transaction price to the performance obligations in the contract, and (5) recognize revenue when a performance obligation is satisfied.

For arrangements with multiple performance obligations, which represent promises within an arrangement that are capable of being distinct and are separately identifiable, the Company allocates the contract consideration to all distinct performance obligations based on their relative stand-alone selling price ("SSP").

Revenues related to ad serving services are recognized at a point in time. The Company recognizes revenue from the display of impression-based ads in the contracted period in which the impressions are delivered. Impressions are considered delivered when an ad is displayed to users.

Revenues related to creative services are recognized at a point in time, when the Company delivers an ad unit, since the Company does not have enforceable right to payment before delivery. Creative services projects are usually delivered within a week.

The Company's accounts receivable, consist primarily of receivables related to providing ad serving and creative services, in which the Company's contracted performance obligations have been satisfied, amount billed and the Company has an unconditional right to payment. The Company typically bills customers on a monthly basis based on actual delivery. The payment terms vary, mainly with terms of net 60 days or less.

Typical contract term is twelve months or less for ASC 606 purposes. Some of the Company's contracts can be cancelled without a cause. The Company has unconditional right to payment for the services provided as of the date of the termination of the contracts.

The Company applies the practical expedient in ASC 606 and does not adjust the promised amount of consideration for the effects of a significant financing component if the Company expects, at contract inception, that the period between when the Company transfers a promised good or service to a customer and when the customer pays for that good or service will be one year or less.

Costs to obtain a contract

Contract costs include commission programs to compensate sales employees for generating sales orders with new customers or for new services with existing customers. The Company elected to apply the practical expedient and recognize incremental costs of obtaining a contract as an expense when incurred if the amortization period of the asset that the Company otherwise would have recognized is one year or less.

Warrants

The Company accounts for warrants as either equity-classified or liability-classified instruments based on an assessment of the warrant's specific terms and applicable authoritative guidance. The assessment considers whether the warrants are freestanding financial instruments, meet the definition of a liability under ASC 480, and meet all of the requirements for equity classification, including whether the warrants are indexed to the Company's own common stock and whether the warrant holders could potentially require "net cash settlement" in a circumstance outside of the Company's control, among other conditions for equity classification. This assessment, which requires the use of professional judgment, is conducted at the time of warrant issuance and as of each subsequent reporting period end date while the warrants are outstanding.

Warrants that meet all the criteria for equity classification, are required to be recorded as a component of additional paid-in capital. Warrants that do not meet all the criteria for equity classification, are required to be recorded as liabilities at their initial fair value on the date of issuance and remeasured to fair value at each balance sheet date thereafter. The liability-classified warrants are recorded under non-current liabilities. Changes in the estimated fair value of the warrants are recognized in "Financial expenses, net" in the consolidated statements of operations.

Fair value of financial instruments

The Company applies a fair value framework in order to measure and disclose its financial assets and liabilities. Fair value is defined as the exchange price that would be received for an asset or paid to transfer a liability (an exit price) in the principal or most advantageous market for the asset or liability in an orderly transaction between market participants on the measurement date. The fair value hierarchy requires an entity to maximize the use of observable inputs, where available, and minimize the use of unobservable inputs when measuring fair value. There are three levels of inputs that may be used to measure fair value:

- Level 1 Observable inputs that reflect quoted prices (unadjusted) for identical assets or liabilities in active markets.
- Level 2 Include other inputs that are directly or indirectly observable in the marketplace.
- Level 3 Unobservable inputs which are supported by little or no market activity.

The Company's financial instruments consist of cash and cash equivalents, restricted deposits, trade receivables, net, and trade payables. Their historical carrying amounts are approximate fair values due to the short-term maturities of these instruments.

The Company measures its investments in money market funds classified as cash equivalents and warrants liability at fair value.

The warrants were classified as level 3 in the fair value hierarchy because some of the inputs used in the valuation (the stock price) were determined based on management's assumptions. The Company estimates the fair value of the Warrants liability using Black-Scholes option pricing model. Gains and losses from the remeasurement of the warrants liability are recognized in finance expenses, net in the consolidated statements of operations.

Stock-based compensation

The Company estimates the fair value of stock-based awards on the date of grant. The fair value of stock options with only service conditions is determined using the Black-Scholes option pricing model. The grant date fair

value of the stock-based awards with graded vesting is recognized on a straight-line basis over the requisite service period. The determination of the fair value of the Company's stock option awards is based on a variety of factors including Company's common stock price, risk-free interest rate, expected volatility, expected life of awards and dividend yield. The Company has limited option exercise history and has elected to estimate the expected life of the stock option awards using the "simplified method" with the continued use of this method extended until such time that the Company has sufficient exercise history. The expected volatility of the price of such stocks is based on volatility of similar companies whose stock prices are publicly available over a historical period equivalent to the option's expected term. The expected term of options granted represents the period of time that options granted are expected to be outstanding, and is determined based on the simplified method, as adequate historical experience is not available to provide a reasonable estimate. The dividend yield is based on the Company's historical and future expectation of dividends payouts. Historically, the Company has not paid cash dividends. Risk-free interest rates are based on the yield from U.S. Treasury zero-coupon bonds with a term equivalent to the expected term of the options.

The Company accounts for forfeitures as they occur.

Goodwill and intangible assets

Goodwill and certain other purchased intangible assets have been recorded in the Company's consolidated financial statements as a result of the acquisition of the Argentinian Subsidiary. Goodwill represents the excess of the purchase price in a business combination over the fair value of identifiable tangible and intangible assets acquired and liabilities assumed.

The Company allocates goodwill to reporting units based on the expected benefit from the business combination. Reporting units are evaluated when changes in the Company's operating structure occur, and if necessary, goodwill is reassigned using a relative fair value allocation approach. The Company currently has one reporting unit.

Goodwill is not amortized and is tested for impairment at least annually, on October 1, or more often if and when circumstances indicate that goodwill is not recoverable. The Company assesses qualitative factors to determine whether it is more likely than not that the fair value of a reporting unit is less than its carrying value. Qualitative factors considered in the assessment include industry and market conditions, overall financial performance, and other relevant events and factors affecting the reporting unit. If, after assessing the qualitative factors, the Company determines that it is not more likely than not that the fair value of a reporting unit is less than its carrying value, then performing a quantitative impairment test is unnecessary. If the Company concludes otherwise, it performs a quantitative goodwill impairment test. The quantitative impairment test, compares the fair value of a reporting unit with its carrying amount. If the fair value of a reporting unit exceeds its carrying amount, goodwill of the reporting unit is not considered impaired. If the carrying value of the reporting unit exceeds its fair value, any excess of the reporting unit goodwill carrying value over the respective fair value is recognized as an impairment loss.

Separately acquired intangible assets are measured on initial recognition at cost including directly attributable costs. Intangible assets acquired in a business combination are measured at fair value at the acquisition date.

Intangible assets with a finite useful life are amortized over their useful life and reviewed for impairment whenever there is an indication that the asset may be impaired.

Capitalized software development costs

Capitalized software development costs, which is included in property and equipment, net, consists of costs to purchase and develop internal-use software, which the Company uses to provide services to its customers. The costs to purchase and develop internal-use software are capitalized from the time that the preliminary project stage is completed, and it is considered probable that the software will be used to perform the function intended. These costs include personnel and related employee benefits for employees directly associated with the software development and external costs of the materials or services consumed in developing or obtaining the software. Any costs incurred during subsequent efforts to upgrade and enhance the functionality of the software are also capitalized. Once this software is ready for use in providing the Company's services, these costs are amortized on a straight-line basis over

the estimated useful life of the software, which is 3 years. The amortization will be presented within cost of revenues in the consolidated statements of operations.

Income taxes and tax contingencies

Income taxes are computed using a balance sheet approach reflecting both current and deferred taxes. Current and deferred taxes reflect the tax impact of all of the events included in the financial statements. The basic principles employed in the balance sheet approach are to reflect a current tax liability or asset that is recognized for the estimated taxes payable or refundable on tax returns for the current and prior years, a deferred tax liability or asset that is recognized for the estimated future tax effects attributable to temporary differences and carryforwards, the measurement of current and deferred tax liabilities and assets is based on provisions of the enacted tax law of which the effects of future changes in tax laws or rates are not anticipated, and the measurement of deferred tax assets is reduced, if necessary, by the amount of any tax benefits that, based on available evidence, are not expected to be realized. There are certain situations in which deferred taxes are not provided. Some basis differences are not temporary differences because their reversals are not expected to result in taxable or deductible amounts.

The Company regularly evaluates deferred tax assets for future realization and establish a valuation allowance to the extent that a portion is not more likely than not to be realized. The Company considers whether it is more likely than not that the deferred tax assets will be realized, including existing cumulative losses in recent years, expectations of future taxable income, carryforward periods, and other relevant quantitative and qualitative factors. The recoverability of the deferred tax assets is evaluated by assessing the adequacy of future expected taxable income from all sources, including reversal of taxable temporary differences, forecasted operating earnings and available tax planning strategies. These sources of income rely on estimates.

ASC 740, Income Taxes ("ASC 740) contains a two-step approach to recognizing and measuring a liability for uncertain tax positions. The first step is to evaluate the tax position taken or expected to be taken in a tax return by determining if the weight of available evidence indicates that it is more likely than not that, on an evaluation of the technical merits, the tax position will be sustained on audit, including resolution of any related appeals or litigation processes. The second step is to measure the tax benefit as the largest amount that is more than 50% (cumulative basis) likely to be realized upon ultimate settlement. The Company classifies interest related to unrecognized tax benefits in taxes on income.

On December 20, 2017, Congress passed the Tax Cuts and Jobs Act (the "U. S. Tax Act"). The U.S. Tax Act requires complex computations to be performed that were not previously required by U.S. tax law, significant judgments to be made in interpretation of the provisions of the U.S. Tax Act, significant estimates in calculations, and the preparation and analysis of information not previously relevant or regularly produced the Act provides that a person who is a U.S. shareholder of any controlled foreign corporation ("CFC") is required to include its global intangible low-taxed income ("GILTI") in gross income for the tax year in a manner generally similar to that for Subpart F inclusions. The term "global intangible low-taxed income" is defined as the excess (if any) of the U.S. shareholder's net CFC tested income for that tax year, over the U.S. shareholder's net deemed tangible income return for that tax year. The Company's policy is to treat GILTI as a period expense in the provision for income taxes.

Functional currency

A majority of the Company's revenues are generated in U.S. dollars. In addition, a substantial portion of the Company's costs are incurred in U.S. dollars. The Company's management believes that the U.S. dollar is the currency of the primary economic environment in which the Company and each of its subsidiaries operate. Thus, the functional and reporting currency of the Company and its subsidiaries is the U.S. dollar. Accordingly, accounts maintained in currencies other than the U.S. dollar are re-measured into U.S. dollars. All translation gains and losses resulting from the re-measurement of monetary assets and liabilities that are not denominated in the functional currency are recorded in Financial expenses, net on the consolidated statements of operations.

Recent Accounting Pronouncements

See Note 2 to our unaudited condensed consolidated financial statements and audited consolidated financial statements included in the Proxy or attached to this Current Report on Form 8-K, for recently adopted accounting pronouncements and recently issued accounting pronouncements not yet adopted.

Quantitative and Qualitative Disclosures about Market Risk

Financial instruments that potentially subject the Company to concentrations of credit risk consist principally of cash and cash equivalents and trade receivables, net.

The majority of the Company's cash and cash equivalents are invested in deposits with major banks in America and Israel. Such investments in the U. S. may be in excess of insured limits and are not insured in other jurisdictions. Generally, these investments may be redeemed upon demand and, therefore, bear minimal risk.

The Company's trade receivables, net are mainly derived from sales to customers located in the U. S., Asia-Pacific region ("APAC"), Europe, the Middle East and Africa region ("EMEA"), and Latin America region ("LATAM"). The Company mitigates its credit risks by performing an ongoing credit evaluations of its customers' financial conditions.

The Company have no off-balance-sheet concentration of credit risk such as foreign exchange contracts, option contracts or other foreign hedging arrangements.

The Company has customer concentration as disclosed in Note 2 to our unaudited condensed consolidated financial statements and audited consolidated financial statements included elsewhere in the Proxy or attached to this Current Report on Form 8-K.